HANFORD TANK WASTE TREATMENT AND

IMMOBILIZATION PLANT



BIDDER REQUEST FOR INTEREST & PRE-QUALIFICATION PACKAGE

OFFICIAL USE ONLY (when completed) May be exempt from public release under the Freedom of Information Act (5 U.S.C. 552), exemption number and category: <u>4.</u> <u>Commercial/Proprietary</u> Department of Energy Review required before public release Name/Org: <u>Jose Velasquez/ P&S</u> Date: <u>12/28/2023</u> Guidance (if applicable): <u>N/A</u>

Requisition No. 24590-QL-MRA-PS02-00020

Dual-Contained Pipe Spools



BIDDER REQUEST FOR INTEREST & PREQUALIFICATION

CRITERIA AND RESPONSE

1.0 Introduction

Bechtel National Inc., herein referred to as Contractor, intends to issue a Request for Proposal (RFP) for a Plant Material Purchase Order for the Hanford Waste Treatment and Immobilization Plant (WTP) project in Richland, WA. Companies must be pre-qualified by Contractor to be included on the bid list. To support the pre-qualification evaluation process, the prime potential bidder (1st tier subcontractor) must provide the requested information and respond to questions within this document. The Experience Statement should include relevant information for both the prime bidder and any planned lower-tier supplier or subcontractor. Additional supporting documentation such as brochures and company profiles may also be submitted.

*Please note that additional supporting documentation will be required as part of the formal RFP process.

2.0 <u>Project Description and Location</u>

The Hanford Tank Waste Treatment and Immobilization Plant (WTP) is a complex of radioactive waste treatment processing facilities designed and constructed by Bechtel National, Inc. for the Department of Energy (DOE). The facility will process the Hanford Site tank waste and convert the waste into a stable glass form.

The Project site is located in the 200 East Area of the Hanford Reservation near Richland, Washington, along the Columbia River. The site elevation varies from 662 to 684 feet above mean sea level. Ambient temperature range is -23 degrees F minimum to 113 degrees F maximum, with relative humidity of 5% minimum to 100% maximum. The project design life is 40 years.

Information about the WTP Project can be found on <u>http://www.hanfordvitplant.com</u>.

3.0 Scope of Work

Award Type: Firm Fixed Price Purchase Order with Economic Price Adjustment Estimated RFP Date: *March 2026*

Fabrication of Dual-Contained Pipe Spools

Work to be included:

- 1. Work performed per ASME B31.3 current version
- 2. Purchase piping bulk material
- 3. Welding of Piping (isometrics will be provided)
- 4. Cold bending of pipe
- 5. Pickling
- 6. Coating (Fusion Bonded Epoxy, FBE) and Insulation (High Density Polyethylene, HDPE jacketing
- 7. Welding in Valves. WTP will provide valves, some valves will have soft seats, temperatures will require monitoring to prevent damage
- 8. Spools will have pipe supports. Pipe support drawings provided by WTP (supports between the inner and outer pipe, stanchions on outer pipe.
- 9. Material Testing
 - a) Independent testing of bulk materials for full physical and chemical properties to associated ASTMs utilizing laboratories accredited to ISO/IEC 17025. Testing will result in Certified Material Test Reports Confirming MTRs received with purchased materials. Sample lot sizes tables for testing will be provided.
 - b) Independent testing of weld rod materials for full chemical properties to associated ASTMs utilizing laboratories accredited to ISO/IEC 17025.

OFFICIAL USE ONLY (when completed)



BIDDER REQUEST FOR INTEREST & PREQUALIFICATION

CRITERIA AND RESPONSE

- c) Impact testing for piping, weld deposit and heat affected zone
- d) Fittings per ASME B16.9, MSS SP-95, and MSS SP-97
- e) Pressure (hydrostatic and pneumatic) testing of spools
- 10. Material Inspections
 - a) All measuring and testing equipment utilized shall be calibrated by an accredited ISO/IEC 17025 laboratory.
 - b) Wall thickness measurements shall be documented and submitted to the Buyer
 - c) NDE of welds (RT), Weld repair, post weld heat treatment results shall be submitted to the Buyer

Equipment and Materials Required

Supplier shall provide all equipment and materials necessary to accomplish work in section above

<u>Codes</u>

- ASME B31.3
- ASME Boiler and Pressure Vessel Code

Standards

• Various ASME and ASTMs for piping components and materials ASNT SNT-TC-1A, Recommended Practice No. SNT-TC-1A Personnel Qualification and Certification in Nondestructive Testing

• ASME B16.9, MSS SP-95, and MSS SP-97

Please note that this solicitation may result in material procurements and proposals greater than \$10,000 and must comply with FAR 52.225-11 Buy American Act – Construction Materials Under Trade Agreements (SEP 2010). If you cannot comply or foresee any issues with compliance, please provide a detailed explanation.

If your company is interested in this solicitation, please proceed to Section 4.0 and complete the below sections as requested. The BNI Acquisition Services Purchasing group is responsible for collection, evaluation, and internal publication of potential bidders' information for the purpose of pre-qualification for all solicitations.

OFFICIAL USE ONLY (when completed)



5.1

BIDDER REQUEST FOR INTEREST & PREQUALIFICATION

CRITERIA AND RESPONSE

4.0 <u>Response Submittal</u>

 4.1 Submission Due Date: May 31, 2025
Submission Method: Submissions must be received no later than the due date to the Purchasing Representative, Andrea Riste, via email at adriste@bechtel.us. For questions, call (509) 430-9055.

5.0 <u>Pre-Selection Criteria</u>

Company Response

Prime Subcontractor Company Name: Address:

Pre-qualification Contact Name:

Phone Number:

E-mail Address:

DUNS No. (Dun & Bradstreet):

5.2 North American Industry Classification System Code (NAICS)

The NAICS (North American Industry Classification System code for this work is **332996.** The SBA size standard for this code is **550 employees**. For pre-qualification purposes, you are a small business if your company's number of employees does not exceed 550 employees.

	□ Small
	Small Disadvantaged Business
Business Size Classification (according to	U Woman Owned Small Business
U.S. Small Business Administration Criteria)	HUBZone Business
	Uveteran-Owned Small Business Concern
	Service-Disabled Veteran-Owned Small Business Concern.
	□ N/A – Registered as a Large Business

5.3 Quality Assurance Requirements Program

Programmatic Quality Assurance (QA) requirements for this purchase orders will be in accordance with Supplier QA program, please mark as applicable:

Commercial Quality - Based on DOE Order 414.1C
Nuclear Level Quality - Based on ASME NQA-1 2022

OFFICIAL USE ONLY (when completed)

Page 4 of 8



BIDDER REQUEST FOR INTEREST & PREQUALIFICATION

CRITERIA AND RESPONSE

A. Does your Company have a written Quality Assurance Program?

□Yes □ No

B. Which QA standards does this program meet?

DOE/RW/0333P		ASME NQA-1		ASME Section VIII	\Box	180-9000 E		Othe
	DOE/RW/0333P	DOE/RW/0333P	DOE/RW/0333P ASME NQA-1	DOE/RW/0333P ASME NQA-1	DOE/RW/0333P ASME NQA-1 ASME Section VIII	DOE/RW/0333P	DOE/RW/0333P ASME NQA-1 ASME Section VIII ISO-9000	DOE/RW/0333P ASME NQA-1 ASME Section VIII ISO-9000

C. If selected "other" above, please furnish a copy of its QA Program Table of Contents and a brief summary identifying each of the requirements listed below. The level of rigor applied to the elements shall be commensurate with the risks associated with the Work.

- A description of the organizational structure, functional responsibilities, levels of authority, and interfaces for those managing, performing, and assessing the Work.
- Personnel Training and Qualifications
- Quality Improvement
- Control of Documents and Records
- Work Processes
- Design
- Procurement
- Product Identification and Traceability
- Inspection and Acceptance Testing
- Control of the Testing Equipment
- Control of Non-Conforming Product
- Corrective and Preventative Actions
- Handling, Storage and Shipping Procedures
- Management Assessment
- Independent Assessment
- D. Your company has the option to submit their full Quality Assurance Plan with this interest.

5.4 Commercial Data

Potential bidders are required to register on the <u>Bechtel Supplier and Contractor Portal:</u> <u>https://www.Bechtel.com/supplier/</u> in order to be considered.

What was the date your company registered or updated its information on the Portal?

Date Updated:

- A. Rough Order of Magnitude for Scope of Work (USD): \$_____
- B. Estimated Delivery Schedule:
 - i. Engineering/Design: _____ weeks ARO
 - ii. Material Procurement: ______ weeks ARO
 - iii. Fabrication: ______ weeks ARO
 - iv. Delivery: _____ weeks ARO

OFFICIAL USE ONLY (when completed)

COMPANY NAME:



5.5

BIDDER REQUEST FOR INTEREST & PREQUALIFICATION

CRITERIA AND RESPONSE

C.	Are	there	long	lead	items	to	be	aware	of	(if	yes,	please	specify)?
D. more	Does cost eff	s your co fective, o	ompany r offers	have a shorter	u sugges delivery	ted a (i.e. "	alterna 'buy w	ite offerin /hat you m	g/pro nake"	duct)?	that off	ers an im	provement,
		lo, we wi	ill supply	y an ide	ntical or	simila	ar pro	duct.					
	□ Y	′es, we h	ave an	alternat	e offering	g. If s	so, ple	ase desci	ribe b	elow	or provi	ide as an a	attachment.
E. mitiga	Wha ite?	t risks do	o you fo	resee w	rith this p	orocui	remen	it that BN	l shou	uld be	e aware	of and po	ssibly help
Techr	nical Cr	iteria											

A. Direct Relevant Experience Documentation: If the Respondent has answered "yes" to the foregoing questions, provide a reference list of example projects over the last ten years, on the included Experience Statement, that demonstrate direct relevant project experience to support each "yes" response. Example projects should be detailed as to both the technical scope of the project and your participation in the project.

Column completion notes for the below, Experience Statement, to be completed by the Prime contractor:

- A. <u>Customer Name, Address, Contact Name and Phone No</u>.- So that we may contact as a reference as needed.
- B. <u>Work Description and Location</u>- Describe work scope and location and indicate if prime or subcontract.
- C. <u>Original/ Final Contract Value</u>- Original award value and final closeout contract value.
- D. <u>Commencement/ Completion Dates</u>- Provide starting date and actual completion (or forecast if still in progress) by month/year format (e.g., Jan 2006/ Sept 2007)

Customer Name, Address, Contact Name and Phone No.	Work Description and Location	Original/Final Contract Values	Commence/ Complete Dates

OFFICIAL USE ONLY (when completed)

Page 6 of 8



BIDDER REQUEST FOR INTEREST & PREQUALIFICATION

CRITERIA AND RESPONSE

6.0 <u>Pre-Qualification Document Checklist</u>

Companies are encouraged to use this checklist to ensure their submittal is complete.

- Interest & Prequalification Criteria and Response Package (this document)
- QA Program Table of Contents and summary *or* a copy of QA Plan

Additional supporting documentation such as brochures and company profiles.

Direct Relevant Experience Documentation (included in this document or provided as attachment)

OFFICIAL USE ONLY (when completed)

This page intentionally left blank.

OFFICIAL USE ONLY (when completed)

24590-WTP-GPP-RAPS-PS-3005 QL-MRA-PS02-00020

Page 8 of 8

Bidder Interest & Pre-Qualification 24590-



RIVER PROTECTION PROJECT – WASTE TREATMENT PLANT

ENGINEERING SPECIFICATION

FOR

Q Dual Containment Piping and CM Black Cell/Hard-to-Reach Dual Containment Piping

Content a	applicable to ALARA	A ?	🛛 Yes 🖾 No			Quality Level		
ADR No. N/A			Rev N/A	•		Q		
Retroactive applicability: Changes are retroactively applied to the full scope of the specification. There are no exisitng, procured, in fabrication, ready-to-ship, delivered, or installed items or materials. As such, there are no material impacts.						DOE Contract No. DE-AC27-01RV14136		
		Originator By: Dave Holland dholland Org Name: Plant Engineering (PENG) Placed: Jan 22, 2024	Checked By: Casey Byrne - cjbyrne Org Name: Bechtel Placed: Jan 22, 2024					
1		D.Holland	C.Byrne	N/A		J.McGrath		
0	12-15-21	J.Fugate	C.Byrne	N/A		J.McGrath		
REV	DATE	BY	CHECK	AUTHORIZAT	ION	APPROVER		
SPECIFICATION No. 24590-WTP-3PS-PS02-T0014						Rev 1		

Revision History

		Q Specification Revision Only Margin Reduced?		CM Only
Revision	Reason for Revision	YES	NO	N/A
0	Initial Issue	N/A	N/A	N/A
1	Issued for Procurement. Comprehensive Revision including, removal of the reference to year 1996 of B31.3 code for fabrication and procurement, updates to correct formatting issues, Source References, added acronym section, updated technical supply conditions for pipe, fittings, and flanges, added minimum wall thickness examination requirement, added proof test data submittal requirement, made corrections and additions to Table 3 Base Material and Weld Filler Metal Composition Requirements (in accordance with MET current specification), removed Identification as a critical characteristic in Section 4.8, made updates to the Welding Requirements Section 5.4.5 & 5.4.6, updated Inspection and Test Plan Section 6.1.2, updated Personnel Qualifications and Certification requirements Section 6.2, updated Section 7 Packing, Cleaning, Handling, and Storage Requirements, updated Marking and Labeling Requirements Section 7.6, and updated Section 10 Documentation and Submittal requirements. This is an extensive revision, the specification has been completely rewritten and therefore the changes are not identified with change bars in the right margin.			

Contents

1	Scop	De	1
	1.1	Work Included ¹	1
	1.2	Related Work Not Included ²	3
	1.3	Safety/Quality Classifications	3
2	Арр	licable Documents	3
	2.1	Codes ³	4
	2.2	Industry Standards ⁴	4
	2.3	Engineering Specifications ⁵	6
	2.4	Reference Documents/Drawings ⁶	6
	2.5	Requirement Source References	6
	2.6	Acronyms	9
3	Desi	gn Requirements	11
	3.1	Basis of Design ⁷	11
	3.2	Codes	11
	3.3	Design Conditions	11
	3.4	Pipe Wall Thickness and Branch Reinforcements	11
4	Mat	erials	11
	4.1	Piping Material Classes	11
	4.2	Technical Supply Conditions for Pipe, Fittings, and Flanges	13
	4.3	Identification Marking Requirements ^{17,18}	17
	4.4	Traceability ^{19, 20, 21}	18
	4.5	Material Commitment ²²	19
	4.6	Damaged Materials ²³	19
	4.7	Positive Material Identification (PMI) ²⁴	19
	4.8	Commercial Grade Dedication ^{27,28}	27
5	Fab	rication	33
	5.1	General ³⁰	33
	5.2	Cold Bending of Pipe ³¹	37
	5.3	Method of Construction	38
	5.4	Welding ^{32, 33, 34}	39
	5.5	Heat Treatment ³⁵	44
	5.6	Pickling ³⁶	44
	5.7	Coating and Insulation System	45

24590-WTP-3PS-PS02-T0014, Rev 1 Shop Fabrication for Q Dual Containment Piping and CM Black Cell/Hard-to-Reach Dual Containment Piping

	5.8	Marking ^{37, 38}					
	5.9	Rework of Fabricated Spools ³⁹	46				
6	Insp	ections, Examinations & Testing ^{40, 41, 42, 43}	46				
	6.1	Inspection	47				
	6.2	Personnel Qualifications and Certifications ⁴⁶	48				
	6.3	Nondestructive Examinations ⁴⁷	49				
	6.4	Shop Testing	50				
	6.5	Site Testing					
7	Pacl	king, Cleaning, Handling, and Storage Requirements ^{49, 50, 51}	51				
	7.1	General	51				
	7.2	Preservation	51				
	7.3	Cleaning	52				
	7.4	Weldable Coating	52				
	7.5	Packaging	52				
	7.6	Marking and Labeling ⁵²	54				
	7.7	Shipping Instructions	54				
	7.8	Storage	55				
8	Quality Assurance						
	8.1	Q Datasheet of ANSI/ASME NQA-1 (2000) Quality Assurance Program Requirements Clarification	56				
	8.2	NCA 3800/4000 SELLER					
	8.3	Commercial Grade Dedication ⁵³					
	8.4	Supplier Deviation Disposition Request					
9	Con	figuration Management ⁵⁴	57				
10	Doc	umentation and Submittals ^{55, 56, 57}	57				
	10.1	General					
	10.2	G-321-E Form Submittals					
	10.3	G-321-V Form Submittals	60				
Ар	pen	dices					
Ар	pendix	x A Nondestructive Examinations (NDE) Summary of Shop Welds	A-1				
App	pendix	x B Forms	B-1				
Ta	bles						

Table 1	Alloy/Element Record Requirements (Min)	23

24590-WTP-3PS-PS02-T0014, Rev 1 Shop Fabrication for Q Dual Containment Piping and CM Black Cell/Hard-to-Reach Dual Containment Piping

Table 2	PMI Requirements for Pipe, Flanges, Fittings and other Component Pieces
Table 3	Base Material and Weld Filler Metal Composition Requirements ²⁶ (Wt%)25
Table 4	Minimum Required Edge Distance Between Welds

Figures

Figure 1	Welding & NDE Matrix (Welding Procedure Application List)B-	-1
Figure 2	WRC Delta Ferrite DiagramB-	-2
Figure 3	Soft Seated Valve Installation RecordB-	-3



1 Scope

This specification defines the requirements and the work necessary for fabrication of Q dual-containment piping subassemblies (pipe spools) in accordance with the American Society of Mechanical Engineers (ASME) B31.3, Process Piping code, other codes and standards, and documents as referenced in this specification. This specification applies to Q Dual Containment piping, Q Dual Containment Black Cell/Hard-to-Reach piping, and CM Dual Containment Black Cell/Hard-to-Reach piping.

The WTP design incorporates the "black cell" concept as a key part of the facility design of the Pretreatment (PT) and the High-Level Waste (HLW) facilities. This entails locating certain equipment in the shielded cells for which no maintenance or entry is planned for the 40-year design life of the plant. Black cell (BC) Piping - all piping and tubing within a black cell up to the first weld outside the black cell.

There are areas of the WTP facilities that have components that are considered to be hard-to-reach (HtR) because of location and expected difficulty to perform repairs or maintenance which has the potential to impact mission life. HtR areas are designated as such based on R5 area radiation levels after removal of transient sources and decontamination and 1) piping and components cannot be manually or remotely maintained, and/or 2) piping and components are isolated physically by permanent plant equipment which cannot be manually or remotely removed. HtR piping extends out to the first accessible weld.

There are no valves in BC areas, and there are no inaccessible valves in HtR areas.

The term "Black Cell Weld" applies to any weld in a BC pipe or pipe spool.

The term "Hard to Reach Weld" applies to any weld in an HtR pipe or pipe spool.

Isometrics issued prior to February, 2008 are reviewed by BUYER's Engineering to determine whether a pipe spool is "BLACK CELL" or "HARD TO REACH" or not. This review and spool identification information is provided to the SELLER.

Isometric drawings issued on or after 1 February 2008 identify the pipe spools that are to be installed in black cells or hard-to-reach areas. The isometric drawing identifies each black cell or hard-to-reach pipe spool with a spool tag which will include the spool ID number and the words "BLACK CELL" or "HARD TO REACH", as appropriate. Additionally, the respective isometric will have the words "CONTAINS BLACK CELL (or HARD TO REACH, as appropriate) PIPE" in large type located just above or beside the title block.

1.1 Work Included¹

- **1.1.1** Perform the following work for piping in systems listed in the purchase order in accordance with the requirements of this specification, its appendices, and reference documents.
- **1.1.1.1** Furnish shop pipe spool sheets, extended spool sheets, or detailed drawings when specifically required by the purchase order.

- **1.1.1.2** Furnish materials as required by the purchase order including pipe, fittings, flanges, and welding materials required for fabrication
- **1.1.1.3** Fabricate pipe spools as follows:
- 1.1.1.3.1 Fabricate pipe spools including the installation of fittings, nozzles, thermowell connections, radiographic access holes and bosses, breather holes when required, and the preparation of field welding ends.
- 1.1.1.3.2 Install valves, valve actuators, and other piping components and specialty items supplied by the BUYER when required by the purchase order.
- 1.1.1.3.3 Supply and install integral structural attachments and supports as shown on piping isometric drawings, orthographic drawings, and/or detailed support drawings.
- **1.1.1.4** Perform post-weld heat treatment (PWHT) as required.
- **1.1.1.5** Perform all required testing and examinations
- **1.1.1.6** Perform all required external cleaning, coating (including Fusion Bonded Epoxy), insulation, preservation, and shipping preparation.
- **1.1.1.7** Chemically clean and/or pickle piping where indicated on the piping isometric drawing and/or the piping material class.
- **1.1.1.8** Mark pipe spools with identification numbers in accordance with the identification numbers shown on the piping isometric drawings, orthographic drawings, or other instructions furnished by the BUYER.
- **1.1.1.9** Apply color coding for material lay down purposes when required by the purchase order.
- **1.1.1.10** Furnish all required submittals and documentation in accordance with requirements as shown on the G-321-E, *Engineering Document Requirements* and G-321-V, *Quality Verification Document Requirements* forms included in the purchase order.
- **1.1.1.11** Package fabricated spools and associated materials, with packing lists, for delivery to the jobsite.
- **1.1.1.12** Perform hydrostatic leak testing of fabricated spools when specifically required by Section 6.4.1 using BUYER reviewed test procedures.
- **1.1.1.13** Perform pneumatic leak testing of fabricated spools when specifically required by Section 6.4.1 using BUYER reviewed test procedures.
- **1.1.1.14** Perform Positive Material Identification (PMI) on materials and fabrication and fabrication per the requirements of this specification. PMI is not to be performed on BUYER-furnished valves or BUYER furnished piping materials.
- **1.1.1.15** Provide pieces for field insulation work if supplier pre-insulates the pipe per 24590-WTP-3PS-PX04-T0005.

1.2 Related Work Not Included²

- **1.2.1** The following items are not included in the SELLER's scope of work:
- **1.2.1.1** Jacket piping leak testing except as noted in section 1.1.1.13.
- **1.2.1.2** Furnish and install flanged valves, specialty items, expansion joints, instrumentation (including sensing devices), relief valves, orifice plates, and flow elements.
- **1.2.1.3** Furnish and install pipe supports (except as noted in Section 1.1.1.3.3).
- **1.2.1.4** Furnish thermowells.
- 1.2.1.5 Furnish and install flow nozzles and associated hardware
- **1.2.1.6** Install fabricated piping subassemblies in the field.
- **1.2.1.7** Furnish and install thermal insulations for piping in the field.
- **1.2.1.8** Design piping and prepare piping drawings (except detailed spool drawings in accordance with Section 1.1.1.1.
- **1.2.1.9** Furnish bolting materials, gaskets, insulated flange kits, or backing rings required for field assembly.
- **1.2.1.10** Pipe wall thickness and branch reinforcement calculations.

1.3 Safety/Quality Classifications

The scope of this specification only includes *Q* Dual Containment Piping and CM Black Cell/Hard-to-Reach Dual Containment Piping safety and quality classified piping.

2 Applicable Documents

The editions of reference codes, standards and specifications shown in Appendix E of ASME B31.3 listed below, and any subsequent edition, up to and including the latest issued edition, is approved for use. This does not include Standard Specifications where the specified material grade tensile requirements have been reduced. A Supplier Deviation Disposition Request (SDDR) is required in this case.

Other codes and standards that are not references of ASME B31.3, are acceptable for use to facilitate ASME B31.3 piping fabrication or are used for pipe fabrication that is not within the scope of ASME B31.3.

2.1 $Codes^3$

ASME, 1995, *Boiler and Pressure Vessel Code*, as amended, American Society of Mechanical Engineers, New York, NY.

ASME B31.3, ASME Code for Pressure Piping, "Process Piping," American Society of Mechanical Engineers, New York, NY.

2.2 Industry Standards⁴

API 5L, 1992, *Specification for Line Pipe*, Fortieth Edition, as amended, American Petroleum Institute Publishing Services, Washington, DC.

ASME B1.20.1, 1983, *Pipe Threads, General Purpose (Inch)*, as amended, American Society of Mechanical Engineers, New York, NY.

ASME B16.5, 1988, *Pipe Flanges and Flanged Fittings NPS 1/2 Through NPS 24 Metric/Inches Standard*, as amended, American Society of Mechanical Engineers, New York, NY.

ASME B16.9, 1986, *Factory-Made Wrought Buttwelding Fittings*, as amended, American Society of Mechanical Engineers, New York, NY.

ASME B16.11, 1991, *Forged Fittings, Socket-Welded and Threaded*, as amended, American Society of Mechanical Engineers, New York, NY.

ASME B16.25, 1986, *Buttwelding Ends*, as amended, American Society of Mechanical Engineers, New York, NY.

ASME B16.28, 1986, *Wrought Steel Butwelding Short Radius Elbows and Returns*, as amended, American Society of Mechanical Engineers, New York, NY.

ASME B16.36, 1988, *Orifice Flanges*, as amended, American Society of Mechanical Engineers, New York, NY.

ASME B16.47, 1990, *Large Diameter Steel Flanges NPS 26 through NPS 60*, as amended, American Society of Mechanical Engineers, New York, NY.

ASME B16.48, 2015, *Steel Line Blanks*, as amended, American Society of Mechanical Engineers, New York, NY.

ASME B36.10M, 1985, *Welded and Seamless Wrought Steel Pipe*, as amended, American Society of Mechanical Engineers, New York, NY.

ASNT SNT-TC-1A, 2006, *Recommended Practice No. SNT-TC-1A Personnel Qualification and Certification in Nondestructive Testing*, The American Society for Nondestructive Testing, Columbus, OH.

ASTM A53-93a, *Standard Specification for Pipe, Steel, Black and Hot-Dipped, Zinc-Coated, Welded and Seamless,* as amended, ASTM International, West Conshohocken, PA.

ASTM A105-93b, *Standard Specification for Carbon Steel Forgings for Piping Applications*, as amended, ASTM International, West Conshohocken, PA.

ASTM A106-93, *Standard Specification for Seamless Carbon Steel Pipe for High-Temperature Service*, as amended, ASTM International, West Conshohocken, PA.

ASTM A123-17, *Standard Specification for Zinc (Hot-Dip Galvanized) Coatings on Iron and Steel Products*, as amended, ASTM International, West Conshohocken, PA.

ASTM A182-93b, Standard Specification for Forged or Rolled Alloy and Stainless Steel Pipe Flanges, Forged Fittings, and Valves and Parts for High Temperature Services, as amended, ASTM International, West Conshohocken, PA.

ASTM A234-92a, *Standard Specification for Piping Fittings of Wrought Carbon Steel and Alloy Steel for Moderate and High Temperature Service*, as amended, ASTM International, West Conshohocken, PA.

ASTM A312-21, *Seamless, Welded, and Heavily Cold Worked Austenitic Stainless Steel Pipes*, as amended, ASTM International, West Conshohocken, PA.

ASTM A403-93, *Standard Specification for Wrought Austenitic Stainless Steel Piping Fittings*, as amended, ASTM International, West Conshohocken, PA.

ASTM B366-20, *Standard Specification for Factory-Made Wrought Nickel and Nickel Alloy Fittings*, as amended, ASTM International, West Conshohocken, PA.

ASTM B675, *Standard Specification for UNS N08367 Welded Pipe*, as amended, ASTM International, West Conshohocken, PA.

EPRI 3002002982, 2014, Plant Engineering: Guideline for the Acceptance of Commercial-Grade Items in Nuclear Safety-Related Applications Revision 1 to EPRI NP-5652 and TR-102260, Electric Power Research Institute, Palo, Alto, CA.

EPRI TR-017218, 1999, *Guideline for Sampling in the Commercial Grade Item Acceptance Process*, Electric Power Research Institute, Palo Alto, CA.

MSS-SP-25-1978 (R1988), *Standard Marking System for Valves, Fittings, Flanges and Unions,* as amended, Manufacturers Standardization Society of the Valve and Fittings Industry, Inc., Vienna, VA.

MSS SP-95-1986, *Swage(d) Nipples and Bull Plugs*, as amended, Manufacturers Standardization Society of the Valve and Fittings Industry, Inc., Vienna, VA.

MSS SP-97-1987, Integrally Reinforced Forged Branch Outlet Fittings - Socket Welding, Threaded, and Buttwelding Ends, as amended, Manufacturers Standardization Society of the Valve and Fittings Industry, Inc., Vienna, VA.

PFI ES-3, 2009, Fabricating Tolerances, as amended, Pipe Fabrication Institute, New York, NY.

PFI ES-5, 2013, Cleaning of Fabricated Piping, as amended, Pipe Fabrication Institute, New York, NY.

PFI ES-24, 2015, *Pipe Bending Methods, Tolerances, Process, and Material Requirements*, as amended, Pipe Fabrication Institute, New York, NY.

PFI ES-31, 1992, *Standard for Protection of Ends of Fabricated Piping Assemblies*, as amended, Pipe Fabrication Institute, New York, NY.

2.3 Engineering Specifications⁵

24590-WTP-3PS-AFPS-T0001, Rev. 5, Engineering Specification for Shop Applied Special Protective Coatings for Steel Items and Equipment

24590-WTP-3PS-G000-T0002, Rev 10, Engineering Specification for Positive Material Identification (PMI) for Shop Fabrication

24590-WTP-3PS-G000-T0019, Rev. 2, Engineering Specification for Acquisition of Commercial Items and Services for Use in Safety Applications at WTP

24590-WTP-3PS-G000-T0050, Rev 1, Engineering Specification for Supplier Documentation

24590-WTP-3PS-NW00-T0002, Rev. 1, Engineering Specification for Chemical Requirements for Materials Used in Contact with Austenitic Stainless Steel and Nickel Based Alloys

24590-WTP-3PS-PX04-T0001, Rev. 1, Engineering Specification for Shop Applied Fusion Bonded Epoxy Coating for Underground Carbon Steel Pipe

24590-WTP-3PS-PX04-T0002, Rev. 5, Engineering Specification for Application of Liquid Epoxy Coating for Piping Applications

24590-WTP-3PS-PX04-T0005, Rev. 0, Engineering Specification for Installation of Insulation and Jacketing for a Pre-Insulated Piping System

24590-WTP-3PS-PX04-T0006, Rev. 0, Engineering Specification for Installation of High Density Polyethylene (HDPE) Shrink Sleeves for Piping Application

24590-WTP-PW-P30T-00001, Rev. 1, WTP End Prep Detail for Field Butt Welds

2.4 Reference Documents/Drawings⁶

N/A

2.5 Requirement Source References

1. 24590-WTP-3PS-PS02-T0001, Rev. 10, Shop Fabrication of Piping, Section 1.2

2. 24590-WTP-3PS-PS02-T0001, Rev. 10, Shop Fabrication of Piping, Section 1.3

3. 24590-WTP-3PS-PS02-T0001, Rev. 10, Shop Fabrication of Piping, Section 1.4

4. 24590-WTP-3PS-PS02-T0001, Rev. 10, Shop Fabrication of Piping, Section 1.4

5. 24590-WTP-3PS-PS02-T0001, Rev. 10, Shop Fabrication of Piping, Section 1.5

6. 24590-WTP-3PS-PS02-T0001, Rev. 10, Shop Fabrication of Piping, Section 1.5

7. 24590-WTP-DB-ENG-18-001, Rev 3, Basis of Design, Sections 9.8.4, 11.5.4, & 12.3.4

8. 24590-WTP-3PS-PS02-T0001, Rev. 10, Shop Fabrication of Piping, Section 2.1.1.5

9. 24590-WTP-3PS-PB01-T0001, Rev. 9, *Technical Supply Conditions for Pipe, Fittings, and Flanges*, Section 5.1

10. 24590-WTP-3PS-PB01-T0001, Rev. 9, *Technical Supply Conditions for Pipe, Fittings, and Flanges*, Section 5.5

11. 24590-WTP-3PS-PS02-T0001, Rev. 10, Shop Fabrication of Piping, Section 2.1.1

12. 24590-WTP-3PS-PB01-T0001, Rev. 9, *Technical Supply conditions for Pipe, Fittings, and Flanges*, Section 5.3, 5.4 & 5.8

13. 24590-WTP-3PS-PV00-T0001, Rev. 9, Technical Supply Conditions for Valves, Section 3.30

14. 24590-WTP-3PS-PS02-T0001, Rev. 10, Shop Fabrication of Piping, Section 2.1.1.1.1

15. 24590-WTP-DB-ENG-18-001, Rev 3, Basis of Design, Appendix C, Section C3.3.

16. 24590-WTP-DB-ENG-18-001, Rev 3, Basis of Design, Section 16.4.1.3

17. 24590-WTP-3PS-PS02-T0001, Rev. 10, Shop Fabrication of Piping, Section 2.1.8

18. 24590-WTP-3PS-PB01-T0001, Rev. 9, *Technical Supply Conditions for Pipe, Fittings, and Flanges*, Section 6

19. 24590-WTP-QAM-QA-06-001, Rev. 21, Engineering, Procurement, and Construction Quality Assurance Manual, Section 8.1.2.2, & 8.1.2.3

20. 24590-WTP-3PS-PS02-T0001, Rev. 10, Shop Fabrication of Piping, Section 2.1.2

21. 24590-WTP-3PS-PB01-T0001, Rev. 9, *Technical Supply Conditions for Pipe, Fittings, and Flanges,* Section 8

22. 24590-WTP-3PS-PS02-T0001, Rev. 10, Shop Fabrication of Piping, Section 2.1.6

23. 24590-WTP-3PS-PS02-T0001, Rev. 10, Shop Fabrication of Piping, Section 2.1.9

24. 24590-WTP-DB-ENG-18-001, Rev 3, Basis of Design, Section 16.4.1.8

25. 24590-WTP-3PS-G000-T0002, Rev. 10, ENGINEERING SPECIFICATION FOR POSITIVE MATERIAL IDENTIFICATION (PMI) FOR SHOP FABRICATION, Sections 1 & 3

26. 24590-WTP-3PS-G000-T0002, Rev. 10, *ENGINEERING SPECIFICATION FOR POSITIVE MATERIAL IDENTIFICATION (PMI) FOR SHOP FABRICATION*, Appendix B

27. 24590-WTP-QAM-QA-06-001, Rev.21, *ENGINEERING, PROCUREMENT, AND CONSTRUCTION QUALITY ASSURANCE MANUAL*, Section 7.1.2.12.1, 7.1.2.12.2, 7.1.2.12.3, 7.1.2.12.4 & 7.1.2.12.9

28. 24590-WTP-3PS-G000-T0019, Rev. 2, *Acquisition of Commercial Items and Services for Use in Safety Applications at WTP*, Sections 3, 4.3, 4.6, 5, 6, 7, 8, 9, 10, and 11

29. 24590-WTP-DB-ENG-18-001, Rev 3, Basis of Design, Sections 9.8.4, 11.5.4, & 12.3.4

30. 24590-WTP-3PS-PS02-T0001, Rev. 10, Shop Fabrication of Piping, Section 3.1.1

31. 24590-WTP-3PS-PS02-T0002, Rev. 2, Cold Bending of Pipe, Sections 1, 2, 3, 4, 5, 6 & 10

32. 24590-WTP-QAM-QA-06-001, Rev. 21, *Engineering, Procurement, and Construction Quality Assurance Manual*, Sections 9.1.2.1, 9.1.2.2 & 9.1.2.4

33. 24590-WTP-3PS-PS02-T0001, Rev. 10, Shop Fabrication of Piping, Section 3.1.5

34. 24590-WTP-3PS-NWP0-T0001, Rev. 2, *General Welding and NDE Requirements for Supplier Fabricated Piping*, Sections 4, 5, 6, 7, 8, 9 & 13

35. 24590-WTP-QAM-QA-06-001, Rev. 21, Engineering, Procurement, and Construction Quality Assurance Manual, Sections 9.1.2.1, 9.1.2.2 & 9.1.2.4

36. 24590-WTP-3PS-PS02-T0001, Rev. 10, Shop Fabrication of Piping, Section 1.7.2

37. 24590-WTP-QAM-QA-06-001, Rev. 21, ENGINEERING, PROCUREMENT, AND CONSTRUCTION QUALITY ASSURANCE MANUAL, Sections 8.1.2.3 & 13.1.2.2

38. 24590-WTP-3PS-PS02-T0001, Rev. 10, Shop Fabrication of Piping, Section 1.8.2

39. 24590-WTP-3PS-PS02-T0001, Rev. 10, Shop Fabrication of Piping, Section 3.1.11

40. 24590-WTP-QAM-QA-06-001, Rev.21, ENGINEERING, PROCUREMENT, AND CONSTRUCTION QUALITY ASSURANCE MANUAL, Section 10.1.2.1

41. 24590-WTP-DB-ENG-18-001, Rev 3, Basis of Design, Section 9.8.4, 11.5.4, & 12.3.4

42. 24590-WTP-3PS-PS02-T0001, Rev. 10, Shop Fabrication of Piping, SECTION 3.2

43. 24590-WTP-3PS-NWP0-T0001, Rev. 2, *General Welding and NDE Requirements for Supplier Fabricated Piping*, Sections, 11, 13

44. 24590-WTP-QAM-QA-06-001, Rev. 21, ENGINEERING, PROCUREMENT, AND CONSTRUCTION QUALITY ASSURANCE MANUAL, Section 10.1.2.4, 10.1.2.7, 11.1.2.3, & A10.1.1.4

45. 24590-WTP-3PS-G000-T0050, Rev. 1, *ENGINEERING SPECIFICATION FOR SUPPLIER DOCUMENTATION*, Section 3.2.2.1, 3.2.2.2, 3.2.2.3, 3.2.2.4, 3.2.2.9 & 3.2.20.1

46. 24590-WTP-QAM-QA-06-001, Rev. 21, *ENGINEERING, PROCUREMENT, AND CONSTRUCTION QUALITY ASSURANCE MANUAL*, Sections 2.6.6

47. 24590-WTP-QAM-QA-06-001, Rev. 21, *ENGINEERING, PROCUREMENT, AND CONSTRUCTION QUALITY ASSURANCE MANUAL*, Section 9.1.2.1

48. 24590-WTP-DB-ENG-18-001, Rev. 3, Basis of Design, Section 16.4.1.3 & 16.4.2.4

49. 24590-WTP-QAM-QA-06-001, Rev. 21, *ENGINEERING, PROCUREMENT, AND CONSTRUCTION QUALITY ASSURANCE MANUAL*, Section 13.1.2

50. 24590-WTP-3PS-PS02-T0001, Rev. 10, Shop Fabrication of Piping, Section 1.8

51. 24590-WTP-3PS-G000-T0003, Rev. 3, *Packaging, Hanging, and Storage Requirements*, Sections 4, 5, 6, 7, 8, 9, 10 & 11

52. 24590-WTP-QAM-QA-06-001, Rev. 21, ENGINEERING, PROCUREMENT, AND CONSTRUCTION QUALITY ASSURANCE MANUAL, Section 13.1.2

53. 24590-WTP-QAM-QA-06-001, Rev.21, ENGINEERING, PROCUREMENT, AND CONSTRUCTION QUALITY ASSURANCE MANUAL, Section 7.1.2.12.5, 7.1.2.12.6 & 7.1.2.12.8

54. 24590-WTP-QAM-QA-06-001, Rev. 21, *ENGINEERING, PROCUREMENT, AND CONSTRUCTION QUALITY ASSURANCE MANUAL*, Policy Q-04.1 Procurement Document Control

55. 24590-WTP-QAM-QA-06-001, Rev. 21, ENGINEERING, PROCUREMENT, AND CONSTRUCTION QUALITY ASSURANCE MANUAL, Section 17.1.2.2, 17.1.2.3 & A17.1.5.1

56. 24590-WTP-3PS-G000-T0050, Rev. 1, *ENGINEERING SPECIFICATION FOR SUPPLIER DOCUMENTATION*, Section 3.1.2, 3.1.3, 3.3.2.3, 3.3.2.4 & 3.3.2.5

57. 24590-WTP-3PS-PS02-T0001, Rev. 10, Shop Fabrication of Piping, Section 1.9

2.6 Acronyms

3D – Three Times Diameter

B&PVC - Boiler and Pressure Vessel Code

BC – Black Cell

BM - Base Metal

C of C – Certificate of Compliance / Conformance

CC - Critical Characteristics

CGD – Commercial Grade Dedication

CGI – Commercial Grade Item

CHW – Chilled Water (System)

CM – Commercial Grade

DCM – Design Compliance Matrix

GMAW - Gas Metal Arc Welding GTAW - Gas Tungsten Arc Welding HDPE - High Density Polyethylene HLW - High Level Waste HtR – Hard to Reach I.D. – Inside Diameter **ID** - Identification Min – Minimum MR - Material Requisition MT - Magnetic Particle Test MTR – Material Test Report N/A – Not Applicable NDE - Nondestructive Examination NPS - Nominal Pipe Size **OES** - Optical Emission Spectroscopy PEEK - Polyether ether ketone PFA – Perfluoroalkoxy (Teflon®) PMI - Positive Material Identification PMIV - Positive Material Identification Verified PO – Purchase Order PQR - Procedure Qualification Record PT – Liquid Penetrant Examination PTFE - Polytetrafluoroethylene **PWHT - Post-Weld Heat Treatment** QA – Quality Assurance QVD – Quality Verification Documentation Rm – Mean Radius of the Pipe RT – Radiographic Examination RTFE - Reinforced Polytetrafluoroethylene (Teflon®) SAW - Submerged Arc Welding SDDR – Supplier Deviation Disposition Request SQR – Supplier Quality Representative SW - Socket Weld t – Thickness of the Pipe (Wall) TE - Threaded End UHMWPE - Ultra-High-Molecular-Weight Polyethylene UT – Ultrasonic Examination VT - Visual Examination WFM - Weld filler Material WPAL - Welding Procedure Application List

WPS – Weld Procedure Specification WTP – Hanford Tank Waste Treatment and Immobilization Plant XRF – X-ray Fluorescence

3 Design Requirements

3.1 Basis of Design⁷

Process piping is designed in accordance with ASME B31.3, Process Piping, unless otherwise justified.

3.2 Codes

The applicable design codes are as noted on the piping class sheets. Any conflicts between this document and the applicable design codes shall be brought to the attention of the BUYER's Engineering.

3.3 Design Conditions

The service limits and the corrosion/erosion allowance referred to in each piping material class are enveloping conditions. Pressures and temperatures listed in the piping material classes shall not be exceeded except as permitted by the applicable code.

3.4 Pipe Wall Thickness and Branch Reinforcements

Each piping class is supported by BUYER's Engineering wall thickness calculations, external pressure calculations (as applicable) and if necessary, by a branch reinforcement calculation.

The commercially available pipe wall thickness, specified in the stock codes of the piping classes, meets or exceeds the minimum calculated wall thickness.

Integrally reinforced branch connections do not require a BUYER's Engineering separate calculation.

4 Materials

4.1 Piping Material Classes

4.1.1 General^{8,9}

This section defines the piping material requirements for pipes, fittings, flanges, valves, and other piping components for each noted service and the pressure and temperature limitations within which they may be used. The design and constituent components for branch connections, vents and drains, pressure, temperature, and orifice instrument connections are also covered. The piping material classes are listed in Part 2, Section 1.2 of the purchase order.

The Bechtel stock code description (identified in Attachment 2 of the referenced Piping Material Classifications), also referred to as long stock code description, for piping components shall be used for the specification and purchase of bulk piping materials. The stock code short description (shown in

Attachment 1 of the Piping Material Classifications and on the isometric bill of materials) is often truncated or otherwise shortened. As such, it shall not be used for the specification and purchase of bulk piping materials.

4.1.2 Pipe

Nominal pipe sizes (NPS) 1/8, 1/4, 3/8, 1 1/4, 2 1/2, 3 1/2, 5, and 22, and any other nominal pipe size not shown in ASME B36.10, should not be used except to match manufacturer's equipment connections or as specified on the isometric drawings. Where necessary to match a manufacturer's equipment connection, a transition piece must be used immediately adjacent to the equipment for the transition to the next acceptable pipe size.

In general, all NPS 2 and larger pipe with a schedule of standard weight and heavier has been specified as double random length and NPS 1 1/2 and smaller as single random length. The majority of pipe included in these classes is multiple stamped.

Example:

(ASTM A106 Grade B and A53 Grade B).

For carbon steel material by ASME B31.3 paragraph 323.2.2/Fig. 323.2.2, impact testing may be required.

Schedule 160 threaded pipe nipples for carbon steel (except galvanized) may be used to provide enhanced mechanical strength.

Pipe stock codes used in the piping material classes reflect pipe in the unpainted condition.

4.1.3 Flanges

NPS 24 and smaller steel flange dimensions and ratings are to meet ASME B16.5, *Pipe Flanges and Flanged Fittings NPS 1/2 Through NPS 24 Metric/Inches.*

NPS 26 through NPS 60 steel flange dimensions and ratings are to meet ASME B16.47, *Large Diameter Steel Flanges NPS 26 through NPS 60*, Series A. However, ASME B16.47 Series B to be used when needed to mate with existing equipment.

Flat faced flanges shall be used to match flat face equipment and other flat faced flanged components (e.g., copper alloy, cast iron flanges, glass reinforced plastic flanges, and for NPS 30 and larger flanges in large diameter water lines).

Orifice flanges shall be in accordance with ASME B16.36, *Orifice Flanges*, and shall be supplied in pairs, complete with jackscrews, but without bolts, gaskets, or orifice plates. Quantities on bills of material, material requisitions, etc., shall indicate sets, not individual flanges.¹⁰

Blanks, either permanent or temporary are required to have sufficient thickness per the ASME B31.3 code. Spectacle blinds (Steel Line Blanks) shall be in accordance with ASME B16.48. These spectacle blinds can be used as permanent blanks; or when required to isolate line for pressure testing, start-up, or maintenance. Gasket contact surface finishes shall be the same as the mating flanges.

4.1.4 Fittings

Butt weld fittings shall conform to the material requirements and wall thickness of the pipe with which they are used.

The wall thickness of reducing tees, concentric, and eccentric reducers shall correspond on each end to the wall thickness of the mating pipe.

Half couplings shall not be used for branch connections per ASME B31.3.

4.1.5 Branch Connections

The branch connections are limited to fittings specified in the piping material classes.

4.1.6 Bolting and Gaskets

Bolting and gaskets are supplied by others.

Machine bolts shall not be used except as jackscrews. Jackscrews shall be threaded full length. The length shall be measured from the bearing surface of the head and shall include the end point.

4.2 Technical Supply Conditions for Pipe, Fittings, and Flanges

4.2.1 General Notes^{11, 12, 13}

- **4.2.1.1** Individual line service class material will be identified on piping isometrics by indicating the appropriate piping class. Stock codes will generally be assigned for each item of shop material. Materials shall be in accordance with BUYER stock code purchase descriptions (i.e., all materials shall be purchased to the BUYER long stock code description as listed in the applicable pipe class specification). The BUYERs short description as shown on the isometric shall not be used for purchasing materials. Materials shall conform to the material specification and the applicable code. Materials shall be new and traceable to the respective material test report (MTR).
- **4.2.1.2** The following are acceptable exceptions to the long stock code description:
- 4.2.1.2.1 Seamless pipe and fittings are an acceptable substitute when welded pipe or fittings are specified. Welded pipe and fittings are an acceptable substitute for seamless pipe or fittings only when approved by SDDR.
- 4.2.1.2.2 Where material is specified as triple stamped, seamless A106/A53/API 5L Grade B, the following single stamped materials are acceptable:
 - a) A106 Grade B
 - b) Seamless A53 Grade B
 - c) Seamless API 5L Grade B
- 4.2.1.2.3 Where material is specified as double stamped, welded A53/API 5L Grade B, the following single and triple stamped materials are acceptable:

a)	Welded A53 Grade B
b)	Welded API 5L Grade B
c)	Seamless A53/A106/API 5L Grade B

- 4.2.1.2.4 Where material is specified as dual grade, Type 316/316L stainless steel, a single grade Type 316L is acceptable.
- 4.2.1.2.5 Where material is specified as dual grade, Type 304/304L stainless steel, a single grade Type 304L is acceptable.
- 4.2.1.2.6 Where BUYER's long stock code description specifies pipe to be furnished "Beveled Both Ends," pipe furnished with "Plain Ends" is acceptable.
- 4.2.1.2.7 Where BUYER's purchase order description specifies pipe to be furnished "Double Random Lengths," pipe furnished in "Single Random Lengths" is not acceptable.

4.2.1.2.8 Where ASTM B675 (*Standard Specification for UNS N08367 Welded Pipe*) N08367 Class 1 pipe with a hydrostatic test is specified, N08367 Class 2 pipe with hydrostatic test is acceptable.

- 4.2.1.2.9 External rust staining of carbon steel due to contact with austenitic stainless steel and nickel alloys is acceptable. This applies to bulk piping materials and not to shop fabricated spools.
- 4.2.1.2.10 Where ASTM B619 UNS N06022 Class I pipe is specified, ASTM B619 UNS N06022 Class II is acceptable. The hydrostatic and nondestructive electric tests specified for Class I shall also apply to Class II pipe.
- 4.2.1.2.11 Double Submerged Arc Welding (SAW) API 5L Grade B is acceptable in lieu of Electric Resistance Welded API 5L Grade B for all pipe sizes larger than NPS 24.
- 4.2.1.2.12 Where ASTM A53 Grade B is specified, ASTM A106 Grade B is an acceptable alternate for NPS 2 and smaller.
- 4.2.1.2.13 Galvanizing of ASTM A106 Grade B shall be per ASTM A123.
- 4.2.1.2.14 Taper boring of standard schedule 40S butt weld reducing fittings, per ASME B16.25, *Buttwelding Ends*, is permitted where NPS 4 and smaller schedule 40S piping transitions to NPS 6 and larger schedule 10S piping. The applicable fitting MTRs shall be annotated accordingly.
- 4.2.1.2.15 Where ASTM B366 (*Standard Specification for Factory-Made Wrought Nickel and Nickel Alloy Fittings*) grades WP6XN, WP6XN-W, or WP6XN-WX are specified in the stock code descriptions for swages manufactured to the requirements of MSS SP-95, ASTM B366 grade CR6XN seamless shall be used.
- 4.2.1.2.16 Where ASTM B366 grades WPHC22-S, WPHC22-W. or WPHC22-WX are specified in stock code descriptions for swages manufactured to the requirements of MSS SP-95, ASTM B366 material Grade CRHC22 seamless shall be used.

24590-WTP-3PS-PS02-T0014, Rev 1 Shop Fabrication for Q Dual Containment Piping and CM Black Cell/Hard-to-Reach Dual Containment Piping

- 4.2.1.2.17 Pipe size 4 in. and smaller fittings, and swages marked as ASTM A403 (*Standard Specification for Wrought Austenitic Stainless Steel Piping Fittings*) and/or ASTM A182 (*Standard Specification for Forged or Rolled Alloy and Stainless Steel Pipe Flanges, Forged Fittings, and Valves and Parts for High-Temperature Services*) with the same grade or type markings are acceptable for use on piping isometrics. No piping isometric bill of material change is required.
- 4.2.1.2.18 Certain ASTM standards list specific available lengths and do not use the terms "Double Random Lengths" or "Single Random Lengths". The lengths specified in the ASTM standard are to be used in those cases.
- 4.2.1.2.19 If purchase order calls out a specific item pipe length (e.g., 20 feet or 40 feet), it is meant as a nominal length for a single random length or a double random length, as applicable.
- 4.2.1.2.20 Pipe size 4 inches and smaller fittings, and swages marked as ASTM A234 and/or A105 with the same grade/type markings are acceptable for use on piping isometrics. No piping isometric bill of material change is required.
- 4.2.1.2.21 Class 600 SW end ball valves, when identified with the proper BUYER stock code, are acceptable to be used where the valve description, both short (listed on the isometric bill of material) and the purchase description (specified in the piping class), specifies 150 or 300 SW end ball valves.
- **4.2.1.3** When appropriate, BUYER will furnish SELLER with pipe, fittings, and/or flanges. These items will be shipped to the SELLER accompanied with the applicable documentation package(s), including MTRs. Once these items are fabricated into spools, the spools shall be shipped with the applicable documentation package including the BUYER-supplied documentation package showing the fabricator assigned material code number. The BUYER will ensure that all supplied material is in compliance with this purchase order when it is shipped to the SELLER and no further receipt inspection, other than ensuring that the paperwork agrees with the markings, checking for damage, and kick and count, will be required. Purchase order fabrication requirements and inspections, as appropriate, are to be applied, prior to shipment of the spool to the BUYER.
- **4.2.1.4** Impact testing shall apply where required per Table 323.2.2 of ASME B31.3 for piping, weld deposit and heat affected zone.
- **4.2.1.5** Threaded pipe, fittings, and flanges shall be supplied with threads in accordance with ASME B1.20.1, *Pipe Threads, General Purchase (Inch)*. Threaded galvanized pipe, fittings, and flanges shall be supplied with threads free of galvanizing.

4.2.2 Wall Thickness

4.2.2.1 For all Q piping and black cell (BC) and all hard-to-reach (HtR) area piping, each heat/lot of bulk pipe and butt weld fittings shall be examined to establish that the minimum wall thickness to establish that the minimum wall thickness meets or exceeds 87 ½% of the generally published specification or standard nominal wall thickness. This requirement applies to both CM and Q piping in black cells and hard-to-reach areas and to all associated butt weld fittings. The term "butt weld fitting" includes elbows, tees, reducers, caps,

laterals, crosses, and swages: however, it excludes pipe nipples, flanges, socket weld fittings, and integrally reinforced welded branch fittings.¹⁴

4.2.2.2 Bulk pipe and butt weld fittings require wall thickness measurements on each heat of material. Butt weld fittings include elbows, tees, reducers, caps, laterals, crosses, and swages. They exclude pipe nipples, flanges, socket weld fittings, and integrally reinforced fittings. Measurements shall be made at 4 locations approximately 90 degrees apart on all ends (i.e., Pipe-2 ends, elbows-2 ends, tees-3 ends, reducers-2 ends, caps-1 end, laterals-3 ends, crosses-4 ends, and swages-2 ends). In addition, pipe greater than or equal to 10 feet in length shall be measured at the midpoint at 4 locations approximately 90 degrees apart. Measurements shall be taken with calipers, micrometers, or ultrasonic equipment. Wall thickness reports shall be submitted (as part of the Quality Verification Documentation (QVD) package) for these components for each heat or lot used for spool fabrication. The applicable bulk pipe and butt weld fitting wall thickness report shall include the ASTM or ASME material specification number, grade/type, NPS, wall thickness/wall thickness designation/schedule number, heat number/material code, and actual measured dimensions. Wall thickness measurements taken as part of commercial grade dedication (CGD) tests and inspections may be used to satisfy this requirement.

4.2.3 Proof Tests¹⁵

4.2.3.1 ASME B16.9, Factory-Made Wrought Buttwelding Fittings, MSS SP-95, Swage(d) Nipples and Bull Plugs, and MSS SP-97, Integrally Reinforced Forged Branch Outlet Fittings - Socket Welding, Threaded and Butt-welding Ends, require adequacy of design to be established by either a fitting mathematical analyses or successful proof test data. The proof test data is the only acceptable data and is required as an engineering submittal prior to use. Mathematical Analyses is not an acceptable engineering submittal. The Buyer will review and provide permission to proceed for use via the return of the submittal with a Code 1 status. Once Code status 1 Submittal is returned, the Supplier shall record the Buyer assigned submittal number, including revision, on the first page of the applicable MTR. As stated per the applicable code/standard, it is not necessary to conduct an individual proof test of fittings with all combinations of sizes, wall thicknesses, and material. A Successful proof test on one representative fitting may represent others to the extent described in the applicable standard.

4.2.4 Additional Nondestructive Examination (NDE) for Pipe and Fitting Materials in Q Black Cell/Hard-to-Reach piping, and CM Black Cell/Hard-to-Reach piping.¹⁶

4.2.4.1 Longitudinal seam welds in seam welded Stainless Steel, Hastelloy, AL-6XN, Inconel, and Titanium pipe and fittings shall have full volumetric examination (100% RT or 100% UT) of the longitudinal seam weld. The material manufacturer is required to perform 100% full volumetric examination, either by radiographic (RT) or ultrasonic (UT) methods, as allowed by the applicable material specification, for all longitudinal seam welds in pipe and fitting material using the examination techniques in accordance with the requirements of applicable material specification. As required by the ASTM material standard, the NDE results shall be shown on the applicable material test report that is supplied with the material. For RT, the acceptance criteria shall be per UW-51 of ASME Boiler & Pressure Vessel Code Section VIII, Division 1. For UT, the acceptance criteria for single longitudinal seam welds shall be per UW-53 of ASME Boiler & Pressure Vessel Code

Section VIII, Division 1. There are no procedure submittal requirements for NDE performed by material manufacturers of pipe and fittings manufactured in accordance with a standard listed in ASME B31.3 Table A-1 or Table 326.1.

- **4.2.4.2** Where these longitudinal seam weld examinations have not been performed as required by the material manufacturer, the SELLER shall perform the examinations in accordance with the above paragraphs, including the following.
 - The SELLER RT and/or UT procedures used for the above examinations shall be submitted for BUYER's review and permission to proceed, prior to use against the scope of the applicable P.O.
 - The results of these SELLER performed examinations shall be included in the material quality verification documentation (QVD) package.

4.3 Identification Marking Requirements^{17,18}

- **4.3.1** Materials shall be new and marked with the information required by the applicable ASTM or ASME material specification at receipt. Materials shall be marked using marking materials required by the specific ASTM or ASME material specification. Fabricator may use a material code for identification and control of materials in the shop after receipt and acceptance. Marking shall be done by any permanent method that will neither result in harmful contamination or sharp discontinuities, nor infringe upon the minimum wall thickness. It is acceptable to use a rounded, low stress, vibro etch tool for this marking.
- **4.3.2** Marking materials in contact with austenitic stainless steel and nickel-based alloys shall meet the following:
- **4.3.2.1** The total leachable halogen content shall not exceed 200 ppm.
- **4.3.2.2** The total leachable sulfur content shall not exceed 400 ppm.
- **4.3.2.3** The total of low melting point metals such as lead, zinc, copper, tin, antimony, and mercury shall not exceed 1 percent. Of this, mercury should not exceed 50 ppm. These low melting metals shall not be intentionally added during the manufacturing of the marking material.
- **4.3.3** Pipe, fittings, and flanges shall be color coded if specified in the material requisition.
- **4.3.4** Fittings, flanges, and unions are marked in accordance with the applicable ASTM material specification requirements, including MSS-SP-25, *Standard Marking System for Valves, Fittings, Flanges and Unions.* This includes the omission of marking information because of small fitting, flange, or union size, or configuration limitations. This may require special tagging and packaging requirements in order to properly match and control the small fitting, flange, or union to the BUYER stock code purchase descriptions / long stock code description discussed in Section 4.2.1.1, and the material certification requirements listed in Section 4.4.1.
- **4.3.5** Valves installed by the SELLER shall be tagged after being welded into the pipe spool (this tag is in addition to the tag being furnished with the valve). Each tag shall be stamped (not

etched) with the individual unique valve identification number specified on the face of the isometric drawing (e.g., CHW-V-04558). Tags shall be ½ inch by 2 inches (min.) rectangle or 1-inch diameter (min.) stainless steel material and securely attached to the valve with 1/32-inch diameter (min.) braided stainless steel wire. Characters shall be 3/16-inch (min.) height.

4.3.6 Any piece of material not readily identifiable during fabrication shall be rejected, including other components welded thereto.

4.4 Traceability^{19, 20, 21}

4.4.1 Material traceability (e.g., identification of the item to applicable material specification, heat, batch, lot, part, or serial number or specified inspection, test, or other records) also includes transferring material identification mark(s) prior to subdividing material for piping material. Traceability is being able to trace the piping material to the applicable Material Test Report (MTR).

An MTR shall be provided for stainless steel, nickel alloy, and carbon steel materials. The material shall have a heat number that is traceable to the MTR produced by the material manufacturer. MTRs shall include, as a minimum, all data required by the applicable material specification (e.g., chemical analysis, mechanical properties, heat treatment statement, nondestructive test results, hydrostatic test results, as applicable, and heat number). A material Certificate of Compliance (C of C) shall be provided for galvanized carbon steel and other materials. MTRs or C of C for all materials, as applicable, shall be included with each shipment of the materials.

The material C of C shall contain the following information:

- a) BUYER purchase order number
- b) BUYER purchase order item number
- c) BUYER stock code number
- d) Manufacturer's name

e) The C of C identifies the specific procurement requirements met by the purchased material, equipment, or service (i.e., codes, standards, and other specifications)

f) Any approved changes, waivers, or deviations applicable to the material, service, or equipment (i.e., SDDRs).

g) The C of C shall be signed or authenticated by an authorized SELLER representative.

4.4.2 Individual spools shall have a pipe spool sheet, extended spool sheet, or detailed drawing that reflects the heat number of pressure retaining piping and items. Heat numbers/material codes of pressure retaining piping and items shall be noted on the pipe spool sheet, extended spool sheet, or detailed drawing for material traceability. A pipe spool sheet, extended spool sheet, or detailed drawing may be defined as the actual fabrication detail drawing, bill of materials, and/or traveling data table for that spool piece. Second sheets, if used, that provide spool material heat number information that is not shown on the pipe spool sheet, extended spool sheet or detailed drawing shall list the pipe spool number in order to maintain traceability. These sheets and drawings shall accompany each shipment of fabricated spools. Identify material by the specific ASTM or ASME material specification number and grade or type on

shop spool drawings and extended spool sheets, as applicable, along with heat number or material code applied to or maintained on the piping material, traceable to the MTR. Weld identification symbols shall also be recorded on detailed spool sheets and extended spool sheets, as applicable, with a cross-reference to any nondestructive examination (NDE) report numbers.

- **4.4.3** The SELLER shall ensure that the stock code shown on the isometric bill of material matches stock codes attached to or inscribed on materials prior to their installation. Fabricator may use a material code in lieu of stock code for identification and control of materials.
- **4.4.4** The SELLER is not required to submit MTRs for customer-furnished weld-in valves. Traceability of these valves to the original procurement source is the responsibility of the BUYER. The SELLER shall notify the BUYER if tags on customer-furnished valves are missing.

4.5 Material Commitment²²

The SELLER shall submit to the BUYER a complete itemized listing of materials purchased or reserved from the SELLER's inventory for each project as requested. The SELLER shall also provide, upon request, the current status of pre-bought or BUYER-furnished material.

4.6 Damaged Materials²³

- **4.6.1** Materials that have been damaged, gouged, or found to have defects affecting their form, fit, function, or encroaching on minimum wall thickness, shall not be used. Minor surface marks may be dressed, after considering the manufacturing tolerances defined in the appropriate material or technical specification and provided that the minimum wall thickness is not encroached upon. Raised metal shall be blended into surrounding base metal. Base material repair of damaged material, using welding shall not be permitted.
- **4.6.2** Particular care must be taken with flange faces. Flanges are required to meet the requirements of ASME B16.5. Flange facing imperfections outside of those specifically discussed in ASME B16.5 are not permitted. In addition, should a scratch be deeper than the flange facing serrations, the flange may be refaced, provided the height of the raised face portion of the flange is maintained. Weld repair of the damaged facing is not permitted. Radial grooves or scratches are not permitted.

4.7 **Positive Material Identification (PMI)**²⁴

PMI is required for all applicable materials used in Q BC/HtR and CM BC/HtR shop fabricated pipe spools provided under this specification²⁵.

The purpose of positive material identification (PMI) is to verify that materials are correctly supplied as specified by industry standards and WTP project specifications. Materials, and piping requiring PMI per this Specification are identified in Table 1 – PMI Requirements for Pipe, Flanges, Fittings and other Components Pieces, and Table 2. The methods/instruments listed below are acceptable. In application, they shall not be used in a "go-no-go" mode. The only acceptable objective is the positive identification, listing, and quantification of the relevant elements listed in Table 1 – Alloy / Element Record

Requirements (Min) and Table 3., Base Material and Weld Filler Metal Composition Requirements (Wt %).

PMI testing shall be done on each pressure retaining component used for a pipe spool. This shall include each individual segment of pipe, each plate, and all other pieces of base materials (e.g., flanges, fittings) and all required pressure retaining welds.

BUYER supplied bulk piping material does not require PMI. This was provided by others at time of procurement in accordance with the applicable BUYER PO requirements.

For purposes of this Specification, the following definitions are used:

<u>Alloy</u>: Metallic materials (including welding filler materials) which contain alloying elements, including but not limited to: Chromium (Cr), Nickel (Ni), Molybdenum (Mo), Copper (Cu) or Tungsten (W).

300 Series Stainless: Austenitic Stainless Steels (304L, 316L, etc.)

<u>6% Mo</u>: AL6XN, 254 SMO, etc.

The Seller shall submit to the BUYER, for review and comment prior to fabrication, procedures covering how PMI will be conducted and documented. The Seller's procedures shall include the instrument manufacturer's procedures and requirements, operator qualification/re-qualification requirements, instrument calibration method(s), calibration: frequency during testing, criteria for acceptance or rejection, material identification method, and record keeping.

4.7.1 Verification Method

4.7.1.1 Instruments used for PMI shall be of the type that shall provide quantitative, recordable, elemental composition results for positive identification of the alloy elements present.

a) PMI shall be done per the procedures outlines by the manufacturer of the PMI instrument being used. Modification of those procedures shall be approved by the BUYER.b) Each PMI instrument shall be calibrated according to the manufacturer's requirements.

- **4.7.1.2** The methods/instruments listed below are acceptable. In application, they shall not be used in a "go-no-go" mode. The only acceptable objective is the positive identification, listing, and quantification of the relevant elements listed in Table 1- Alloy / Element Record Requirements (Min) and Table 3.
 - a) Portable X-ray fluorescence analyzers. Use is limited to the following instruments or their equivalent unless prior approval is given by the BUYER:
 - TN Technologies Alloy Analyzer 9266, 9277 (The Metallurgist XR) or 9288
 - Outokumpu X-Met 840, or X-Met 880
 - Thermo Scientific Niton Metal Alloy Analyzers
 - Metorex X-Met 920, X-Met 3000TA or X-Met 3000TX Metal Analyzer
 - Innov-X Systems XT Series Analyzer
 - Innov-X Systems Alpha 6500 Series Analyzer

- Innov-X Systems Delta Model DP-2000, DS-2000, or DC-200 Analyzer
- Bruker S1 Titan Series or Tracer 5 Series Analyzer
- Hitachi X-MET8000 Series Analyzer

b) Portable optical emission analyzers. Use is limited to the following instruments or their equivalent unless prior approval is given by the BUYER, regardless of whether analysis for Carbon is required:

- SpectroPort Model TP-07 or TFO-02
- Outokumpu ARC-MET 900 or the New Spectrotest
- SpectroLab, Spectrotest and Spectrotest Jr.

c) Laser Induced Breakdown Spectroscopy (LIBS) analyzers. Use is limited to the following instruments or their equivalent unless prior approval is given by the BUYER:

- Thermo Scientific Niton Apollo handheld LIBS analyzer
- Hitachi High-Technologies Vulcan Smart+ and Vulcan Expert+
- SciAps handheld LIBS Z-200 C+ and Z-300
- d) Any other instrument will require BUYER's approval via the submittal process.
- **4.7.1.3** Optical Emission Spectroscopy (OES) may be used verify the Carbon content in 304L and 316L stainless steel materials.
- **4.7.1.4** Arc strikes, if any, shall be ground/sanded smooth.
- **4.7.1.5** Parts that are too small to be tested using an alloy analyzer are exempt from PMI testing. If such exemption is claimed, the PMI procedure shall specify the minimum part size capable of being tested.

4.7.2 Welding Consumable Control

- 4.7.2.1 PMI of completed pressure retaining welds is required as indicated in Appendix A.
- **4.7.2.2** Production "Run Off" weld test coupons may be used for chemical analysis checks of deposited metal for each heat of filler material.

4.7.3 Extent of PMI

PMI shall be performed on all bulk pipe, fitting, flanges, components, and welding consumable materials used in pipe spool fabrication prior to usage. In the case where a pipe, fitting, flange, component or welding consumable materials that did not get PMI prior to usage in spool fabrication then the component or weld shall be PMI tested at time of spool fabrication and noted as such on the PMI documentation.

BUYER supplied bulk piping material does not require PMI.

4.7.3.1 Shop Fabricated Piping

PMI is required for all pressure retaining piping and piping components, circumferential welds, and nonautogenous longitudinal welds as indicated in Table 2. For welds it is acceptable to supply the results of base metal welding consumables prior to fabrication as indicated in Table 3.

PMI is not required on autogenous welds, fillet welds, or socket welds.

Bulk Materials (straight run piping, fittings, and flanges, welding consumable materials, etc.) PMI is required as indicated in Table 2 and Table 3.

4.7.4 Identification

4.7.4.1 General

All pipe, fitting, flanges, and components used in the shop fabricated pipe spools that have been successfully subjected to the required PMI shall be marked "PMIN". The mark shall be durable and last through shop fabrication, transportation, and receiving inspection at the BUYER's facility. See Section 4.7.5 for items that do not pass PMI.

When a material is cut after PMI testing and identification, each piece shall be identified and other items/pieces with the same heat number shall be identified in accordance with Seller's written practice, including the transferring of the "PMIV" marking.

BUYER supplied bulk piping material does not require PMI. This was provided at time of BUYER in accordance with the applicable BUYER PO requirements.

4.7.4.2 PMI Marking Materials

Marking materials selected shall not cause corrosion or other harmful effects and shall be in accordance 24590-WTP-3PS-NW00-T0002, Rev 1, *Engineering Specification for Materials Used in Contact with Austenitic Stainless Steel and Nickel Based Alloys*.

4.7.4.3 Piping Materials After PMI

When it has been verified that the material has a composition consistent with the material specified, it shall be marked in accordance with the Seller's procedure using a hard marking method, an indelible ink, paint. Or other BUYER approved marking method. The marking shall be applied at one end to facilitate proper identification. The use of stamps for identification is prohibited on pipe. It is acceptable to use a rounded, low stress, vibroetch tool for this marking. To the maximum extent possible, the stamping/marking shall be located for ease of future reference/verification.

When a material is cut after PMI testing and identification, each piece shall be identified and other items/pieces with the same heat number shall be identified in accordance with Seller's written practice.

4.7.4.4 Fasteners and Small Parts

Fasteners and small parts shall be marked in accordance with the Seller's procedure using a hard marking method, an indelible ink, or paint.

4.7.5 Records of PMI

Results shall be recorded on PMI report forms, which shall indicate, as a minimum, the following for each examination:

- a) Name of inspector
- b) Date of testing
- c) Test method, including PMI instrument name and serial number
- d) Stock code number for bulk pipe, fittings, flanges, components or consumable welding materials used in pipe spool fabrication
- e) Pipe spool number for the specific pipe, fitting, flange or components which did not get PMI prior to usage in spool fabrication
- f) Quantitative analysis results for relevant elements (See Section 4.7.1.3)

In the case where a pipe, fitting, flange, components or consumable welding materials that did not get PMI prior to usage in spool fabrication, then is it required that a PMI map shall be prepared for the affected pipe spool. The map shall include applicable components and welds and include the applicable PMI test locations. This may be noted on the pipe spool sheet, extended spool sheet or detailed drawing.

Results shall be reported 100% of each heat/lot of material and shall include the following:

- a) Name of inspector
- b) Date of testing
- c) Test method, including PMI instrument name and serial number
- d) Type and number of pieces tested
- e) Acceptable composition ranges for the relevant
- f) Material identified

PMI forms shall become a part of the permanent inspection records. Seller shall submit the completed forms as part of the Final Document Package when required by the Form G-321-V in the Purchase Order.

4.7.6 Acceptance, Rejection, and Retesting Requirements

All materials tested shall be identified by the PMI instrument as being consistent with the composition of the specified material. The verification results shall fall within the chemical composition requirements of the ASTM, ASME, AWS or other applicable material specification (including the product analysis tolerances allowed therein) and allowing for the accuracy of the instrument. Any questionable PMI result shall be re-analyzed by the same or another instrument, after verification of proper surface preparation. See below, for materials and welds that fail to meet requirements on the second analysis.

The following elements shall be identified and recorded, even if the instrument does provide immediate identification (e.g., display of "316", "6 Mo", etc.):

Table 1Alloy/Element Record Requirements (Min)

Alloy	Elements
304L	Ni, Cr,C
304	Ni, Cr
316L	Ni, Cr, Mo, C
-------------------------------------	----------------
316	Ni, Cr, Mo
347	Ni, Cr, Nb
Duplex, 6% Mo, 254 SMO, AL6XN, etc.	Ni, Cr, Mo, Cu

For PMI measurement tolerance, refer to ASME Boiler and Pressure Vessel Code Section I, Nonmandatory Appendix B, Section B-8.2

Welds joining dissimilar base materials or having filler materials that do not match the base material composition may include an allowance for dilution. Acceptable composition ranges for commonly used combinations of base material and weld filler metals are included in Table 3. Other combinations, when required, shall be identified to the BUYER for approval. Please note that the only proper use of Table 3 is for assessing dissimilar welds.

4.7.6.1 If any material, component, or weld is found to be unacceptable, all other materials, components, or welds (same heat, lot, etc.) represented by that failed item shall be considered suspect. The BUYER shall be notified immediately if a component is confirmed to have failed the PMI. The Seller will then have the following options, with BUYER concurrence:

a) Scrapping/removing all materials, components, or welds represented by the test piece (all of that heat lot, etc.) and replacing with new components or filler metals, or

b) Performing 100% examination of the remainder of the represented materials, components, or welds, and replacing each item that fails the PMI check, or

c) Verifying correct chemistry by laboratory chemical analysis (OES method may be used on a case-by-case basis, specifically when the XRF method was used when unacceptable results were obtained).

- **4.7.6.2** If questionable values obtained with portable analyzers are verified by laboratory analysis, the laboratory analysis data shall be used and recorded.
- **4.7.6.3** Any item or component containing materials that have not passed the PMI shall be clearly marked as "DO NOT USE PMI FAILED" and segregated from the remainder of the stock. Included in the material segregation shall be all other materials and components with the same heat number.

ITEM – (Note 1)	VERIFICATION REQUIRED
TYPE 304, 304L, 316, 316L, & 347 Stainless	Yes – 100%, including non-autogenous
Steel Components	longitudinal pipe and fitting welds, and pressure
	retaining shop circumferential welds.
6% Mo Components	Yes – 100%, including non-autogenous
	longitudinal pipe and fitting welds, and pressure
	retaining shop circumferential welds.

Table 2 PMI Requirements for Pipe, Flanges, Fittings and other Component Pieces

24590-WTP-3PS-PS02-T0014, Rev 1 Shop Fabrication for Q Dual Containment Piping and CM Black Cell/Hard-to-Reach Dual Containment Piping

ITEM – (Note 1)	VERIFICATION REQUIRED
Duplex Stainless Steel Components	Yes – 100%, including non-autogenous
	longitudinal pipe and fitting welds, and pressure
	retaining shop circumferential welds.

Note:

1. The following items are exempted unless specifically designated for PMI by the PO:

a) C content for low grade material (see section 4.7.6) does not need to be verified if there is no weld in the fabrication process or there is a solution annealed post of welding process.

b) Non pressure-retaining parts, such as baffles, trays, tray clips, supports, pall-rings, support rings, etc.

c) Non pressure-retaining welds and sections of piping, such as drains, vents, overflows, hanger guides, gussets, etc.

d) BUYER supplied bulk piping material does not require PMI

Table 3Base Material and Weld Filler Metal Composition Requirements²⁶(Wt%)

MATERIAL	Cr	Ni	Mo	Cu	W	Notes
304L BM ³	18.0-20.0	8.0-12.0				
308L WFM ³	19.5-22.0	9.0-11.0	0.75 max			E/ER308L &
2041 337 11	10.0.22.0	0.0.10.0	0.75			
304L Welds	18.0-22.0	8.0-12.0	0.75 max			Note I
304L BM ³	18.0-20.0	8.0-12.0				
316L BM ³	16.0-18.0	10.0-14.0	2.0-3.0			
316L WFM ³	17.0-20.0	11.0-14.0	2.0-3.0			E/ER316L & LT
304L/316L	16.0-20.0	8.0-14.0	3.0 max			Note 1
Welds						
304L BM ³	18.0-20.0	8.0-12.0				
316L BM ³	16.0-18.0	10.0-14.0	2.0-3.0			
308L WFM ³	19.5-22.0	9.0-11.0	0.75 max			E/ER308L &
						LT
304L/316L	16.0-22.0	8.0-14.0	3.0 max			Note 1
Welds						
304L BM ³	18.0-20.0	8.0-12.0				
308L WFM ³	19.5-22.0	9.0-11.0	0.75 max			E/ER308L & LT
CD4MCu BM	24.5-26.5	4.75-6.0	1.75-2.25	2.75-3.25		
304L/CD4Mcu	18.0-26.5	4.75-12.0	2.0 max			Note 1
Welds						
316L BM ³	16.0-18.0	10.0-14.0	2.0-3.0			
316L WFM ³	17.0-20.0	11.0-14.0	2.0-3.0			E/ER316L &
						LT
316L Welds	16.0-20.0	10.0-14.0	2.0-3.0			Note 1
316L BM ³	16.0-18.0	10.0-14.0	2.0-3.0			
316L WFM ³	17.0-20.0	11.0-14.0	2.0-3.0			E/ER316L &
						LT
CD4Mcu BM	24.5-26.5	4.75-6.0	1.75-2.25	2.75-3.25		

24590-WTP-3PS-PS02-T0014, Rev 1 Shop Fabrication for Q Dual Containment Piping and CM Black Cell/Hard-to-Reach Dual Containment Piping

MATERIAL	Cr	Ni	Mo	Cu	W	Notes
316L/CD4Mcu	16.0-26.5	4.75-16.0	1.75-3.0			Note 1
Welds						
AL6XN BM	20.0-22.0	23.5-25.5	6.0-7.0	0.75 max		
625 WFM	20.0-23.0	58.0 min	8.0-10.0	0.50 max		E/ERNiCrMo
						-3
AL6XN Welds	20.0-23.0	25.5 min	7.0-10.0	0.75 max		Note 2
C-22 BM	20.0-22.5	Remainder	12.5-14.5		2.5-3.5	
CD4Mcu BM	24.5-26.5	4.75-6.0	1.75-2.25	2.75-3.25		
C-22 WFM	20.0-22.5	Remainder	12.5-14.5	0.5 max	2.5-3.5	E/ERNiCrMo
						-10
C-22/CD4Mcu	16.0-22.5	25.0 min	10.5-14.5		2.5-4.5	
Welds						
C-22 BM	20.0-22.5	Remainder	12.5-14.5		2.5-3.5	
C-22 WFM	20.0-22.5	Remainder	12.5-14.5	0.5 max	2.5-3.5	E/ERNiCrMo
						-10
C-22 Welds	20.0-22.5	52.0 min	12.5-14.5		2.5-3.5	Note 1
Alloy 625 BM	20.0-23.0	58.0 min	8.0-10.0			
to $304L^3$	18.0-20.0	8.0-12.0				
to $316L^3$	16.0-18.0	10.0-14.0	2.0-3.0			
625 WFM	20.0-23.0	55.0 min	8.0-10.0	0.50 max		E/ERNiCrMo
						-3
Alloy 625 to	19.0-23.0	50.0 min	8.0 min	0.50 max		
304L/316L						
Welds						
AL6XN	18.0-22.0	23.5-25.5	6.0-7.0	0.75 max		
(N08367) BM						
to 304L ³	18.0-20.0	8.0-12.0				
to $316L^3$	16.0-18.0	10.0-14.0	2.0-3.0			
625 WFM	20.0-23.0	55.0 min	8.0-10.0	0.50 max		E/ERNiCrMo
						-3
AL6XN	19.0-23.0	25.0 min	4.0-10.0			
(N08367) to						
304L/316L	-					
Welds						

BM = Base Metal; WFM = Weld Filler Material, Wt = Weight

Notes:

- 1. Acceptance is based on the combined base metal and WFM spec requirements.
- 2. Acceptance is based on WFM spec and the expected amount of dilution for molybdenum.
- 3. Carbon content of base metals (prior to being welded) and undiluted filler metal shall be verified by the applicable ASME/ASTM/AWS specification.
- 4. The measured value of alloying elements shall be within 10% of the specified level or range in the applicable materials standards.

4.8 Commercial Grade Dedication^{27,28}

4.8.1 Commercial Grade Dedication Documentation for Q Material Applications

While it is the intent that all items be procured from ASME NQA-1 qualified SELLER's and Sub-SELLER's, it is recognized that cases will likely exist where ASME NQA-1 SELLER's or Sub-SELLER's are not available. In such cases, Commercial grade components shall be commercially grade dedicated for use in safety functions material requisitions in accordance with a commercial grade dedication (CGD)procedure and acceptance plan developed under the Seller's ASME NQA-1 program. Section 4.8.3 below lists the BUYER's minimum critical characteristics for acceptance that are to be included in the procedure/plan. The CGD methodology is based on 24590-WTP-3PS-G000-T0019, *Engineering Specification for Acquisition of Commercial Items and Services for Use in Safety Applications at WTP*.

The BUYER will review the SELLER's program for verification of commercial grade items and services, including sub-supplier dedication activities. The program shall include requirements for special tests, inspections, and/or analysis, commercial grade survey of suppliers, and/or source verifications required to perform dedication activities as appropriate for the item or service provided. The SELLER shall submit for review and permission to proceed, prior to use against the scope of the applicable P.O the SELLER's commercial grade dedication procedure/plan and any sub-supplier's dedication procedures/plans. See 4.8.2 for sample size determination. CGD quality verification documentation shall be submitted with each shipment as part of the G-321-V documentation package. This documentation shall include test reports or results, inspection reports, analysis reports, commercial grade survey reports and/or source verification reports, as applicable depending on dedication method. Documentation of the commercial grade item or service dedication process shall be traceable to the item, group of items, or service. In addition, a CGD release form shall be included as the first page of Document Category Number (DCN) 33.0 in the PO G-321-V document package. The CGD release form shall list each spool identification number in the shipment. It shall also include page numbers to the CGD documentation (i.e., test reports or results, inspection reports, analysis reports, commercial grade survey reports and/or source verification reports, as applicable) for each spool that demonstrates all required CGD procedure/plan activities have been completed for each critical characteristic. The form to be used shall also be included in the submitted CGD procedure/plan.

CGD is not required for materials used for CM BC/HtR material applications.

4.8.2 Commercial Grade Dedication Sample Size Determination

The sampling acceptance process is valid only if all technical supply conditions have been met as evidenced by MTR and appropriate markings on the material. A material test report documents the product form specification for the purchased item. In addition to listing the product specification, the MTR includes product description, heat number, chemistry, heat treatment (if completed), physical properties, mechanical properties and the results of any supplemental testing if requested.

4.8.2.1 Lot Formation

Products are grouped into lots for product sampling to allow commercial dedication. A lot is comprised of one product form (i.e., pipe segment, flange, fitting or integral attachment), and one type of material (based on material specification). A lot is products from the same manufacturer and manufacturing process, preferentially, with the same heat number. If a lot needs to be formed using components from

multiple manufacturers, then refer to Section 4.8.2.4 below. Forming lots from products made by multiple manufactures is not preferred.

Markings placed on components by the original manufacturer shall be used to provide traceability to MTRs, and to forms lots, providing the marking follows the product form specification. For the purpose of this sampling strategy, re-marked products, illegible markings, or partial markings are not acceptable.

Items in a lot are accepted on the basis of the associated samples passing applicable tests for critical characteristics.

4.8.2.2 Sample Size for Testing Dimension and Seam Quality (Single Manufacturer)

When testing product, dimensions or seam quality from a single manufacturer with single heat numbers in the lot, a "Reduced" sampling plan approach following EPRI TR-017218 R1 Table 2-1 shall be used to determine sample quantities. Lot acceptance shall be based on zero defective items with respect to the defined acceptance criteria. Samples shall be randomly selected.

When testing product, dimensions or seam quality from a single manufacturer with multiple heat numbers in the lot, a "Normal" sampling plan approach following EPRI TR-017218-R1 Table 2-1 shall be used to determine sample quantities. Lot acceptance shall be based on zero defective items with respect to the defined acceptance criteria. Samples shall be randomly selected.

4.8.2.3 Sample Size for Testing Chemical Composition and Material Strength (Single Manufacturer)

When testing product chemical composition or material strength from a single manufacturer with a single heat number/production lot number/batch number in the lot, a sample size of one is acceptable when production traceability is verified by audit of the manufacturer and all intermediate supply chain entities. When testing product chemical composition or material strength from a single manufacturer with a single heat number/production lot number/batch number in the lot, EPRI TR-017218-R1 Table 2-2 shall be used to determine sample quantities. When testing product chemical composition or material strength from multiple heat numbers/production lot numbers/batch numbers in the lot, EPRI TR-017218-R1 Table 2-2 shall be used to determine sample quantities. When testing product chemical composition or material strength from multiple heat numbers/production lot numbers/batch numbers in the lot, EPRI TR-017218-R1 Table 2-1 "Reduced" sampling shall be used to determine sample quantities. Lot acceptance shall be based on zero defective items with respect to the defined acceptance criteria. Samples shall be randomly selected.

When a documented history of supplier product compliance has been established, or when a manufacturer's processes has been surveyed, WTP may analyze the data to determine if a change in sampling strategy is warranted.

4.8.2.4 Sample Size for a Lot with Components from Multiple Manufacturers

Forming a lot with components produced by multiple manufacturers should be avoided since there is reason to question whether a randomly selected sample shall represent the whole population. However, if it is necessary to form such a lot, then apply the rules of this section.

If the following conditions are true, then the sample sizes can be selected as indicated below.

MTRs are available for each component included in the lot.

Components have the manufacturers' original markings and are traceable to the MTRs received.

Components in the lot were manufactured to the same national standard that specifies the critical characteristics of interest.

All manufacturers included in the lot have a history of providing consistently conforming components of the kind included in the lot. For the purpose of this strategy, history information includes quality records indicating previous pass or failure of parts or lots as indicated on supplier quality records. For quality records to be acceptable, they must address lot acceptance based on the same critical characteristics identified in this material requisition.

When testing, dimensions or seam quality on products from multiple manufacturers, a "Tightened" sampling plan approach following EPRI TR-017218-R1 Table 2-1 shall be used to determine sample quantities. Lot acceptance shall be based on zero defective items with respect to the defined acceptance criteria. To the extent possible, sample selection shall be biased to select one sample from each manufacturer, and thereafter selected proportionately to the quantities provided by each manufacturer. If components cannot be distinguished by manufacturer, then samples shall be randomly selected.

When testing chemical composition, and material strength on products from multiple manufacturers a "Reduced" sampling plan approach following EPRI TR-017218-R1 Table 2-1 shall be used to determine sample quantities. Lot acceptance shall be based on zero defective items with respect to the defined acceptance criteria. To the extent possible, sample selection shall be biased to select one sample from each manufacturer, and thereafter selected proportionately to the quantities provided by each manufacturer. If components cannot be distinguished by manufacturer, then samples shall be randomly selected.

4.8.2.5 Material Control

For the sampling results produced through the strategy outlined above to be valid, strict material control is required. Lots shall be assigned a unique identifier. For each lot, work processes shall maintain traceability among lot identity, heat numbers, manufacturer, MTRs, sampling test results and each member of the lot. Each member of the lot shall be marked with either its heat or lot number. Any part that loses this identity cannot be used. Pieces cut from a component in the lot shall be marked with either the heat or lot number of the part from which it was cut.

4.8.3 Critical Characteristics

The following are the BUYER's critical characteristics used for the verification requirements for pipe, fittings, flanges, and plate that are commercial grade dedicated by the SELLER for pressure boundary confinement safety functions at WTP. These critical characteristics are based on the definitions listed in EPRI 3002002982, September 2014, *Plant Engineering: Guideline for the Acceptance of Commercial-Grade Items in Nuclear Safety-Related Applications Revision 1 to EPRI NP-5652 and TR-102260.* These critical characteristics are to be included in the SELLER's material commercial grade dedication procedure/plan discussed in Section 4.8.1. Sampling, if used, shall follow Sections 4.8.2.2, 4.8.2.3, and 4.8.2.4 in accordance with EPRI TR-017218, Rev. 1, January 1999, *Guideline for Sampling in the Commercial Grade Item Acceptance Process.* These critical characteristics for pipe, fittings, flanges, and plate that are already required of the SELLER. All commercial grade pipe, fittings, flanges, and plate that are commercial grade dedicated shall be manufactured to nationally recognized ASTM material specifications and ASME or MSS standards as applicable.

Note: The SELLER can choose to perform a Method 2 survey of a CM sub-supplier.

4.8.3.1 Pipe

- 4.8.3.1.1 –Dimension Outside diameter and wall thickness at each end and the wall thickness at mid-length of each pipe length in accordance with the applicable pipe ASTM material specification and the material requisition requirements to establish the ASTM required minimum wall thickness of 87-1/2% of the specified nominal wall is not violated.
- 4.8.3.1.2 Materials –

Chemical verification of materials shall be performed.

For austenitic stainless steel or high alloy pipe, the SELLER shall use a PMI procedure which meets the requirements of this Specification and has been submitted to the BUYER for review and permission to proceed, prior to use against the scope of the applicable P.O, for the applicable pipe ASTM specification for the specified material type and grade.

Tensile strength, yield strength, and elongation verification shall be performed on material final product form. The tensile strength, yield strength, and elongation properties shall meet the applicable ASTM material specification requirements for the material type and grade.

4.8.3.2 Fittings

- 4.8.3.2.1 –Dimension Dimensions shall be in accordance with the requirements of the applicable ASME B16, ASTM, or MSS standard that the particular fitting is being manufactured under. This includes the tolerances and other dimensional notes discussed and listed in the applicable standard. As a minimum the following are the dimensions to be verified.
 - Elbows centerline to end dimension
 - Tees centerline to end dimension
 - Reducers end to end dimensions
 - Swages end to end dimensions
 - Integrally reinforced fittings (These are specialty fittings which are ordered in accordance with the specific fitting manufacturer catalog requirements related to header size, branch size, end connection, and branch style) specified MSS-SP-97 dimensions.

4.8.3.2.2 Materials

Chemical verification of materials shall be performed.

For austenitic stainless steel or high alloy fittings, the SELLER shall use a PMI procedure, which meets the requirements of this specification and has been submitted to the BUYER for review and permission to proceed, prior to use against the scope of the applicable P.O, for the applicable ASTM material specification for the specified material type and grade.

Tensile strength, yield strength, and elongation verification are to be performed on material final product form. The tensile strength, yield strength, and elongation properties shall meet the applicable ASTM material specification requirements for the specified material type and grade.

Longitudinal seam welds in fittings shall be 100% volumetrically examined (RT or UT) in accordance with the SELLER requirements listed in Section 4.2.3.1.

4.8.3.3 Flanges

- 4.8.3.3.1 –Dimension The following dimensions shall be verified to be in accordance with flange dimensions listed in ASME B16.5 or B16.47 for the particular nominal pipe size, flange type, and flange pressure rating class being checked. This includes the tolerances and other dimensional notes discussed and listed in ASME B16.5. The letter listed after each dimension is those listed, for reference purposes, in ASME B16.5 for that particular dimension:
 - Outsider diameter of flange, O.
 - Thickness of flange, C.
 - Diameter of hub, X, not applicable to blind flanges.
 - Length though hub, Y, not applicable to blind flanges.
 - Thread length, threaded flange only, T, not applicable to blind flanges.
 - Bore, B, not applicable to blind flanges.
 - Depth of socket, socket welding flange only, D, not applicable to blind flanges.
 - Hub diameter beginning of chamfer welding neck only, A, not applicable to blind flanges.
 - Outside diameter of the raised face, R.
 - Heigh of raised face (not applicable to flat face flanges).
 - Diameter of bolt circle.
 - Diameter of bolt holes.
 - Number of bolt holes.

4.8.3.3.2 Materials –

Chemical verification of materials shall be performed.

For austenitic stainless steel or high alloy flanges, the SELLER shall use a PMI procedure, which meets the requirements of this specification and has been submitted to the BUYER for review and permission to proceed, prior to use against the scope of the applicable P.O, for the applicable ASTM material specification for the specified material type and grade.

4.8.3.4 Plates and Shapes

4.8.3.4.1 –Dimension – Nominal thickness shall be in accordance with BUYER drawing.

4.8.3.4.2 Materials

Chemical verification of materials shall be performed.

For austenitic stainless steel or high alloy plate, the SELLER shall use a PMI procedure, which meets the requirements of this specification and has been submitted to the BUYER for review and permission to proceed, prior to use against the scope of the applicable P.O, for the applicable ASTM material specification for the specified material type and grade.

Tensile strength, yield strength, and elongation verification are to be performed on material final product form. The tensile strength, yield strength, and elongation properties shall meet the applicable ASTM material specification requirements for the material type and grade.

4.8.3.5 Welding Filler Materials

Welding material consumables are supplied as bare wire for GTAW, GMAW, or SAW processes or with a flux for FCAW or SMAW processes. A material test report (MTR) must be provided for all welding material consumables. The MTR with actual test results is required for each heat. Alternatively, a Certified Material Test Report (CMTR) in accordance with ASME Boiler and Pressure Vessel Section II, Part C, is acceptable.

Since the method of verification used is dependent upon the configuration of filler metal used, a dedication procedure that includes each configuration of weld filler metal intended for use on WTP must be supplied for BUYER review and permission to proceed, prior to use against the scope of the applicable P.O.

The following are the BUYER's critical characteristics used for the acceptance requirements for commercially dedicated weld filler metal:

4.8.3.5.1 –Materials:

Chemistry – For reasonable assurance, all that is required is a chemistry check to ensure consistency with the applicable AWS A / ASME SFA specification. As a minimum, the SELLER can use a SELLER prepared PMI procedure, that meets the requirements of this specification. The minimum requirements for the chemical check on the weld filler material is an analysis for the concentrations of the major elements of the alloy. These tests may be performed using portable PMI techniques. X-ray fluorescence PMI techniques are appropriate for stainless steel.

For GTAW, GMAW, or SAW, the solid bare electrode metal can be chemically checked by performing PMI directly on a piece of wire to confirm the PMI specification major elements. If the chemical analysis is performed on a weld deposit, the cast button (or slug) shall be made in accordance with the applicable AWS / ASME specification. Chemical check must provide results which are between the ranges listed in the PMI procedure for the AWS A / ASME SFA specification and be in reasonable agreement with the provided MTR, allowing for the accuracy of the test / equipment used.

For SMAW and FCAW filler metal, a chemical check needs to be made on the top layer of a weld test pad. The test pad shall be made in accordance with the respective AWS

/ASME specification. This includes using a qualified welder and a qualified Weld Procedure Specification (WPS) to deposit a minimum of four passes on a plate. The plate shall have a suitable chemistry (See the applicable section of the AWS / ASME specification) that ensures electrode deposits are not contaminated by dilution. The top of the deposited weld shall be cleaned, and PMI may be used to confirm the major elements are within the specified values of the SELLER PMI procedure for the respective AWS / ASME specification and also be within reasonable agreement of the provided MTR, allowing for the accuracy of the test equipment used.

5 Fabrication

Fabricated piping shall be in accordance with ASME B31.3, and BUYER-furnished piping isometrics and specifications.²⁹

5.1 General³⁰

- **5.1.1** Use welded joint construction for all piping, except where otherwise called for on the design drawings and piping material class.
- **5.1.2** Bending to the specified radius indicated on the orthographic/isometric drawing shall be utilized in the fabrication of the dual contained pipe spools (carrier and encasement piping). Butt weld fittings may be utilized if specified on the drawing.
- 5.1.3 For dual contained piping, intermediate bend profiles shall meet the following criteria:
- **5.1.3.1** The carrier pipe shall not contact the I.D. of the encasement pipe at any point after fabrication. A minimum of 1/2 in. clearance (all around) between the carrier and encasement pipe shall be maintained for thermal expansion.
- 5.1.3.2 Misalignment between the centerlines of the carrier and encasement pipe is permitted.
- 5.1.3.3 Pipe slopes and fit up at field welds shall be maintained as indicated on the design drawing.
- **5.1.4** Make all welds full penetration butt welds except as permitted otherwise for welded attachments, socket welds, slip-on flanges, and joggle pipe jackets. Mark all welds with welder's unique identification marks. It is acceptable to use a rounded, low stress, vibroetch tool for this marking.
- **5.1.5** Dual contained pipe supports shall be fabricated and installed in accordance with the applicable pipe support and fabrication details referenced on the isometric drawings.

5.1.6 Pipe Supports

5.1.6.1 The pipe fabricator is responsible for welded attachments to pipe spools when referenced on the isometric drawings in the bill of material under FIELD MATERIALS with the following text: WELDED PIPE ATTCHMT.

- **5.1.6.2** The WELDED PIPE ATTACH applies to engineered and the standard pipe support detail drawings.
- **5.1.6.3** Welded pipe attachments include any material such as a base plate that is attached with a shop weld (e.g., pipe stanchion and base plate).
- **5.1.6.4** Welded pipe attachment vent holes shall be within the lower 20% of the vertical members unless otherwise noted on the drawing. If this criterion cannot be met, the fabricator may locate the vent holes as required for fabrication and installation. Vent holes shall be 1/8 inch with a tolerance of + 1/8 inch and 0 inches.
- 5.1.6.5 The pipe support drawing must show a shop weld for the pipe fabricator attachments.
- **5.1.6.6** Welded pipe support attachments are considered an integral component of the pipe spool. As such, welded pipe support attachments shall be coated with the same system code identified for the piping on the isometric.
- 5.1.6.7 No pipe support tags are required on the pipe spool.
- **5.1.7** Where field-supplied components are shown within an overall dimension, it shall be considered a reference dimension for both the spool fabricator and the Hanford Tank Waste Treatment and Immobilization Plant (WTP) supplier quality representative (SQR), and not a dimension used to determine the acceptability of the shop fabricated spool. The dimension(s) of the field installed component(s) will be verified during the field installation and verification process.
- **5.1.8** Flanges shall be installed with the bolt holes straddling the vertical centerline unless otherwise specified on the piping isometrics.
- **5.1.9** Threaded branch fittings on piping spools that have been hot-dipped galvanized shall have the threads "chased" after dipping to ensure that threads are free from excess galvanizing materials that would prevent the threaded joints from being made up upon installation.
- 5.1.10 Straight pipe orifice runs shall not contain welds, except at flanges. Pipe ends at orifice flanges shall be machine cut perpendicular to the pipe axis. Welds at orifice flanges shall be ground smooth inside. Pipe tap connections (if required) are to be drilled through the pipe wall and shall be smooth inside. Orifice flanges shall be drilled and tapped for jack screws. BUYER supplied drawings show the orientation of orifice flange taps and the locations of pipe taps. Unused orifice flange taps shall be supplied with appropriate plugs.

5.1.11 BUYER Supplied Valves

- **5.1.11.1** The BUYER will supply valves to the SELLER for welding into the piping spools.
- **5.1.11.2** The SELLER shall store BUYER-supplied valves indoors in a weather-tight and well-ventilated building or enclosure.
- **5.1.11.3** Valves shall be handled and supported with care to preclude damage to hand wheels and appurtenances. Lifting lugs shall be used whenever they are provided on a valve. In no case shall a valve be picked up by the valve actuator.

- **5.1.11.4** Valve stems shall be positioned in accordance with the isometric. Also, flow arrows, when present on the valve, shall align with the flow arrow shown on the isometric.
- **5.1.11.5** Valves shall be installed so there is no interference with other piping items (e.g., valve handles shall operate without interference including other valve operations).
- **5.1.11.6** SELLER shall install valves in accordance with the manufacturer's recommended instructions, and design drawings. BUYER will furnish valve manufacturer's installation and disassembly instructions for SELLER's use and reference. The SELLER shall notify the BUYER if they have not been sent the applicable valve manufacturer's installation and disassembly instructions.

To prevent damage or distortion to valve seat and disc, follow the vendor's instructions with respect to position of the valve stem and the disc during installation and welding.

- **5.1.11.7** The first time a valve is to be welded into a spool, the SQR shall witness the setup and welding to ensure compliance with the manufacturer's instructions. The SQR need only witness one valve installation for each valve instruction provided.
- **5.1.11.8** Temperatures shall be monitored at the seat location on the external surface of soft seated (UHMWPE, PTFE, RTFE, PFA, PEEK, etc.) valves that do not require disassembly for installation. Temperatures shall be monitored with pyrometers or temperature crayons. Temperatures shall not exceed the valve manufacture's recommendation. If no recommendation exists, then the limitation in the applicable Piping Class applies. This limitation is found in the "Temperature/Pressure Limits" or "Notes for Specific Stock Sizes".

Care should be taken to minimize heat input and maximize heat dissipation. This may include but not limited to weld sequencing (1/4 circumferential weld, allow to cool, repeat), use of heat absorbing paste, and/or wrapping valve with heat shield/damp cloth. These recommendations are of particular concern with respect to NPS 1 and small valves.

5.1.11.9 Valves shall be disassembled only when required by the valve manufacturer's instructions for installation. Valves shall be disassembled and reassembled, if required, in accordance with the manufacturer's disassembly and reassembly procedures.

If disassembly beyond the vendor's standard installation instruction is required, valves and actuators shall be disassembled and reassembled only after documented concurrence has been obtained from the BUYER by an SDDR.

- **5.1.11.10** The SELLER shall complete a copy of Appendix B, Figure 3 "Soft Seated Valve Installation Record" for each soft seated, weld end valve installed. Records shall be submitted with each shipment per the G-321-V form.
- **5.1.11.11** Instrument legs and plugged or capped vent and drain legs containing any of the following valves shall have a fabrication tolerance of +/- 1/2 inches.

24590-WTP-3PS-PS02-T0014, Rev 1 Shop Fabrication for Q Dual Containment Piping and CM Black Cell/Hard-to-Reach Dual Containment Piping

Stock Code	Sizo	Matarial	Manufacturor	End Conn	E to E
Stock Code	Size	Material	Wallulactulei	Ella Colli	(menes)
PVVGGZ80AN00	0.5	A105	Bonney Forge	SW	3.125
PVVGGZ80AN01	0.75	A105	Bonney Forge	SW	3.563
PVVGGZ80AN02	1	A105	Bonney Forge	SW	4.313
PVVGGZ80AN03	1.5	A105	Bonney Forge	SW	5.000
PVVGGZ80AN04	2	A105	Bonney Forge	SW	5.250
PVVGGZ80NN00	0.5	A105	Velan	SwxTE	3.130
PVVGGZ80NN00	0.5	A105	Bonney Forge	SwxTE	3.125
PVVGGZ80NN01	0.75	A105	Velan	SwxTE	3.250
PVVGGZ80NN01	0.75	A105	Bonney Forge	SwxTE	3.563
PVVGGZ80NN02	1	A105	Bonney Forge	SwxTE	4.313
PVVGGZ80NN04	2	A105	Velan	SwxTE	5.250
PVVLGZ80JJ00	0.5	A105	Powell	SW	3.150
PVVLGZ80JJ01	0.75	A105	Powell	SW	3.540
PVVLGZ80JJ02	1	A105	Powell	SW	4.330

SW = Socket Welded

TE = Threaded End

- **5.1.11.12** The SELLER shall store all in-process and completed piping spools that contain BUYER-provided valves in a weather-tight and well-ventilated building or enclosure until the time of shipment.
- **5.1.12** Certain BUYER isometric drawings are dimensioned to the centerline of a BUYER field supplied and installed welded component between the two spools. For these isometrics, the SELLER shall fabricate that portion of the spool length to the centerline of that component. Each of these spools shall have a plain end at the center point of the BUYER field installed welded component. BUYER is responsible for cutting and end preparation, as required, for field installation of this BUYER supplied component. These plain ends are not field fit up welds; as such, they do not require an additional 6 inches of pipe length.
- **5.1.13** Shop butt welds not represented on the isometric drawing(s) may be added to fabricate the spool(s) in accordance with the general configuration portrayed on the isometric drawing(s), provided that the added straight length of pipe is greater than or equal to 6 feet and additional weld(s) do not rest on pipe supports. The minimum required edge distance (D) between an integral attachment to the pipe and pipe weld or other inline components is provided in the table below. (The equation used to develop this table is sqrt(Rm x t). Rm = mean radius of the pipe and t = thickness of the pipe.)

Nominal Pipe Size (inches)	D (inches)
3 & Under	1
3.5 to 6	1 5/8
8 to 12	2 3⁄4
14 to 18	3 7/8
20 to 24	5 1/8

Table 4 Minimum Required Edge Distance Between Welds

- **5.1.13.1** The addition of butt welds is preferred over the addition of socket welds, even in a socket welded system. However, socket welded couplings may be added by the SELLER to socket welded piping systems provided that the added straight length of pipe is greater than or equal to 6 feet and the added coupling(s) do not rest on pipe supports.
- 5.1.13.1.1 It is acceptable for the SELLER to add a socket welded coupling to one side of the field weld joint when just a field weld (without a coupling) is shown on an isometric drawing for a socket welded system.

5.1.14 Dimensional Tolerances, Except for Pipe Bends

Unless otherwise indicated on the piping isometric, do not exceed the dimensional tolerances of PFI-ES-3, *Fabricating Tolerances*, for fabricated piping assemblies.

5.2 Cold Bending of Pipe³¹

Cold bending is permitted for seamless ASTM A106 Grade B, ASTM A53 Grade B, API 5L Grade B carbon steel and seamless ASTM A312 Type 304L and Type 316L stainless steel pipe. This also includes all double stamped combinations thereof and triple stamped materials. Pipe may be cold bent by rotary draw bending or ram and roll bending. Material allowances are dependent upon the equipment used and will vary with the cold bending method, material, pipe size, and bend radius. The maximum size pipe to be bent shall be 4 NPS schedule 80S for stainless steel and 6 NPS schedule 40 carbon steel with a maximum wall thickness of 0.375 inches. Bending is restricted to 3D or larger radius bends. BUYER designed piping includes a 20% thinning allowance due to bending.

The SELLER shall submit a detailed pipe bending procedure for review and receive permission to proceed prior to use, against the scope of the applicable P.O. The procedure shall include the bending method to be used, equipment descriptions, lubricants, inspections, and documentation.

5.2.1 Qualification Testing

5.2.1.1 Qualification testing shall be performed in accordance with ASME B31.3 and this specification. Sample bends shall be made using schedule 40 pipe with the minimum bend radius used in production, or smaller. Bend qualification record shall be attached to or included in submitted bending procedure.

5.2.2 Acceptance Criteria

- **5.2.2.1** Pipe bends shall be free of cracks, seams, gouges, and notches. Refer to PFI-ES-24, paragraph 6, for form tolerances related to buckling. Minor defects shall be removed by grinding or buffing provided the wall thickness is not less than 70 % of the specified wall thickness. Bends with cracks or other major defects shall be rejected.
- **5.2.2.2** Pipe bend dimensional tolerances shall be as follows:
 - Total angularity ±.5 degree
 - Plane of bend ±1.0 degree
 - Linear PFI ES-3
 - Minimum Radius and tangent PFI ES-24

- **5.2.2.3** Ovality shall not exceed 8% of the nominal outside diameter [Ovality $\% = (Dmax Dmin) \times 100 / Dnom].$
- 5.2.2.3.1 Ovality shall be checked at the bend start, finish, 1/4, mid, and 3/4 points on the first five bends with each procedure for each diameter and schedule or wall thickness on each shift.
- 5.2.2.3.2 If ovality test results are acceptable, the ovality test may be reduced to a single test at the mid-point of 10% of the bends for the remainder of the shift, as long as the ovality remains acceptable.
- 5.2.2.3.3 If any bend is found with out of tolerance ovality, all previous bends from that procedure shall be verified by the full ovality check (start, finish, 1/4, mid, and, 3/4).
- 5.2.2.3.4 Full ovality testing may again be reduced after five consecutive acceptable full examinations.
- **5.2.2.4** Wall thickness measurements shall be made along the extrados at the start, finish, 1/4, mid, and 3/4 points on the first five bends with each procedure on each shift. The measured wall thickness shall not be less than 70% of the specified nominal wall thickness. Thinning that exceeds that shall be brought to BUYER's attention via an SDDR.
- 5.2.2.4.1 If wall thickness measurements are acceptable, all subsequent bends made with the same procedure on the same shift of the same diameter and schedule, or wall thickness of pipe shall have wall thickness measurements taken at the location of greatest thinning only.
- 5.2.2.4.2 If any bend is found with out-of-tolerance thickness, all previous bends from that procedure shall be verified by the full thickness check (start, finish, 1/4, mid, and 3/4).
- 5.2.2.4.3 Full wall thickness testing may again be reduced after five consecutive acceptable full examinations.
- **5.2.2.5** Where bends shown on isometric drawings are labeled with no specific bend angle shown, the bend angle is understood to be 90 degrees. The tolerances in the previous subparagraph also apply to these bends.
- **5.2.2.6** The SELLER shall submit dimensional and tolerance acceptance verification reports in accordance with this technical specification. Applicable wall thickness measurements (as required above) shall also be included in these reports. Wall thickness reporting shall be actual dimensions. Verification reporting format may be either by tabular listing of reports or by tabular listing and sketch, as specified by SELLER's accepted procedure.

5.3 Method of Construction

Use welded joint construction for piping, except where otherwise called for on the piping isometrics and the piping material class. In general, the method of pipe spool construction is as follows:

• Shop fabricated carbon steel NPS 1 ¹/₂ and smaller pipe are socket welded while NPS 2 and larger pipe are butt-welded construction.

• Small-bore valves are socket welded up to size NPS 2 for both cases.

However, certain isometrics require all butt weld piping construction based on the service requirements.

5.4 Welding^{32, 33, 34}

5.4.1 Welding Procedures and Welder Qualifications

5.4.1.1 Welding (including tack welds) shall be performed by qualified welders in accordance with welding procedure specifications prepared and qualified in accordance with the applicable code and/or standard, and ASME BPVC, Section IX.

Welder Qualification procedure, Welding Procedure Specifications (WPS) and Procedure Qualification Records (PQR) shall be submitted to the BUYER for review and permission to proceed, prior to use against the scope of the applicable P.O. Welder personnel qualification records shall be available for review by the BUYER's SQR upon request.

It is not the BUYER's intent to require revisions to the SELLER's standard WPSs to meet these requirements, yet the limitations shall be clearly acknowledged, and the instructions employed by the welders on the shop floor. To accommodate this, Sellers may revise WPSs or issue job-only amendments or addenda to standard WPSs. Any amendment or addenda shall become part of the WPS for the purchase order and shall be issued to the welders employed in the work.

- **5.4.1.2** A Welding and NDE Matrix (Welding Procedure Application List) shall be submitted to the BUYER for review and permission to proceed, prior to use against the scope of the applicable P.O.
- **5.4.1.3** Weld repairs shall be in accordance with a Weld Repair procedure. The Weld Repair procedure shall be submitted to the BUYER for review and permission to proceed prior to start of repairs. The Weld Repair procedure shall include the following:
 - a) Method of evaluating the type and extent of the defect.
 - b) Methods used for removing the defect.
 - c) NDE conducted to ensure the defect has been removed.
 - d) WPS/PQR employed. For weld repair, either the original WPS/PQR or one originally submitted as a designated repair WPS/PQR shall be used.
 - e) NDE methods used to inspect completed weld.
 - f) Example of "Major Weld Repair" record.

Major Weld Repair records shall be submitted as part of the quality verification documentation. A major weld repair is when the depth of repair encroaches on 50% of a generally published specification or standard nominal wall thickness and any incident of breaching the pressure boundary. Major Weld Repair records shall include the following:

- a) Spool ID.
- b) Weld number/symbol.
- c) NDE conducted to ensure the defect has been removed.

- d) WPS/PQR used for weld repair.
- e) NDE methods used to inspect completed weld.

5.4.2 Weld Joint Preparation

5.4.2.1 Unless otherwise noted on the piping isometrics, field butt weld end preparations shall be in accordance with drawing 24590-WTP-PW-P30T-00001, *WTP End Prep Detail for Field Butt Welds*.

Dual contained spools requiring field welds shall be furnished with the standard 37 1/2 (± 2 1/2) degree end preparation as depicted in ASME B31.3. This applies to all field welds for the carrier and encasement pipe.

- 5.4.2.2 Misalignment (high-low) in butt joints shall not exceed 1/16" maximum.
- **5.4.2.3** Prior to welding, socket welds shall have a gap of 1/16-inch minimum to 1/8-inch maximum between the bottom of the socket and the end of the pipe.

5.4.3 Welding Process Limitations

- **5.4.3.1** Root Pass of Single Side Butt Welds without backing subject to the limitations listed, <u>only</u> the following welding processes shall be used:
- 5.4.3.1.1 Shield Metal Arc Welding (SMAW)
 - a) EXX10 or EXX11 electrodes on P-No. 1, S-No. 1, S-No. 3, P-No. 3, SP-I, SP-2, and SP-3 (refer to B31.3) steels.
 - b) Maximum electrode diameter of 3/32 inch for NPS less than $2\frac{1}{2}$ inches.
 - c) EXX10, EXX11, EXX12, and EXX13 limited to impact test temperature not lower than -20°F.
- 5.4.3.1.2 Gas Tungsten Arc Welding (GTAW)
 - a) All groove welds 1-1/2" NPS and smaller shall be welded with gas tungsten arc welding (GTAW).
 - b) Back purge two layers minimum for alloy content higher than 2-1/4 wt% Cr and 1wt% Mo.
 - c) GTAW is required for all lubrication, hydraulic and instrument piping (lines).
 - d) Filler material addition is required, except for thin wall expansion bellows and titanium.
- 5.4.3.1.3 Gas Metal Arc Welding (GMAW)
 - a) Globular transfer with mixed gas or CO₂
 - b) Pulsed spray transfer
 - c) Short circuiting transfer using solid wire and shielding gas neutral to the base metal.
 - d) Back purge two layers minimum for alloy content higher than 2-1/4 wt% Cr and 1 wt% Mo.

- e) For NPS less than 6 inches, all shop welding should be performed in the 1G rotated position whenever possible.
- **5.4.3.2** Other than Open Butt Root Passes (fill and cap passes, sockets, fillets)

Subject to the limitations listed, <u>only</u> the following welding processes shall be used:

- 5.4.3.2.1 Shielded Metal Arc Welding (SMAW)
 - a) Maximum electrode diameter 3/32 inch on socket welds.
 - b) Filler metal "F" numbers (see ASME, Section IX, QW 432).
 - F1 & F2 Not permitted on pressure retaining welds.
 - F3 Not permitted for welding on castings. Acceptable for P-No. 1 with 0.30% maximum carbon or 70 ksi maximum ultimate tensile strength. Not permitted for use when base material requires impact testing below - 20°F (minus 20°F).
 - F4 Maximum electrode diameter 3/32 inch for NPS less than 2-1/2 inches.
- 5.4.3.2.2 Gas Tungsten Arc Welding (GTAW)
 - a) All groove welds in 1-1/2-inch NPS and smaller shall be welded with GTAW.
- 5.4.3.2.3 Gas Metal Arc Welding (GMAW)
 - a) GMAW-P Pulsed spray transfer No limitations.
 - b) GMAW-S Short circuiting or globular transfer maximum base metal thickness 3/8 inch
 - c) Continuous spray No limitations transfer
 - d) For NPS less than 6 inches, all welding shall be performed in the 1G rotated position.
- 5.4.3.2.4 Flux Cored Wire Arc Welding (FCAW)
 - b) Arc shielding gas required for pressure retaining welds.
- 5.4.3.2.5 Submerged Arc Welding (SAW)
 - a) A minimum of 6-inch NPS.
 - b) Neutral flux (no active or alloy)
 - c) Run on/Run off tabs where possible.

5.4.4 Welding Filler Materials

AWS Classification "G" electrodes and filler metals shall not be used without prior acceptance by the BUYER. The specific brand name, product name, and number, used for procedure qualification (as stated on the PQR) shall also be stated on the accompanying WPS. The MTR for the specific "G" Class electrode or filler proposed, as well as the electrode or filler material manufacturer's published product

information, shall be submitted with the WPS and PQR, for review and acceptance by the BUYER. The WPS will not be reviewed without the PQR and the electrode/filler metal MTR and information.

Production welding using "G" Class electrodes and filler metals shall be restricted to the specific brand name, and product name and number stated on the BUYER-accepted PQR and WPS. Changes in the specific brand name, and product name and number shall require re-qualification and re-submittal for BUYER acceptance.

- **5.4.4.1** Except where specifically allowed, welding materials shall be selected so that the deposited weld metal is similar in chemical composition and not significantly harder or stronger than the base material. Test temperature shall be at or below the design metal temperature.
- **5.4.4.2** For filler metal, an MTR with actual test results is required. A certified material test report in accordance with ASME, Section II, Part C, is an acceptable alternative. It shall be maintained in the shop for each heat/batch in storage and made available to the BUYER's SQR upon request.
- **5.4.4.3** Filler material shall be controlled and stored in accordance with a procedure submitted to the BUYER for review and permission to proceed, prior to use against the scope of the applicable P.O. The storage, baking and drying of welding consumables (e.g., covered electrodes, flux cored electrodes, and fluxes) shall be as recommended by the manufacturer. Segregation of carbon steel, low alloys, and alloy consumables is required during storage, baking, holding, and handling (including portable ovens).

5.4.5 General Welding Requirements

- **5.4.5.1** Welding shall be protected from wind, rain and other harmful weather conditions which may affect weld quality.
- **5.4.5.2** Surfaces to be welded shall be dry and substantially free of mill scale, oil, grease, dirt, paint, galvanizing, and other contaminants.
- **5.4.5.3** Weld bevel preparations for P-No. 4 and higher alloys shall be machined or ground back to bright metal if they have been flame, arc, plasma, or laser cut.
- **5.4.5.4** Permanently installed backing rings or straps shall not be used. Temporary backing requires BUYER approval prior to use.
- **5.4.5.5** Make welds full penetration butt welds except as permitted otherwise for welded attachments, socket welds, slip-on flanges, and joggle pipe jacket.
- **5.4.5.6** Each weld shall be marked with the welder's unique symbol or number identification using low stress stamps or "vibro etch." Alternate methods shall be submitted for BUYER review and approval.
- **5.4.5.7** Welds performed by a SELLER shall be uniquely identified on vendor spool sheets and extended spool sheets, as applicable.
- **5.4.5.8** All weld joints for pressure retaining applications shall have a minimum of two passes. This includes socket welds.

- **5.4.5.9** The individual weld layer thickness for all processes shall not exceed 3/8 inch for materials less than 1-1/4-inch-thick, or 1/2 inch for greater material thicknesses.
- **5.4.5.10** Each layer of welding shall be smooth and free of slag inclusions, porosity, excessive undercut, cracks, and lack of fusion prior to beginning the next layer. In addition, the final weld layer shall be sufficiently free of coarse ripples, non-uniform bead patterns, high crown and deep ridges to permit the performance of any required inspection. All arc strikes, starts, and stops shall be confined to the welding groove or shall be removed by grinding. Welds containing cracks shall be locally repaired.
- **5.4.5.11** Where joints are welded from both sides, the first pass shall be back chipped, ground or arcgouged to sound metal before welding the second side. This requirement shall be stated on the WPS. (Except on welds that are not full thickness)
- **5.4.5.12** Peening shall not be used. The use of pneumatic tools or steel shot for slag removal is not considered peening.
- **5.4.5.13** Welding sequences in which a layer is not completed before the next layer is started (block welding) shall require approval by the BUYER.
- **5.4.5.14** Except for root passes, vertical welding shall be vertical up unless approved otherwise by BUYER for each specific application.
- **5.4.5.15** Gas mixture combinations for weld shielding and purging shall be documented and approved as part of the WPS. Any change in gas mixture requires revision of the WPS and may require requalification. Welding consumables shall be qualified using the gas mixture of the WPS.
- **5.4.5.16** Welding of stainless steel and nickel based materials shall be physically separated from carbon steel to ensure contamination by tools, grinding dust, etc. does not occur.
- **5.4.5.17** All slip-on flanges shall be double welded in accordance with ASME B31.3, paragraph 328.5.2, unless directed otherwise by design document(s).
- **5.4.5.18** When welding titanium, each bead and the adjacent metal shall be cleaned to remove all surface discoloration prior to deposition of the next bead. The Final weld surface may have intermittent iridescent, straw-colored oxides.

5.4.6 Preheat and Inter-pass Temperatures

- **5.4.6.1** Preheat temperature shall be in accordance with the applicable codes and Welding Procedure Application List (WPAL), except that code recommended minimum preheat temperatures shall be mandatory. Preheat requirements shall apply to all welding, including tack welding and welding of temporary attachments. Preheat requirements also apply to all thermal gouging and cutting operations except for P-No. 1 steels. For GTAW root pass welding <u>only</u>, reduced preheat levels as permitted by the applicable code maybe used. Preheat shall be maintained a minimum of 3 inches on either side of the joint.
- **5.4.6.2** The interpass temperature shall not exceed:

- 350°F for austenitic stainless steels and nickel based alloys
- 800°F for hard facing of low carbon austenitic stainless steels
- 300°F for superaustenitic stainless steels, with heat input not exceeding 50 KJ/in
- **5.4.6.3** For welds requiring preheating, the weld joint shall be completed with no intermediate cooling except that cooling under an insulating blanket is permitted provided at least applicable code required minimum joint depth has been filled, and provided that applicable documents do not specify more stringent requirements.
- **5.4.6.4** Preheat shall be determined by temperature indicating crayons, contact pyrometers or other equally suitable means. Temperature indicating crayons used on austenitic stainless steels and nickel base alloys shall not cause corrosive or other harmful effects. It is the SELLER's responsibility to determine suitable brands that may be used. This information shall be made available to the BUYER's representative on request.
- **5.4.6.5** If oxy-fuel torches are used for preheating, the torch tip shall be appropriate for the work (i.e., a "rosebud," not a cutting or welding tip).

5.5 Heat Treatment³⁵

5.5.1 Postweld Heat Treatment

- **5.5.1.1** If Postweld Heat Treatment (PWHT) is required, it shall be performed in accordance with procedures that have been reviewed and provided permission to proceed by the BUYER before use against the scope of the applicable P.O. The procedure shall describe cleaning requirements, heating and cooling rates, type of heat treatment atmospheres, thermocouple locations, type of heating equipment, etc. The PWHT requirements for piping shall be in accordance with the applicable code and project specifications. The heat treatment temperature range and holding time at temperature shall be as specified in the WPAL.
- **5.5.1.2** Direct impingement by furnace burner flames is not permitted.
- **5.5.1.3** Only resistance, induction, furnace or quartz lamp heating methods are permitted. Exothermic heat treatment is prohibited.
- **5.5.1.4** PWHT cycles shall be recorded on print out charts. Thermocouples shall be independently recorded. When PWHT is performed outside of a controlled furnace environment, each weld shall have at least one thermocouple attached for temperature monitoring. The heat treat chart shall be submitted to the BUYER with each shipment.

5.6 Pickling³⁶

When pickling of carbon steel piping is required by the design drawing and/or piping material class, the following shall apply:

- **5.6.1** Degrease to remove grease, dirt, and other contaminants.
- **5.6.2** Acid clean to remove mill scale, rust, and other foreign substances on the inside of the pipe, without loss of base metal.

- **5.6.3** Neutralize, rinse, and completely remove pickling solution.
- **5.6.4** Dry thoroughly. The interior of the pipe shall be free of mill scale, rust, acid, and other foreign materials.
- **5.6.5** Coat the inside of pipe with one of the following rust preventatives (or BUYER-approved equal):
- **5.6.5.1** Lectra Shield or SP-400, CRC Industries.
- **5.6.5.2** Mobilarma 247, Mobil.

5.7 Coating and Insulation System

- **5.7.1** Coating and insulation with High Density Polyethylene (HDPE) jacketing requirements may apply to carbon steel or stainless-steel dual contained piping.
- **5.7.2** Coatings shall be in accordance with specifications 24590-WTP-3PS-AFPS-T0001, 24590-WTP-3PS-PX04-T0002.
- **5.7.3** Insulation and jacketing for underground piping shall be in accordance with 24590-WTP-3PS-PX04-T0005.
- **5.7.4** HDPE shrink sleeves are installed per 24590-WTP-3PS-PX04-T0006.
- **5.7.5** Shop welds, located on the carbon steel or stainless steel encasement piping, shall not be coated or insulated prior to the pneumatic test being completed for that particular spool.
- 5.7.6 Coat the exterior surface of above ground piping in accordance with specification 24590-WTP-3PS-AFPS-T0001 as follows: Prepare surfaces in accordance with Appendix D for the supplied piping. Apply the coating system as listed on the individual isometric drawing. Appendix C, Table 2 defines the coating system details. Appendix C, Table 1 defines approved coating materials and thicknesses. Submittal requirements are defined in the Material Requisition (MR) on Form G-321-E and in coating specification section 9.0. Submittals include, but may not be limited to, personnel training records, coating application procedures, a listing of specific products to be applied including submittal of a Product Identity Certification Record (Appendix F) for each batch of coating material used, and daily inspection record of all work completed (Ref. Appendix G).
- **5.7.7** Coat the exterior surfaces of buried carbon steel in accordance with specifications 24590-WTP-3PS-PX04-T0001 for straight sections of pipe and/or 24590-WTP-3PS-PX04-T0002 for fittings, weld areas, and straight pipe (shop and field). Do not paint the gasket seating surface of flange faces.
- **5.7.8** Certain fabricated spools identified on the piping isometrics shall be furnished complete with a coating, on the exterior of the carbon steel encasement pipe, plus a shop applied insulation system installed in accordance with 24590-WTP-3PS-PX04-T0005 and as applicable 24590-WTP-3PS-PX04-T0006.

5.8 Marking^{37, 38}

- **5.8.1** The fabricator shall apply the spool piece mark number, indicated on the isometric/orthographic drawing, to all spools. Mark spools of 15 feet or less in length once on one end. Mark spools greater than 15 feet in length once on each end, with markings appearing on opposite sides of the spool. Piece marks shall be stenciled or legibly hand lettered in letters not less than 2 inches high (when possible), with a visible weatherproof marker. Piece marks shall not be placed on end caps or flange protectors. Piece marks shall be readable from a reasonable distance.
- **5.8.1.1** For isometric drawings, the encasement pipe spool shall be marked to indicate both the carrier and the encasement piece mark number. This will assist in receiving of the dual contained spools.

Example: Spool pc. Mk.= LAW-LCP-PB01368001-A/LAW-LCP-PB01368002-A

5.8.1.2 For orthographic drawings, the encasement pipe spool shall be marked indicating only one piece mark number as indicated (the carrier and encasement spools will have the same number).

Example: Spool pc. Mk.= BOF-TCP-PH00118-A

- **5.8.2** Paint markings shall be at least 4 inches away from a surface to be welded.
- **5.8.3** A chemical analysis report for marking materials shall be submitted for BUYER review and permission to proceed prior to use against the scope of the applicable P.O.
- **5.8.4** Marking materials shall meet the requirements of Section 4.3 of this specification.

5.9 Rework of Fabricated Spools³⁹

5.9.1 Reworked spools being shipped to the jobsite shall be provided with a Certificate of Compliance (C of C) stating the spools identified are the same spool number(s) originally delivered and received. The C of C shall state that the re-worked spools are in strict accordance and fully comply with the purchase order and all procedures and specifications. The C of C shall also state the re-work performed on each spool.

If additional materials are added, or if additional documentation is provided (NDE reports, MTRs, etc.) a complete documentation package for that spool shall be furnished.

6 Inspections, Examinations & Testing^{40, 41, 42, 43}

Inspection, examination, and testing shall be in accordance with ASME B31.3, this specification, and other governing codes and standards, as applicable.

The SELLER is responsible for inspection, NDE, and testing of piping furnished under this specification.

6.1 Inspection

BUYER's representative shall be provided free access to the SELLER's and SELLER's subcontractors facilities, to witness, inspect, and report progress of work. This includes manufacture, fabrication, heat treatment, assembly, examination, and testing of piping. They have the right to audit any examination, to inspect the piping using any examination method specified by the engineering design, and to review all certifications and records necessary to ensure piping conforms to all applicable examination and testing requirements of the Code and of the engineering design.

6.1.1 SELLER Inspections

An organized process shall be followed for the purpose of determining that specified requirements (dimensions, properties, performance results, etc.) are met. Procedures for dimensional inspections and wall thickness measurements shall be submitted for BUYER review and permission to proceed.

6.1.2 Inspection and Test Plan^{44, 45}

- **6.1.2.1** SELLER shall submit a detailed description of the inspections and tests planned during the receipt, fabrication, and conformance verification activities for permission to proceed, prior to use against the scope of the applicable P.O. This shall include characteristic to be inspected, methods of inspection, identification of the organization responsible for performing the inspection, acceptance criteria (which is measurable and verifiable), measuring and test equipment to be used (ensure equipment is of the proper type, range, accuracy and tolerance to accomplish the intended function), identification of sample plan if applicable, and the identification of BUYER and SELLER witness and hold points. When sample plans procedures are used, they shall be based on valid statistical methods.
- **6.1.2.2** If an outside accredited lab is selected to perform the inspection / test, objective evidence of their accreditation shall be provided in the plan.
- **6.1.2.3** Sequential hold and notification points for inspection and tests to be performed, including BUYER designated source inspection/witness notification point in accordance with the BUYER's quality surveillance plan included in the material requisition shall be provided in the plan.
- **6.1.2.4** The inspection and test report form to be used shall be provided in the plan.
- **6.1.2.5** Other requirements as required by codes and standards, or other invoked specifications, shall be provided in the plan.

Appropriate inspection records shall be established, maintained and, as a minimum, identify the following:

- Item inspected (Spool ID or Piece part #).
- Description of inspection or test performed.
- Type of observation or method of inspection.
- Date of inspection.
- Identification of the organization responsible for performing the inspection

- Name or unique identifier of the inspector who documented, evaluated, and determined acceptability.
- The inspection criteria, sample plan, or reference documents (including revision levels) used to determine acceptance. General note drawings, detail drawings, consensus standard codes, and vendor documents do not need to be identified on the record provided those documents are traceable from the documents that are identified on the record.
- Results indicating acceptability of characteristic inspected.
- Measuring and test equipment used to document inspection results during the inspection, including the identification number and the most recent calibration date and calibration due date.
- Reference to information on action taken in connection with nonconformances.

Test procedures shall include or reference the test configuration and test objectives. Test procedures shall also include provisions for assuring that prerequisites and suitable environmental conditions are met, adequate instrumentation is available and used, appropriate tests and equipment are used, and necessary monitoring is performed. Prerequisites shall include the following, as applicable: calibrated instrumentation, appropriate equipment, including accuracy requirements, trained personnel, condition for test equipment, and completeness of item to be tested, suitable environments conditions, and provisions for data acquisition and storage.

6.2 Personnel Qualifications and Certifications⁴⁶

6.2.1 All NDE, including visual examination and UT thickness measurements, shall be performed by personnel certified in accordance with American Society for Nondestructive Testing Recommended Practice SNT-TC-1A, *Personnel Qualification and Certification in Nondestructive Testing*.

Personnel who perform NDE (including RT, magnetic particle testing, UT, UT thickness measurements, liquid penetrant testing, electromagnetic testing, neutron RT, leak testing, acoustic emission testing, and visual testing) to specified acceptance criteria shall be qualified to a "Written Practice" (or procedure) meeting the requirements of the American Society for Nondestructive Testing Recommended Practice No. SNT-TC-1A. This "Written Practice" must meet all the requirements from the 2006 edition or later of SNT-TC-1A with the following supplement:

This "Written Practice" must meet all of the requirements from any one given edition of SNT-TC-1A and its applicable supplements.

The re-certification interval for Level III NDE personnel shall not exceed 5 years.

- **6.2.2** The SELLER's NDE "Written Practice" (or procedure) shall be submitted to the BUYER for review and permission to proceed, prior to use against the scope of the applicable P.O.
- **6.2.3** The SELLER NDE personnel certifications shall be available to the BUYER's SQR upon request.

6.3 Nondestructive Examinations⁴⁷

6.3.1 Perform and evaluate examinations in accordance with procedures and acceptance standards prepared in accordance with the ASME B31.3 and/or standard, and the ASME, Section V.

Procedures for each method of examination shall be submitted to the BUYER for review and permission to proceed, prior to use against the scope of the applicable P.O. No sub-supplier shall perform NDE work without prior submittal of the sub supplier's NDE procedure and BUYER's review and permission to proceed, prior to use against the scope of the applicable P.O.

NDE reports shall be submitted with the shipment as part of the quality verification documentation.

Radiography film with technique and reader sheets shall be submitted as part of the G-321-E submittals in accordance with the material requisition.

Per ASME B31.3, paragraph 344.2.2, records of individual visual examinations are not required, except for those of in-process examinations.

- **6.3.2** WPAL shall identify the extent of NDE per individual pipe class to address the NDE requirements listed in Sections 6.3.3.
- **6.3.3** The extent and acceptance criteria are as listed in ASME B31.3, para 341.4.1 and Table 341.3.2 respectively, for normal fluid service conditions with the following additions. These requirements are summarized in Appendix A of this specification. Weld repair shall be examined in accordance with the requirements used for the original weld.
- **6.3.3.1** Where RT is allowed or specified for shop welds, the radiographic acceptance criteria for normal fluid service apply, except that incomplete penetration shall not be permitted.
- **6.3.3.2** Where liquid penetrant examination or magnetic particle examination is allowed or specified, no cracks shall be permitted.

6.3.4 Piping Outside BC and HtR Weld NDE and Inspection Requirements

- **6.3.4.1** Shop girth welds require full visual examination in accordance with the requirements of ASME B31.3, para 341.4.1(*a*), and NDE examination in accordance with the requirements of ASME B31.3, para 341.4.1(*b*). If the RT examination cannot be performed then the selected welds shall be examined by in-process examination in accordance with ASME B31.3, para 344.7, including PT of the root pass and the final pass in accordance with ASME B31.3, para 344.7.1(e).
- **6.3.4.2** Integrally reinforced forged branch fitting connection welds, welded to the main piping run shall have 100% visual examination in accordance with the requirements of ASME B31.3, paragraph 341.4.1. If these welds are selected for RT then the selected welds shall be examined by in-process examination, in lieu of volumetric examination, in accordance with ASME B31.3, para 344.7, including PT of the root pass and the final pass in accordance with ASME B31.3, para 344.7.1(e).

- **6.3.4.3** Fillet welds including socket welds, integral support welds, and non-pressure and non-load bearing piping attachment welds (e.g., cathodic protection clips) welds require 100% visual examination and NDE in accordance with ASME B31.3, paragraph 341.4.1.
- **6.3.4.4** Full penetration welded stub-in lateral branch connection fittings used on gravity drain piping require 100% VT and 100% PT of final pass only.

6.3.5 Piping Inside BC and HtR Weld and Inspection Requirements⁴⁸

- **6.3.5.1** Shop girth welds require 100% full visual examination in accordance with the requirements of ASME B31.3, para 341.4.1(*a*), and 100% NDE examination in accordance with the requirements of ASME B31.3, para 341.4.1(*b*). If this RT examination cannot be performed then the selected welds shall be examined by in-process examination in accordance with ASME B31.3, para 344.7, including PT of the root pass and the final pass in accordance with ASME B31.3, para 344.7.1(e).
- **6.3.5.2** Integrally reinforced forged branch fitting connection welds, welded to the main piping run shall have 100% visual examination in accordance with the requirements of ASME B31.3, paragraph 341.4.1 In lieu of 100% volumetric examination these welds shall be examined by in-process examination in accordance with ASME B31.3, para 344.7, including PT of the root pass and the final pass in accordance with ASME B31.3, para 344.7.1(e).
- **6.3.5.3** Fillet welds including socket welds, integral support welds, and non-pressure and non-load bearing piping attachment welds (e.g., cathodic protection clips) welds require 100% visual examination and NDE in accordance with ASME B31.3, paragraph 341.4.1.

6.4 Shop Testing

6.4.1 Leak Testing

- **6.4.1.1** Shop hydrostatic leak testing shall be performed in accordance with ASME B31.3 by the SELLER on inner piping prior to the jacket piping being completed. Potable water shall be used for the leak test performed by the SELLER. Test pressure shall be determined based on the applicable pipe class. All water shall be drained and spools dried within 72 hours of testing. Seller shall submit their hydrostatic leak test procedure for BUYER's review and permission to proceed. Test results shall be submitted as part of G-321-V.
- **6.4.1.2** Shop pneumatic leak testing shall be performed in accordance with ASME B31.3 by the SELLER on fabricated jacket piping insulated and/or wrapped in the SELLER's shop. Test pressure shall be in determined based on the applicable pipe class. Seller shall submit their pneumatic leak test procedure for BUYER's review and permission to proceed, prior to use against the scope of the applicable P.O. Test results shall be submitted as part of G-321 V. Perform leak testing prior to coating (if applicable) and insulating and/or wrapping.
- **6.4.1.3** Welding of integral supports shall be avoided in previously hydrostatically or pneumatically leak tested piping. BUYER's engineering shall be consulted immediately when this situation occurs.

6.5 Site Testing

After installation, all fabricated spool assemblies will be leak tested by others in accordance with the applicable code. Should any fabrication defect be detected in shop welds during field testing, the defect shall be repaired by the SELLER or BUYER in accordance with the Purchase Order Terms and Conditions.

7 Packing, Cleaning, Handling, and Storage Requirements^{49, 50, 51}

This section covers general requirements for packaging, shipping preparation, handling, and storing of components procured for use at the WTP.

7.1 General

- **7.1.1** The handling, storage, cleaning, packaging, and preservation of items shall be controlled to prevent damage or loss and to minimize deterioration. These activities shall be conducted in accordance with established work and inspection instructions, drawings, specifications, or other pertinent documents or procedures specified for use in conducting the activity. SELLER shall submit their general procedure for handling, storage, and preparation for shipments. It shall also include a section addressing the requirements in 7.8.3, 7.8.4, 7.8.5 and 7.8.6 of this specification.
- 7.1.2 Material and equipment (cargo) must be prepared for shipment such that it can be secured for protection against shifting and falling cargo as specified in 49 CFR 393 Subpart I, *Protection Against Shifting and Falling Cargo*. For cargo that cannot be crated or that requires direct securement to a transport vehicle, the Seller shall ensure that the cargo is designed or packaged with sufficient and adequate attachment points to limit movement in accordance with the requirements of 49 CFR 393 Subpart I.
- **7.1.3** Materials in contact with austenitic stainless steel shall meet the chemical requirements specified in 24590-WTP-3PS-NW00-T0002.
- **7.1.4** Prior to shipment of equipment made of stainless steel or nickel based alloys, the SELLER shall perform a visual inspection of all surfaces in contact with wooden supporting structures looking for evidence of surface corrosion. A nonconformance report (NCR) shall be created to resolve any corrosion issues identified.

7.2 Preservation

- **7.2.1** Pipe spool surfaces shall be cleaned and coated as specified on the isometric and purchase order.
- 7.2.2 Piping spools shall be prepared for long-term (greater than 6 months) storage.

7.2.3 WTP has limited capacity for storage of equipment and materials in climate-controlled areas. Equipment and materials should be prepared for outside storage. Indoor storage or climate-controlled storage should be specified only when absolutely necessary.

7.3 Cleaning

- **7.3.1** Perform cleaning after fabrication has been completed. Cleaned piping shall be free of loose rust or mill scale, blisters, grease, sand, oil, dirt, and other foreign materials.
- **7.3.2** Fabricated spools shall be cleaned in accordance with the standard cleaning method described in PFI ES-5, *Cleaning of Fabricated Piping*.
- **7.3.3** After cleaning, blow dry the interior surfaces of all piping with clean, filtered, oil-free air. There shall be no standing liquids (i.e., water) on the interior of piping.
- **7.3.4** The gasket seating surface of flange faces shall be cleaned and coated with a one of the following rust preventives:
- **7.3.4.1** Grease (manufacturer's standard).
- **7.3.4.2** Lectra Shield or SP-400, CRC Industries.
- **7.3.4.3** Mobilarma 247, Mobil.
- 7.3.4.4 Any preservative listed in specification 24590-WTP-3PS-AFPS-T0001.

7.4 Weldable Coating

Apply a preservative coating of Bloxide from Tempil or Deoxaluminite from D.C. Sales Engineering (or BUYER-approved equal) weldable primer on field weld end preparations to carbon steel surfaces including coating the hold back area. Overlap onto the coating is acceptable. Weldable primers containing zinc are not acceptable.

7.5 Packaging

This section contains requirements for packaging of items for protection against corrosion, contamination, physical damage, or any effect that would lower the quality or cause the components to deteriorate during the time they are shipped, handled, and stored.

- **7.5.1** Package design requirements shall be used for environmental protection to avoid the deleterious effects of shock and vibration, to control temperature and humidity within specified limits, or for any other special requirements.
- **7.5.2** Items shall be inspected for cleanliness immediately before packaging. Dirt, oil, residue, metal chips, or other forms of contamination shall have been removed by cleaning methods approved by the BUYER. Any trapped water shall have been removed and the component dried.
- 7.5.3 Components not immediately packaged shall be protected from contamination.

24590-WTP-3PS-PS02-T0014, Rev 1 Shop Fabrication for Q Dual Containment Piping and CM Black Cell/Hard-to-Reach Dual Containment Piping

- **7.5.4** Items requiring protection from damage from water vapor, salt air, dust, dirt, and other forms of contamination penetrating the package shall be packaged with a barrier. The barrier shall be based on the type, size, and weight of equipment and shall not be easily damaged by puncture, abrasion, weathering, cracking, temperature extremes, wind conditions, and the like. Barrier and wrap materials shall be noncorrosive and not harmful to the component protected. Also, barrier and wrap materials shall not readily support combustion. Vapor proof barrier materials used with desiccants constitute another preservation system that protects against potential damage by water vapor condensate. Adequate protection shall be provided against mechanical damage and atmospheric corrosion in transit and, for equipment suitable for outside storage, for at least 12 months outdoor storage at the job site prior to installation.
- **7.5.4.1** Waterproof barrier material Waterproof barrier material shall be resistant to grease and water; it shall protect items from airborne and windblown soils.
- **7.5.4.2** Components that can be damaged by condensation trapped within the package shall be packaged with approved desiccant inside the sealed waterproof or vapor proof barrier or an equivalent method. Signs indicating that desiccant has been installed shall be conspicuously located on the exterior of the packaging, adjacent to where the desiccant has been installed. A humidity indicator shall be included in every waterproof or vapor proof envelope containing desiccant.

7.5.5 Sealing Openings

- **7.5.5.1** Comply with the minimum end protection requirements criteria outlined in PFI ES-31, *Standard for Protection of Ends of Fabricated Piping Assemblies*, to protect openings and (or) as required in the purchase order. The SELLER shall provide to the BUYER for review and permission to proceed prior to use, against the scope of the applicable P.O of each specific type of desiccant material. Fabrications must be clearly marked, indicating desiccant inside.
- **7.5.5.2** Cover pipe openings with metal, polyethylene, or nonmetallic end caps, flange protectors, or plugs. Caps or plugs shall consist of materials that enable them to perform their intended function adequately without causing deleterious effects on the items or system operation. Simplicity of installation, inspection, and removal without damage to the item shall be considered.

Metallic plugs and caps contacting metal surfaces shall not cause galvanic corrosion at the contact areas. Gasketing or other nonmetallic materials used in conjunction with metallic caps or plugs shall exhibit no corrosive effect on the material.

Polyethylene or nonmetallic end caps and plugs shall be friction fit or secured by other means. Nonmetallic plugs and caps shall be brightly or contrastingly colored. Clear plastic closures are not to be used except when indicated for a special purpose.

At a minimum, one of the caps or plugs on each spool shall be provided with a 1/8-inch maximum diameter vent hole to preclude the buildup of internal pressure.

Avoid placing the cap or plug with the vent hole on a spool that is oriented in an upward, vertical position. Tape shall not be used to secure end caps or plugs.

- **7.5.5.3** Welding end connections shall be provided with adequate bevel protectors to protect from corrosion and physical damage.
- **7.5.6** Components packed in containers shall be blocked, anchored, braced, or cushioned to prevent physical damage to the item or barrier.

7.6 Marking and Labeling⁵²

- **7.6.1** Measures shall be established for marking and labeling for the packaging, shipping, handling, and storage of items as necessary to adequately identify, maintain, and preserve the item.
- **7.6.2** Markings and labels shall indicate the presence of special environments or the need for special controls if necessary.
- **7.6.3** Items or containers of equipment or material shall be marked "Special Storage Requirements" when the storage is other than outside.
- **7.6.4** Temporary structural members for bracing or protection of spools shall be painted bright, fluorescent yellow (including nuts and bolts).
- **7.6.5** Components and their containers shall be identified by marking. Shipping marks shall be on all four sides of the package excluding top and bottom. The shipping marks shall be at least 3-inches high where space permits. Markings are to be in black or white paint or ink depending on the shade of the package surface. International cautionary symbols shall be stenciled in red waterproof paint or ink. Bundles and reels and other packages which cannot be marked directly will have attached corrosion resistant metal tags with raised markings. Hazardous materials will have appropriate markings. For the following types of packages, special markings shall apply:
 - Containers of liquids, sensitive equipment, other items that must be maintained in a specified orientation during storage, handling, or shipping: Arrows, "This Side Up."
 - Packages or pieces exceeding 8 feet in length or height: "Center of Gravity."
 - Packages or pieces to be hoisted by tack or cranes: "Sling Here."
 - Top heavy packages or pieces: "Top Heavy" or "End Heavy."

7.7 Shipping Instructions

- 7.7.1 Pipe spools shall be protected from exposure to adverse environmental conditions.
- **7.7.2** Block, strap, or otherwise hold pipe, fittings, and valves in position and further separate them by dunnage as necessary to prevent damage during shipment.

Note: Pipefitters must be able to walk around the shipment and place slings as required to unload safely. This must be considered when separating pipe, fitting, and valves in position and separating them by dunnage.

7.7.3 The weight, lifting points, or center of gravity shall be indicated on the crate, skid, or package by the shipper and shall be used to ensure proper handling during loading, transfer between carriers, and unloading.

- **7.7.4** Package or preservative coatings shall be visually inspected after loading and damaged areas repaired prior to shipment. Components shipped with desiccant shall be inspected to ensure closures are intact.
- **7.7.5** Sealed openings shall be visually inspected after loading to ensure closures are intact. Materials used for resealing shall be in accordance with original packaging requirements.
- **7.7.6** Where special care is deemed necessary to avert damage, written instructions concerning the location or stacking limits for crates or boxes shall be marked on the containers.
- **7.7.7** Precautions shall be taken to minimize the possibility of theft or vandalism during shipment of components.
- **7.7.8** Pipe spools and other equipment (i.e., valves) subject to internal moisture damage and protected by desiccant shall be provided with sufficient desiccant during shipment. Temporary bracing or supports, marked and tagged for removal after installation, shall be provided to prevent damage during shipment and shall be painted bright, fluorescent yellow.

7.8 Storage

- **7.8.1** SELLER shall store BUYER supplied in-line components indoors in a weather-tight and well-ventilated building or enclosure.
- **7.8.2** SELLER shall store all in-process and completed piping spools that contain welded inline components in a weather-tight and well-ventilated building or enclosure until the time of shipment.
- **7.8.3** The Seller shall specify storage requirements to minimize the possibility of damage or lowering of quality due to corrosion, contamination, deterioration, or physical damage from the time the component is received by the BUYER until it is removed from storage and placed in its final location.
- **7.8.4** For components that may be stored outdoors, the Seller shall specify any cribbing or equivalent to allow for air circulation and to avoid trapping water.
- **7.8.5** Seller shall provide specific guidelines for storage beyond 6 months at the construction site.
- **7.8.6** Seller shall specify the shelf life for any items that may become unserviceable as a result of extended storage.

8 Quality Assurance

SELLER shall implement a Quality Assurance (QA) program in accordance with the QA requirements specified in the Material Requisition. BUYER will specify QA requirements in QA Program Requirements Data Sheets, BUYER specifications, and/or requirements written directly into the Material Requisition (MR). SELLER shall submit Quality Assurance Manual for review.

SELLER's shall perform all work in accordance with a program that meets the requirements of the Q Datasheet of ANSI/ASME NQA-1 (2000)/2004 Addendum for Commercial Grade Dedication. Quality Assurance Program Requirements for Fabricator (Build to Print) and/or Site Installer SELLER/Subcontractor attached to the MR.

8.1 Q Datasheet of ANSI/ASME NQA-1 (2000) Quality Assurance Program Requirements Clarification

MR Attachment A, Q Datasheet of ANSI/ASME NQA-1 (2000) Quality Assurance Program Requirements, Section A. SELLER Qualification, 1. E) is clarified by the following:

- **8.1.1** The term "periodic" means, testing the first items (one of each lot) for each of the first five shipments from each Sub-supplier's PO. Thereafter, for each Sub-supplier's PO, test one of each lot for every tenth (10th) shipment.
- **8.1.2** The term "physical properties" testing means tensile testing which will provide results comparable those reported on the MTR. As an alternate approach, test blanks or test coupons, in accordance with the applicable ASTM, procured from the same heat / lot as the delivered item, maybe used to generate the tensile test specimen.

8.2 NCA 3800/4000 SELLER

This note applies to all PO's issued by the SELLER to their Sub-SELLERS, where an item has to meet the requirements of NCA 3800/4000.

8.2.1 After awarding a PO, the SELLER may evaluate the procurement requirements and determine that implementation audits are not necessary for procuring items that are:

a) Relatively simple and standard in design, manufacturing, and testing; and

b) Adaptable to standard or automated inspections or tests of the end product to verify quality characteristics after delivery to the SELLER's facility.

8.2.2 Pipe, flanges, fittings, and weld filler metal are considered to be relatively simple and standard in design; therefore, audits are not necessary for these items when procured from an NCA 3800 certificate holder.

8.3 Commercial Grade Dedication⁵³

See section 4.8.

8.4 Supplier Deviation Disposition Request

The SELLER shall identify and document conditions that do not meet the requirements of the BUYER's procurement document(s) and/or the BUYER accepted SELLER document(s). Those deviations, along with the SELLER's suggested corrective actions, shall be transmitted to the BUYER on the Supplier Deviation Disposition Request (SDDR) Form for review and disposition. Any deviation is considered unacceptable until approved by the BUYER in writing. Detailed instructions are provided as part of the SDDR Form, a copy of which is attached to the PO. Note that deviations from any code or regulation are not permitted.

9 Configuration Management⁵⁴

Configuration management is maintained by conformance to standards and procedures. The BUYER shall be promptly notified by SDDR of any potential deviation or conflict with the requirements. Work shall not proceed until a resolution is accepted and proper documentation is provided to the BUYER.

10 Documentation and Submittals^{55, 56, 57}

10.1 General

Preparing, maintaining, and submitting any documents for BUYER's review and permission to proceed does not relieve the SELLER from the responsibility of complying with the requirements of the purchase order.

Records shall be legible and complete. Documentation shall be traceable to the applicable hardware. Each document and each entry on each document shall be complete. Blank spaces or incomplete entries are not acceptable. Items not applicable are marked as such. Elements that are not applicable may be marked by "NA" or "N/A". Documents shall be considered valid records only if stamped, initialed, or signed and dated by authorized personnel or otherwise authenticated. Corrections to quality assurance records, including documents that will become quality assurance records, shall be corrected using a single strikethrough method and include the initials of the initiator and date the correction was made immediately adjacent to the change. WhiteoutTM or similar type products are specifically prohibited and will not be accepted. It shall be clear as to what was changed, who initiated the change, and when the change occurred.

Corrections made by manual correction should be avoided.

Submit the following documents as specifically noted in the G-321-E or G-321-V forms attached to the purchase order. Documents listed below in 10.2 and, on the G-321-E form require early submittal. This is necessary to allow time for proper BUYER's review and possible revision required to incorporate BUYER's comments. A review status of "Work May Proceed" shall be obtained prior to use. Submittal of procedures for review applies to any sub-vendor used by the SELLER.

10.2 G-321-E Form Submittals

- **10.2.1** Inspection and test plan. (Section 6.1.2.1) (DCN 6.1)
- **10.2.2** Supplier shall submit a typical quality verification document (QVD) package for BUYER's review and permission to proceed. This submittal will be representative of a QVD package of a completed spool and is required to afford the project an advance review of how a QVD package is formed and is constituted. It is expected that the QVD package will be ordered in sequence of the Document Category Number (DCN) and each DCN will be paginated. This document shall include forms represented in the SELLERs procedures and BUYERs specifications (when applicable) throughout all the processes used in executing the procurement, receipt and tracking of materials, fabrication, coating, insulation, wrapping, including all inspection, repair (if applicable) and testing (as applicable). (DCN 10.1)

- **10.2.3** The SELLER shall identify the specific products by manufacturer and catalog number and shall submit the coating manufacturer's latest published product data sheet, application instructions, and Safety Data Sheets (SDS). Conflicts, if any, between the SELLER's normal procedures, the coating manufacturer's recommendations, and this specification shall be brought to the attention of the BUYER for resolution and require written permission to proceed (Ref Para 9.2.4 of 24590-WTP-3PS-AFPS-T0001) (DCN 11.0).
- **10.2.4** The SELLER shall submit a current copy of the Product Data Sheet, application instructions and SDS for each coating product to be applied. (Ref Para 9.2.2.4 of 24590-WTP-3PS-PX04-T0002) (DCN 11.0)
- **10.2.5** The SELLER shall submit original coating manufacturer's Product Identity Certification Records (PICR) for every batch of coating material intended for use (Appendix A). (Ref Para 9.2.2.5 of 24590-WTP-3PS-PX04-T0002) (DCN 11.0)
- **10.2.6** The SELLER shall identify the specific products by manufacturer and catalog number and shall submit the manufacturer's latest published Product Data Sheet, application instructions and Safety Data Sheets (SDS). (Ref Para 9.2.2 of 24590-WTP-3PS-PX04-T0005)(DCN 11.0)
- **10.2.7** The SELLER shall submit a current copy of the Product (or Technical) Data Sheet, application instructions and SDS for each product to be applied. (Ref Para 9.2.2.3 of 24590-WTP-3PS-PX04-T0006)(DCN 11.0)
- **10.2.8** Chemical analysis report for shop marking materials and tape. (Section 5.8.3) (DCN 11.0)
- **10.2.9** Material description (safety data sheet) for each specific type of desiccant material. (7.5.5.1) (DCN 11.0)
- **10.2.10** Weld Procedure Specification (WPS) and associated Procedure Qualification Record (PQR) (submit each WPS with the respective PQR). Welding includes tack welds. (Section 5.4.1.1) (DCN 12.0)
- **10.2.11** Filler Material Control and Storage Procedure. (Section 5.4.4.3) (DCN 13.0)
- 10.2.12 Weld Repair Procedure. (Section 5.4.1.3) (DCN 14.0)
- **10.2.13** The final procedure and documentation forms shall be submitted and reviewed with BUYER's permission to proceed prior to the start of coating Work. Submittal requirements for manufacturer's standard coating are found in 24590-WTP-3PS-AFPS-T0001 Section 3.2.(Ref Para 9.2.1 of 24590-WTP-3PS-AFPS-T0001) (DCN 15.0)
- **10.2.14** Post Weld Heat Treatment (PWHT) Procedure. (If applicable) (Section 5.5.1.1) (DCN 16.0)
- **10.2.15** UT Ultrasonic Examination Procedure. (Section 4.2.4.1 & 6.3) (DCN 19.0)
- **10.2.16** RT Radiographic Examination Procedure. (Section 4.2.4.1 & 6.3) (DCN 20.0)
- **10.2.17** Radiograph examination film and technique reader sheets. (Section 4.2.4.1 & 6.3) (DCN 20.1) Shipped directly to PDC with notification of such to the BUYER.

24590-WTP-3PS-PS02-T0014, Rev 1 Shop Fabrication for Q Dual Containment Piping and CM Black Cell/Hard-to-Reach Dual Containment Piping

- **10.2.18** MT Magnetic Particle Examination Procedure. (Section 6.3) (DCN 21.0)
- **10.2.19** PT Liquid Penetrant Examination Procedure. (Section 6.3) (DCN 22.0)
- **10.2.20** Hydrostatic leak testing procedure. (Section 6.4.1.1) (DCN 24.0)
- **10.2.21** Pneumatic leak testing procedure. (Section 6.4.1.2) (DCN 24.0)
- **10.2.22** Bending procedure for pipe bends and bend qualification record. (Sections 5.2 and 5.2.1.1) (DCN 25.0)
- 10.2.23 VT Visual Test Procedure. (Section 6.3) (DCN 25.0)
- **10.2.24** Inspection procedure (s) (Section 6.1.1) (Dimensional DCN **25**.0)
- **10.2.25** Inspection procedure (s) (Section 6.1.1) (Wall Thickness DCN 25.0)
- **10.2.26** Inspection procedure (s) (Section 6.1.1 & 5.2) (Bending DCN 25.0)
- **10.2.27** Proof Tests (Section 4.2.3) (DCN 27.0)
- **10.2.28** Welder Personnel Qualification Procedure. (Section 5.4.1.1) (DCN 28.0)
- **10.2.29** Nondestructive Examination (NDE) "Written Practice" (NDE Personnel Qualification Procedure). (Section 6.2.2) (DCN 28.0)
- **10.2.30** SELLER shall prepare detailed written procedures for material receiving, marking, storage, handling, surface preparation, environmental control, application, curing, inspection, testing, touch-up/repair, application personnel qualification, inspector qualification, and proposed documentation forms as described within specification 24590-WTP-3PS-AFPS-T0001. (DCN 28.0
- **10.2.31** The SELLER shall provide a personnel training and certification plan for applicators and inspectors. (refer to Specification 24590-WTP-3PS-AFPS-T0001 Sections 1.4.17, 4.1.2, 5.1.1, 7.1.6 & 9.2.7) (DCN 28.0).
- **10.2.32** SELLER shall submit a detailed written procedure that addresses material receiving, marking, storage, handling, surface preparation, environmental control, application, curing, inspection, testing, touch-up/repair, and proposed documentation forms. The procedure and documentation forms shall be submitted for the BUYER approval prior to the start of coating work. (Ref Para 9.2.2.1 of 24590-WTP-3PS-PX04-T0002) (DCN 28.0)
- **10.2.33** The SELLER shall submit a daily inspection record as part of the work procedures that includes the required elements provided in Appendix C. (Refer to Section 7.1.1) (Ref Para 9.2.2.2 of 24590-WTP-3PS-PX04-T0002) (DCN 28.0)
- **10.2.34** SELLER shall submit a detailed written procedure that addresses material receiving, marking, storage, handling, surface, preparation, environmental control, application, curing, inspection, testing, touch-up/repair, and proposed documentation forms. The suppliers submitted
procedure shall clearly define each step in the work process including inspection and installation. (Ref Para 9.2.2 of 24590-WTP-3PS-PX04-T0005)(DCN 28.0)

- **10.2.35** SELLER shall submit a detailed written procedure that addresses material receiving, storage, handling, surface preparation, environmental control, application, inspection, testing, touch-up/repair, and proposed documentation forms. (Ref Para 9.2.2.1 of 24590-WTP-3PS-PX04-T0006)(DCN 28.0)
- **10.2.36** The SELLER shall submit their general procedure for handling, storage, and preparation for shipment of pipe spools. (Section 7.1.1) (DCN 29.0)
- **10.2.37** Positive Material Identification (PMI) Procedure (Section 4.7) (DCN 31.0)
- **10.2.38** Commercial Grade Dedication (CGD) Procedure(s) & Plan(s) (Section 4.8.1 & 8.3) (DCN 33.0)
- **10.2.39** Welding Procedure Application List (WPAL) identifying WPSs being used for each individual pipe class. This list shall also identify the extent of NDE per individual pipe class, preheat, and PWHT as applicable. The attached form (Figure 1) in Appendix B is a typical example of a welding procedure application list. (Section 5.4.1.2) (DCN 36.0)

10.3 G-321-V Form Submittals

The following quality verification documentation (QVD) listed on the G-321-V form shall be submitted for each spool as applicable, and provided with the spool shipment:

- **10.3.1** The SELLER shall submit the original or copies of the original Coating Manufacturer's Product Identity Certification Records (24590-WTP-3PS-AFPS-T0001 Appendix F) for each and every batch of coating material used.(Ref Para 9.2.5 of 24590-WTP-3PS-AFPS-T0001) (DCN 13.0).
- **10.3.2** All original documentation generated during the coating work shall be submitted within 30 days after the last item in the order is completed. The documentation package shall have a cover page that identifies the project, contract number and the report issue date. There shall be a letter of certification signed by management certifying the work performed complied with all the contract performance and quality requirements. The submittal package shall contain a table of contents and organized sections that contain the original documents generated during the work. (Ref Para 10.7 of 24590-WTP-3PS-PX04-T0001) (DCN 13.0)
- **10.3.3** There shall be a letter of certification signed by management certifying the work performed complied with all the contract quality requirements (Ref Para 9.2.3.2 of 24590-WTP-3PS-PX04-T0002) (DCN 13.0)
- **10.3.4** There shall be a letter of certification signed by management certifying the work performed complied with all the contract quality requirements. (Ref Para 9.2.3.3 of 24590-WTP-3PS-PX04-T0005) (DCN 13.0)
- **10.3.5** There shall be a letter of certification signed by management certifying the work performed complied with all the contract quality requirements. (Ref Para 9.2.5.2 of 24590-WTP-3PS-PX04-T0006)(DCN 13.0)

- **10.3.6** A Certificate of Conformance (C of C) from the HDPE shrink sleeve manufacturer for each batch of material (sleeve, filler & patch material) shall be provided, the C of C shall certify the supplied material meets all the physical properties contained in their published technical data sheet. The C of C shall be provided with purchased material (Attach "C of C") to an Appendix A Product Identity Record filling in applicable sections, signed and dated by the product manufacturer. (Ref Para 9.2.2.4 of 24590-WTP-3PS-PX04-T0006)(DCN 13.0)
- **10.3.7** Major Weld Repair Record. (Section 5.4.1.3) (DCN 14.0)
- **10.3.8** The SELLER shall submit a daily inspection record as part of the work procedures that includes the required elements as noted in specification 24590-WTP-3PS-AFPS-T0001 Appendix G (Ref Para 9.2.6 of 24590-WTP-3PS-AFPS-T0001) (DCN 15.0)
- **10.3.9** A record of all required daily production test results for pre-surface preparation, chloride contamination testing, environmental conditions, compressed air cleanness, surface preparation and profile, acid wash, steel defects, FBE coating, water quench, etc. shall be included as part of the shipping documentation package. (Ref Para 10.2 of 24590-WTP-3PS-PX04-T0001) (DCN 15.0)
- **10.3.10** A record of all required daily production test results on each FBE coated pipe, such as dry film thickness, holiday testing, visual defects including X cut knife adhesion tests shall be included as part of the shipping documentation package. (Ref Para 10.3 of 24590-WTP-3PS-PX04-T0001) (DCN 15.0)
- **10.3.11** A record of all required daily laboratory test results for each Lot of FBE coated pipe shall be included as part of the shipping documentation. (Ref Para 10.4 of 24590-WTP-3PS-PX04-T0001) (DCN 15.0)
- **10.3.12** Material test reports (MTR). (Section 4.4.1) (DCN 17.1)
- **10.3.13** A record of all material test reports (individual batches of FBE powder, phosphoric acid, abrasive, rinse water quality, silver nitrate, etc.) for materials used for a given Lot of FBE coated pipe, shall be included as part of the shipping documentation package. (Ref Para 10.1 of 24590-WTP-3PS-PX04-T0001) (DCN 17.1)
- **10.3.14** Certificates of Compliance. (Sections 4.4.1 and 5.9.1) (DCN 17.4)
- **10.3.15** UT Ultrasonic Examination Reports. (Section 4.2.4.1 & 6.3) (DCN 19.0)
- **10.3.16** RT Radiographic Examination Reports. (Section 4.2.4.1 & 6.3) (DCN 20.0)
- **10.3.17** MT Magnetic Particle Examination Report. (Section 6.3) (DCN 21.0)
- **10.3.18** PT Liquid Penetrant Examination Report. (Section 6.3) (DCN 22.0)
- **10.3.19** Hydrostatic leak testing results. (Section 6.4.1.1) (DCN 24.0)
- **10.3.20** Pneumatic leak testing results. (Section 6.4.1.2) (DCN 24.0)
- **10.3.21** Bulk pipe and buttweld fitting wall thickness reports. (Section 4.2.2) (DCN 25.0)

- **10.3.22** Bend dimensional verification reports. (Section 5.2.2.6) (DCN 25.0)
- **10.3.23** A record of all required inspections shall be provided that includes pre-surface preparation, compressed air cleanliness, environmental conditions, surface preparation and profile, location of repair or field joint protected, sleeve application, visual inspection, and holiday testing, and all touch-up and repair of applied HDPE shrink sleeves. These records shall include the batch or lot number of the HDPE shrink sleeve materials used, and the ID of the items protected to provide traceability. Multiple items, areas or joints may be documented on the same inspection record for each day's production. (Ref Para 9.2.4 of 24590-WTP-3PS-PX04-T0006)(DCN 25.0)
- **10.3.24** Positive Material Identification (PMI) results (Section 4.7.5) (DCN 31.0)
- **10.3.25** Commercial grade dedication (CGD) quality verification documentation (Section 4.8.1) (DCN 33.0)
- **10.3.26** Pipe spool sheets, extended spool sheets, or detailed drawings. (Section 4.4.2) (DCN 36.0)
- **10.3.27** Completed Soft Seated Valve Installation Records. (Section 5.1.11.10) (DCN 36.0)



Appendix A

Nondestructive Examinations (NDE) Summary of Shop Welds

SEE SECTION 6.3 FOR APPLICABLE SHOP WELD NDE,	Dual Containm	ent Piping ¹
INSPECTION, AND ACCEPTANCE CRITERIA REQUIREMENTS.	Inner Piping	Jacket Piping
Type of Weld		
All Girth and Miter Welds	100% VT	100% VT
	100% RT or 100% UT	5% RT or 5% UT
Outer Pipe Shop Performed Longitudinal/Clam Shell Welds for Dual	N/A	100% VT
Contained Piping		5% RT or 5% UT
All Pipe and Integral Attachment Fillet Shop Welds – including thermowell	100% VT	100% VT
socket welds, integral support weld, non-pressure & non load bearing piping	100% PT	
attachment welds		
<u>All</u> integrally reinforced forged branch fittings welded to main piping run.	100% VT	100% VT
If weld is chosen as part of 5% RT population and RT will not produce an	100% PT	
interpretable radiograph, In-Process Examination and liquid penetrant		
examination of the exterior of both the root and final pass is acceptable.		
Footnote:		
The circumferential weld that connects an integrally reinforced forged branch		
fitting to the branch pipe run shall be examined as a girth weld.		
Legend: VT = Visual Examination per ASME B31.3 para 344.2		
RT = Radiographic Examination per ASME B31.3 para 344.5		

UT = Ultrasonic Examination per ASME B31.3 para 344.6

PT = Liquid Penetrant Examination per ASME B31.3 para 344.4

¹Includes embedded piping, if applicable

Appendix B Forms

										*						
					Pipi	ng Cl	asses									
Design Code: Category:																
Pipe Class(es):	Line Services		Limits: - Temperat				ature ºF, ºC / Pressure psig, bars									
Piping Material Type:					P Number	r.			Hard	ness						
CVN Requirements:			PMI	Requirer	ments:	Ferrite Requireme			ements:	ments:						
Pipe Class or Weld	Pipe Size	Max W.T.	Joint	Heat	Freatment	Wolding Dropoduro Spooif		Specific	NDE (%			,)				
Description	(nps)	inches mm	Туре	PRE	POST		vvolaling i	1 1000		Opcomo		RT	MT	PT	VT	Other
					1											
. N																
Notes:																

Figure 1Welding & NDE Matrix (Welding Procedure Application List)





Chromium Equivalent = % Cr + % Mo + (1.5 x % Si) + (0.5 x % Cb)

GENERAL NOTE:

The actual nitrogen content is preferred. If this is not available, the following applicable nitrogen value shall be used:

a) GMAW welds–0.08% (except self-shielding flux colored electrode GMAW welds–0.12%)
 b) welds of other processes–0.06%

Figure 3 Soft Seated Valve Installation Record

1.	Valv	e Information
	a.	Manufacturer:
	b.	Bechtel Stock No: PVV
	C.	Bechtel Valve Tag No:
	d.	Manufacturer's Instructions WTP No: 24590
	e.	Size: NPS
2.	Ass	embled Valve Welding
	a.	Assembled Welding Permitted:
		[] - Yes [] - No - Skip to Step 3
	b.	Assembled Welding Selected:
		[] - Yes [] - No - Skip to Step 3
	C.	Valve body heat control in accordance with manufacturer's steps:
	d.	Maximum valve body temperature observed during welding.
		Pyrometer ID No:
		Temperature Crayon Brand/Rating:
		Signature / Date:/
3.	Valv	e Disassembly / Reassembly / Torquing
	a.	Valve disassembled in accordance with manufacturer's steps:
	h	Value reassambled in accordance with manufacturer's steps:
	D.	valve reassembled in accordance with manufacturer's steps.
	c.	Final fastener torque required (max / min or range):
	d.	Actual fastener torque:
		#1#2#3#4
		#5 #6 #7 #8
		Torque Wrench ID No:
		Signature / Date: /



	Applicat	ENGINEER	ING SPECIFICA FOR oxy Coating for F	ATION	
	Applicat	ion of Liquid Ep	FOR oxy Coating for F		
	Applicat	ion of Liquid Ep	oxy Coating for H		
				Piping Appl	ications
Conten	t applicable to ALARA	λ?	🗌 Yes 🖾 No Rev	F	Quality Level
Retroa	ctive applicability: No		N/A (alpha revision o	or revision 0)	DOE Contract No.
					DE-AC27-01RV14130
5	2/26/2020	By: Mark Malloy - mmalloy Org Name: MET Placed: Feb 14, 2020	Checked By: Ming Fang - mang Org Name: Placet: Foi 14 2020	N/A	DE-AC27-01RV14130
5	2/26/2020 11/18/08	Driginator By: Mark Malloy - mmalloy Org Name: MET Placed: Feb 14, 2020 TRC	Checked By: Ming Fang - mfang Org Nama: Placed: Feb 14, 2020 LRL	N/A LDK	DE-AC27-01RV14130
5 4 3	2/26/2020 11/18/08 11/8/04	Driginator By: Mark Malloy - mmalloy Org Name: MET Placed: Feb 14, 2020 TRC TRC	Checked By: Ming Fang - mfang Org Nama: Placed: Feb 14, 2020 LRL LDK	N/A LDK LDK	DE-AC27-01RV1413
5 4 3 2	2/26/2020 11/18/08 11/8/04 6/3/03	Originator By: Mark Malloy - mmalloy Org Name: MET Placed: Feb 14, 2020 TRC TRC TRC TRC TRC TRC	Checked By: Ming Fang - mfang Org Nama: Placed: Feb 14, 2020 LRL LDK LDK MSR	N/A LDK LDK DTS	DE-AC27-01RV14130
5 4 3 2 1	2/26/2020 11/18/08 11/8/04 6/3/03 4/5/02	Originator By: Mark Malloy - mmalloy Org Name: MET Placed: Feb 14, 2020 TRC	Checked By: Ming Fang - mfang Org Name: Placed: Feb 14, 2020 LRL LDK MSR LDK LDK	N/A LDK LDK DTS DTS	DE-AC27-01RV14130 JP MB DTS DTS FOR SF
5 4 3 2 1 0	2/26/2020 11/18/08 11/8/04 6/3/03 4/5/02 10/23/01	Originator By: Mark Malloy - mmalloy Org Name: MET Placed: Feb 14, 2020 TRC TRC	Checked By: Ming Fang - mfang Org Name: Placed: Feb 14, 2020 LRL LDK MSR LDK LDK LDK	N/A LDK LDK DTS DTS DTS	DE-AC27-01RV14130 JP MB DTS DTS FOR SF DTS

Revision History

	Q Specificat Revision O Margin Redu	tion nly iced?	CM Only
Reason for Revision	YES	NO	N/A
Issued for Construction	N/A	N/A	N/A
Revised for Construction Direct Hire.	N/A	N/A	N/A
Identify Coal Tar Epoxy; previously incorporated 24590-WTP-FCR-P-01-013 in rev.1; incorporate 24590-WTP-FCR-P-02-047; incorporate 24590-WTP-FCR-P- 03-008; incorporate 24590-WTP-SDDR-PROC-02-0009; incorporate by reference 24590-WTP-NCR-CON-02-079	N/A	N/A	N/A
Include field coating of buried stainless steel items; Clarify shrink sleeve temperature limits; FBE touch up w/Scotchkote 323; incorporated 24590-WTP-FCR-P-03-058; incorporated 24590-WTP-3PN-PX04-00003; incorporated 24590-WTP-FCR-P-04-0097.	N/A	N/A	N/A
Incorporated 24590-WTP-3PN-PX04-00008; incorporated 24590-WTP-FC-C- 07-0244; incorporated 24590-WTP-FC-P-06-0108; incorporated 24590-WTP- FC-P-07-0225; incorporated 24590-WTP-FC-P-08-0201; added DuPont Nap- Gard 7-2508 fusion bonded epoxy coating to Section 5.1.1.3	N/A	N/A	N/A
Major revision - Superseding specifications 24590-WTP-3PS-PX04- T0003 and 24590-WTP-3PS-PX04-T0004 by incorporation into this specification. Moved insulation data into specification 24590-WTP-3PS- PX04-T0005. Moved HDPE shrink sleeve data into specification 24590- WTP-3PS-PX04-T0006. The following affected by documents, found in EDMS, against Rev. 4 have been Incorporated: 24590-WTP-3PN-PX04-00010; 24590-WTP- FC-P-13-0365; 24590-WTP-FC-P-16-0175; 24590-WTP-FC-P-19-1106. The following change documents against Rev. 4 of this spec were incorporated with no changes; 24590-WTP-FC-P-09-0413 was a one- time exception and 24590-WTP-SDDR-PL-17-00041 was an as-built material clarification that also met original spec requirements. The following change documents were written against spec 24590-WTP- 3PS-PX04-T0003 and were included into this specification revision: 24590-WTP-3PN-PX04-00004; 24590-WTP-3PN-PX04-00021; 24590- WTP-SDDR-PROC-04-00513. The following affected by documents are incorporated with no changes made, as the information has been included in new specifications 24590- WTP-3PS-PX04-T0005 and/or 24590-WTP-3PS-PX04-T0006: 24590- WTP-3PN-PX04-00022; 24590-WTP-FC-P-18-0248; 24590-WTP-FC-P-	N/A	N/A	
	Reason for Revision Issued for Construction Direct Hire. Identify Coal Tar Epoxy; previously incorporated 24590-WTP-FCR-P-01-013 in rev.1; incorporate 24590-WTP-FCR-P-02-047; incorporate 24590-WTP-FCR-P. 03-008; incorporate 24590-WTP-SDDR-PROC-02-0009; incorporate by reference 24590-WTP-NCR-CON-02-079 Include field coating of buried stainless steel items; Clarify shrink sleeve temperature limits; FBE touch up w/Scotchkote 323; incorporated 24590-WTP- FCR-P-03-058; incorporated 24590-WTP-3PN-PX04-00003; incorporated 24590-WTP-FCR-P-04-0097. Incorporated 24590-WTP-3PN-PX04-00008; incorporated 24590-WTP- FC-P-07-0225; incorporated 24590-WTP-FC-P-08-0201; added DuPont Nap- Gard 7-2508 fusion bonded epoxy coating to Section 5.1.1.3 Major revision - Superseding specifications 24590-WTP-3PS-PX04- T0003 and 24590-WTP-3PN-PX04-0004 by incorporation into this specification. Moved insulation data into specification 24590-WTP-3PS- PX04-T0005. Moved HDPE shrink sleeve data into specification 24590-WTP-3PS- PX04-T0006. The following affected by documents, found in EDMS, against Rev. 4 have been Incorporated: 24590-WTP-FC-P-09-0413 was a one- time exception and 24590-WTP-SDDR-PL-17-00041 was an as-built material clarification that also met original spec requirements. The following affected by documents were written against spec 24590-WTP- 3PS-PX04-T0003 and were included into this specification revision: 24590-WTP-3PN-PX04-00004; 24590-WTP-FC-P-09-0413 was a one- time exception and 24590-WTP-SDDR-PL-17-00041 was an as-built material clarification that also met original spec requirements. The following affected by documents are incorporated with no changes made, as the information has been included into this s	Reason for RevisionQ Specificat Revision 0Issued for ConstructionN/ARevised for Construction Direct Hire.N/AIdentify Coal Tar Epoxy; previously incorporated 24590-WTP-FCR-P-01-013 in rev.1; incorporate 24590-WTP-FCR-P02-047; incorporate 24590-WTP-FCR-P03-008; incorporate 24590-WTP-SDDR-PROC-02-0009; incorporate by reference 24590-WTP-NCR-CON-02-079N/AInclude field coating of buried stainless steel items; Clarify shrink sleeve temperature limits; FBE touch up w/Scotchkote 323; incorporated 24590-WTP- FCR-P-03058; incorporated 24590-WTP-3PN-PX04-00003; incorporated 24590-WTP-FCR-P-04-0097.N/AIncurporated 24590-WTP-3PN-PX04-00008; incorporated 24590-WTP- FC-P-07-0225; incorporated 24590-WTP-FC-P-08-021; added DuPont Nap- Gard 7-2508 fusion bonded epoxy coating to Section 5.1.1.3N/AMajor revision - Superseding specifications 24590-WTP-3PS-PX04- T0003 and 24590-WTP-3PS-PX04-10004 by incorporation into this specification. Moved insulation data into specification 24590-WTP-3PS- PX04-10005. Moved HDPE shrink sleeve data into specification 24590-WTP- FC-P-13-0365; 24590-WTP-FC-P-06-0113 was a one- time exception and 24590-WTP-SDDR-PL-17-00041 was an as-built material clarification that also met original spec requirements. The following change documents against Rev. 4 of this spec were incorporated 24590-WTP-SDDR-PL-17-00041 was an as-built material clarification that also met original spec requirements. The following affected by documents are incorporated with no changes rads as the information has been included into this specification revision: 24590-WTP-3DN-PX04-00004; 24590-WTP-SDN-PX04-00021; 24590-WTP- SPS-PX04-T0003 and were included into the specification revision: 24590-WTP-3DN-PX04-00004; 24590-WTP-SDN-PX04-00021; 24590-WTP- SPS-PX04-T0003 and/ve	Q Specification Revision Only Margin Reduced?Reason for RevisionYESNOIssued for ConstructionN/AN/ARevised for Construction Direct Hire.N/AN/AIdentify Coal Tar Epoxy; previously incorporated 24590-WTP-FCR-P-01-013 in rev.1; incorporate 24590-WTP-FCR-P02-047; incorporate 24590-WTP-FCR-P. 03-008; incorporate 24590-WTP-RCC-02-0009; incorporate by reference 24590-WTP-NCR-C0.00-079N/AN/AInclude field coating of buried stainless steel items; Clarify shrink sleeve temperature limits; FBE touch up w/Sochchote 323; incorporated 24590-WTP- FCR-P-03-058; incorporated 24590-WTP-SDP-N2N04-00003; incorporated 24590-WTP- FCR-P-04-0097.N/AN/AIncorporated 24590-WTP-SDP-N2N04-00008; incorporated 24590-WTP- FCR-P-04-0097.N/AN/AMajor revision - Superseding specifications 24590-WTP-SPS-PX04- T0003 and 24590-WTP-FC-P-06-0108; incorporated 24590-WTP-SPS- PX04-70005. Moved Insplation data into specification 24590-WTP- SPS-PX04-70006.N/AN/AThe following affected by documents, found in EDMS, against Rev. 4 have been Incorporated 24590-WTP-FC-P-09-0413 was a one- time exception and 24590-WTP-FC-P-19-0413 was a one- time exception and 24590-WTP-FC-P-19-0413 was a one- time exception and 24590-WTP-FC-P-19-0413 was a one- time exception and 24590-WTP-FC-P-00-0413 was an as-built material clarification that also met original spec requirements.N/AN/AN/AN/AN/ANPS-PX04-T0003 and were included into this specification s24590-WTP- SPS-PX04-T0003 and vere included into this specifications 24590-WTP- SPS-PX04-T0003 and vere included into this specifications 24590-WTP- SPS-PX04-T0004N/A<

Contents

1	Scop	e1
	1.1	Equipment, Material, and Services Required1
	1.2	Work by Others1
	1.3	Definitions2
	1.4	Safety/Quality Classifications
	1.5	Personnel Training
2	App	licable Documents4
	2.1	Codes and Standards4
	2.2	Reference Documents/Drawings
3	Mate	erials6
	3.1	Abrasives
	3.2	Coating Materials
	3.3	Prohibited Materials
4	App	lication8
	4.1	General
	4.2	Pre-Surface Preparation
	4.3	Surface Preparation
	4.4	Application of Liquid Epoxy10
	4.5	Remedial Work11
5	Test	s and Inspections12
	5.1	General12
	5.2	Environmental Conditions Inspection
	5.3	Surface Preparation Inspection13
	5.4	Liquid Epoxy Inspection14
	5.5	Holiday Inspection
6	Stor	age, Handling and Shipping16
	6.1	General16
	6.2	Shelf Life
	6.3	Material Use17
7	Qua	lity Assurance
	7.1	General
8	Cont	figuration Management18
	8.1	General

24590-WTP-3PS-PX04-T0002, Rev 5 Application of Liquid Epoxy Coating for Piping Applications

9	Docu	ımentation and Submittals1	9
	9.1	General	9
	9.2	Submittals	9

Appendices

Appendix A Coating Manufacturer's Product Identity Certification Record	A-1
Appendix B Procedure for Holiday Testing	B-1
Appendix C Inspection Form	C-1

Tables

Table 1 - Coating Materials for Carbon Steel Substrates	7
Table 2 - Coating Materials for Stainless Steel Substrates	7
Table 3 - Repair Only Coating Materials	7
Table 4 - Coating Material DFT Requirements 11	l
Table 5 - Coating Inspection Frequency	3
Table 6 - Number and Locations of DFT Spot Measurements 14	5

1 Scope

1.1 Equipment, Material, and Services Required

- 1.1.1 This Specification defines the minimum requirements for procurement, application and inspection of liquid epoxy coating for piping applications; including but not limited to underground, embedded (including incidental sections that may be above ground or not embedded) and insulated installations. This includes initial application of coating to pipe and pipe components such as fittings and valves (before or after installation) to either Carbon Steel (CS) or Stainless Steel (SS) substrates. This includes application of coating to shop and field weld areas, spot repairing damage to Fusion Bonded Epoxy (FBE) coated pipe, and spot repair of liquid epoxy coated components, and other applications as deemed necessary/applicable.
- 1.1.2 For initial coating application to a component that will later be welded to another component the ends of the component shall not be painted. This coating holdback area shall be two to three inches (2.0" 3.0"). For irregular shaped components such as elbows or valves if this holdback is greater than 50% of the surface area of the item then the item does not require shop coating.
- 1.1.3 The epoxy coating application and/or repair work, described herein, is considered Commercial Grade quality and NQA-1 does not apply.
- 1.1.4 Surface preparation, material application, and inspection shall be in accordance with this Specification and approved procedures.
- 1.1.5 All materials shall be stored in accordance with this specification and individual product manufacturer recommendations.
- 1.1.6 Application and inspection documentation for all liquid epoxy coated pipe, valves and fittings shall be provided in accordance with this specification. After installation, all exposed portions of bare carbon steel bolts and stainless-steel bolts shall be coated the same as required for repair areas on fitting and valves.
- 1.1.7 Inspection equipment shall be calibrated and controlled by an established Measuring and Test Equipment (M&TE) calibration program.
- 1.1.8 Personnel performing application and/or inspection shall be trained and/or certified as further defined in Section 1.5.
- 1.1.9 For shop applications this scope includes handling of pipe and pipe components and may include shipping as defined in the Material Requisition (MR).

1.2 Work by Others

1.2.1 Welding.

- 1.2.2 Application of FBE coating.
- 1.2.3 Handling of components in the field.

1.3 Definitions

- 1.3.1 Batch/Lot A quantity of material made in one production run. A unique batch number is assigned for each production run of each component for the resin material (Part A), the curing agent Part B, any fillers (Part C or other) and the thinner for coating materials. Single component products will only have 1 batch number assigned to them.
- 1.3.2 BUYER The Bechtel entity shown on the PURCHASE ORDER form.
- 1.3.3 Dry Film Thickness (DFT) The thickness of an applied coating once dry or cured. Usually expressed in mils (each mil is 1/1000th of an inch (0.001")).
- 1.3.4 Field Work Includes work performed on the project jobsite as described in Section 1.1.
- 1.3.5 Fingernail Hard Using a thumbnail perpendicular to the surface, press down hard and move thumbnail across the surface. If coating surface is only slightly marred with no indentation, the coating is fingernail hard.
- 1.3.6 Fish Eyes (cratering) Formation of holes or visible depressions in the coating film. Usually from contamination on the surface prior to applying the coating.
- 1.3.7 Holiday A pinhole, skip, discontinuity, or void in the applied coating.
- 1.3.8 mil a unit of measurement equal to 0.001 inches.
- 1.3.9 NIST National Institute of Standards and Technology
- 1.3.10 Non-Radiation areas The Special Protective Coating (SPC) quality designation for items that are NOT located in an area where radioactive material is transported, processed or stored (areas designated C2 and less, or R2 and less. Refer to 24590-WTP-3DI-W12W-00004 (formerly 24590-WTP-GPP-SRAD-007), Classification of Areas for Design).
- 1.3.11 Pinholes Minute holes through the entire thickness of the coating.
- 1.3.12 Profile The surface roughness resulting from surface preparation by abrasive blasting or power tool cleaning. The specified angular profile can be destroyed by chemical etching, power tool cleaning or other methods.
- 1.3.13 Radiation areas The Special Protective Coating (SPC) quality designation for items that are located in an area where radioactive material is transported, processed or stored) areas designated C3 and greater, or R3 and greater. Refer to 24590-WTP-3DI-W12W-00004 (formerly 24590-WTP-GPP-SRAD-007), Classification of Areas for Design.

- 1.3.14 Sag A nonuniform downward flow of coating that results in an uneven application having a thicker lower edge. (Same definition for runs and drips)
- 1.3.15 SELLER Means the company, corporation, partnership, individual or other entity to which this subcontract (Purchase Order/ Material Requisition) is issued. Including its authorized representatives, successors, and permitted assigns. This also includes Buyer construction personnel when the BUYER is self-performing the work.
- 1.3.16 Shop Work Includes all work performed not on the project jobsite as described in Section 1.1.
- 1.3.17 Training and Certification Training to include applicable work specific skills and an understanding of the specification, approved work procedures and/or manufacturers published instructions. Certification constitutes documented attendance at a training session and a documented performance test demonstrating satisfactory capability.

1.4 Safety/Quality Classifications

- 1.4.1 Surface preparation and coatings work shall comply with all applicable environmental and safety provisions, laws, regulations, ordinances, etc., of the City, County, and State or Federal (National) requirements for the coating materials being used. Work shall be in strict accordance with OSHA 29 CFR 1910, and OSHA 29 CFR 1926.26.
- 1.4.2 Compliance with OSHA Hazard Communication Standard 29 CFR 1910.1200 is required. Safety Data Sheets (SDS) (formerly Material Safety Data Sheets or MSDS) shall be obtained from the materials manufacturer and shall be readily available at the construction site.
- 1.4.3 The Volatile Organic Compound (VOC) content of all materials shall meet Federal, State, and Local Regulations.
- 1.4.4 Solvents and coatings have volatiles, and their vapors shall be kept away from sources ignition. This is especially critical when working in trenches or excavations where solvent vapors tend to collect or settle.

1.5 Personnel Training

1.5.1 Individual applicators shall be trained, qualified and certified in accordance with the SELLERS prepared procedures. Hands-on training shall encompass satisfactorily installing the liquid epoxy materials in accordance with approved procedures. Satisfactory training and installation proficiency testing shall be documented for each applicator.

- 1.5.2 Individual inspectors shall be trained, qualified and certified in accordance with the SELLER's prepared and approved procedures, the procedure shall be based on ASTM D4537. Training shall include a review of the applicable requirements of this specification and the SELLERS approved procedures. The training procedure shall ensure that inspectors meet or exceed the minimum requirements for a Level I coatings inspector, as described in ASTM D4537. Inspectors will preferably satisfactorily demonstrate their proficiency during applicator proficiency testing on at least one liquid epoxy application and be issued a certification in accordance with ASTM D4537. The NACE Coating Inspector Program is considered equivalent to the ASTM D4537 program for the execution of this scope of work.
- 1.5.3 The applicator trainer/qualifier shall have had previous work experience with multicomponent epoxy coatings and be fully familiar with this specification, approved procedures, multi-component epoxy materials, and associated installation equipment.
- 1.5.4 The inspector trainer/qualifier shall have the same qualifications as the applicator trainer/qualifier in addition to a full understanding of the inspection requirements, use of inspection equipment and documentation requirements.

2 Applicable Documents

2.1 Codes and Standards

The applicable requirements of the following codes, standards and specifications form part of this Specification.

2.1.1 ASTM International formerly American Society for Testing and Materials (ASTM)

ASTM E337	Test for Relative Humidity by Wet-and-Dry Bulb Psychrometer
ASTM D4285	Test Method for Indicating Oil or Water in Compressed Air
ASTM D4417	Field Measurement of Surface Profile of Blast Cleaned Steel
ASTM D4537	Standard Guide for Establishing Procedures to Qualify and Certify Personnel Performing Coating and Lining Work Inspection in Nuclear Facilities.
ASTM D4940	Test for Conductimetric Analysis of Water-Soluble Ionic Contaminants of Blasting Abrasives.
ASTM D5162	Practice for Discontinuity (Holiday) Testing of Non-Conductive Protective Coating on Metallic Substrates

- 2.1.2 The Society for Protective Coatings (SSPC)
 - SSPC-AB 1 Mineral and Slag Abrasives

24590-WTP-3PS-PX04-T0002, Rev 5 Application of Liquid Epoxy Coating for Piping Applications

- SSPC-PA 2 Procedure for determining Conformance to Dry Coating Thickness Requirements SSPC-SP1 Solvent Cleaning SSPC-SP 6 **Commercial Blast Cleaning** SSPC-SP7 **Brush-Off Blast Cleaning** SSPC-SP 10 Near White Metal Blast Cleaning SSPC-SP 11 Power Tool Cleaning to Bare Metal SSPC-SP 16 Brush-Off Blast Cleaning of Coated and Uncoated Galvanized Steel, Stainless Steels, and Non-Ferrous Metals. SSPC-Vis 1 Guide and Reference Photographs for Steel Surfaces Prepared by Dry Abrasive Blast Cleaning SSPC-Vis 3 Guide and Reference Photographs for Steel Surfaces Prepared by Power and Hand Tool Cleaning
- 2.1.3 Occupational Safety and Health Administration (OSHA)

OSHA 29 CFR 1910	Occupational Safety and Health Standards
OSHA 29 CFR 1926	Safety and Health Regulations for Construction

2.1.4 American Water Works Association

AWWA C210 Liquid-Epoxy Coating Systems for the Interior and Exterior of Steel Water Pipelines

2.2 Reference Documents/Drawings

24590-WTP-3DI-W12W-00004, Classification of Areas for Design; (formerly 24590-WTP-GPP-SRAD-007, Classification of Areas)

24590-WTP-3PS-NW00-T0002, Engineering Specification for Chemical Requirements for Materials Used in Contact with Austenitic Stainless Steel and Nickel Based Alloys

24590-WTP-3PS-PX04-T0001, Engineering Specification for Shop Applied Fusion Bonded Epoxy Coating for Underground Carbon Steel Pipe

24590-WTP-GPP-CON-7101, Construction Quality Control Program.

CCN 054843, DCN No. 24590-PTF-M6C-PWD-00047 Drain Collection System Line Calculations (Information Only)

24590-WTP-RPT-ENG-16-015, Thermally Insulated and High Density Polyethylene Jacketed Waste Transfer Line NACE SP-0169-2013 Evaluation.

24590-WTP-LIST-ESH-16-0001 Restricted Materials List Hanford Tank Waste Treatment and Immobilization Plant (WTP Project)

2.2.1 This specification revision (Rev. 5) Supersedes the following specifications, all change papers to these documents have also been incorporated.

24590-WTP-3PS-PX04-T0003 Revision #2 – Shop and Field Application of an Epoxy Coating for Underground Carbon Steel Fittings, Valves and Special Items

24590-WTP-3PS-PX04-T0004 Revision #1 – Epoxy Coating of Stainless Steel Items that are Buried, Embedded or Insulated

3 Materials

3.1 Abrasives

- 3.1.1 Abrasives for blast cleaning carbon steel surfaces shall be clean, free of oil or contamination, and dry. The particle size shall be capable of producing the specified surface profile. Mineral and Slag abrasive used on site shall meet the requirements of SSPC-AB 1 Class A (<1% Crystalline Silica).
- 3.1.2 Abrasives for blast cleaning stainless steel surfaces shall be clean, free of oil or contamination, dry and suitable for use on SS surfaces. Ferrous metallic abrasives such as carbon steel and chilled iron in either shot or grit form is **NOT** acceptable for use on SS surfaces. The particle size shall be capable of producing the specified surface profile.
- 3.1.3 Each batch or lot of mineral abrasive used on stainless steel shall be tested for water soluble contaminants and the conductivity shall not exceed 500 micro siemens when tested in accordance with ASTM D4940.
- 3.1.4 As an alternate, a chloride ion test kit such as the Chlor*Test "A" manufactured by Chlor-Rid International, Inc or approved equal may be used. The maximum allowable chloride level is 200 ppm.
- 3.1.5 Abrasives shall not be recycled. Pre-packaged abrasives do not require conductivity or chloride testing.

3.2 Coating Materials

3.2.1 Coating materials shall be one of the products as listed below or as approved by the BUYER. The coating manufacturer shall complete a Product Identity Certification Record (PICR) form as provided in Appendix A for each batch of material used per this scope of work as further defined in Section 9.2.2.5.

Material	Manufacturer	Generic Type	Use
Protal 7000	Denso	100% Solids Epoxy	Full application and repair work
SPC-3888	Specialty Polymer Coatings	Novolac Epoxy	Full application and repair work
Coal Tar Epoxy such as: Bitumastic 300M Amercoat 78 HB Targuard	Varies From: Carboline PPG Sherwin Williams	Coal Tar Epoxy	Use same material as originally applied to repair damage to existing Coal Tar Epoxy materials.
Nap-Gard 1847	Axalta (formerly DuPont)	Liquid Epoxy Coating	Repair of Axalta Nap-Gard 7-2500 or 7-2501 FBE coatings
Scotchkote 323*, Scotchkote 323+	3M	Liquid Epoxy Coating	Repair of Scotchkote 206N, 226N and 6233 FBE coatings and Axalta Nap-Gard 7-2500, 7-2501, 7-2599, 7-5208 FBE coatings

Table 1 - Coating Materials for Carbon Steel Substrates

* This product has been modified to remove potentially hazardous components, only 323+ will be available after $\sim 4^{\text{th}}$ quarter of 2019.

Table 2 -	Coating	Materials fo	r Stainless	Steel	Substrates
-----------	---------	--------------	-------------	-------	-------------------

Material*	Manufacturer	Use
Carboguard 890	Carboline	Full application and repair work
Macropoxy 646	Sherwin Williams	Full application and repair work

* Materials comply with project specification 24590-WTP-3PS-NW00-T0002 and contain <200ppm leachable halogen, <400ppm sulfur and only trace amounts of listed low melting point metals.

Table 3 - Repair	• Only	Coating	Materials
------------------	--------	---------	-----------

Material	Manufacturer	Generic Type	Use
Scotchkote 226P	3M	Patch Melt Stick	Repair of 3M Scotchkote 206N, 226N and 6233 FBE coatings
Protal 7125 or Protal 7200	Denso	Liquid Epoxy Coating*	Repair of Denso's Protal 7000

Page 7

3.3 Prohibited Materials

3.3.1 Refer to 24590-WTP-LIST-ESH-16-001 for a list of materials that are restricted from use on the River Protection Project – Waste Treatment Plant (WTP) Project job site. These materials may be used in shop applications provided the shop has appropriate personnel protection measures in place, their use does not violate local or federal regulations and they will not present a problem upon delivery to the project site.

4 Application

4.1 General

- 4.1.1 Surface preparation and the application of liquid epoxy shall be immediately stopped at any time when conditions exist that might adversely affect the quality. This includes but is not limited to contaminated surfaces, rusted surfaces, expired materials, rain, snow, condensation, deviations in environmental conditions, improper equipment or materials, non-qualified applicators or inspectors, poor workmanship, etc.
- 4.1.2 Prior to the start of surface preparation, sharp edges, slag, weld spatter or burrs shall be removed.

4.2 **Pre-Surface Preparation**

- 4.2.1 Prior to abrasive blasting or mechanical cleaning, the surfaces to be coated shall be cleaned in accordance with SSPC-SP 1, to remove oil, grease, dirt, and other foreign matter that can interfere with the proper bonding of the coating. Pneumatic tools shall not be used unless they are fitted with effective oil and water traps on the exhaust air.
- 4.2.2 For stainless steel substrates ferrous metallic (i.e. carbon steel) brushes, grinders, files or other such tools shall NOT be used. Only liquids containing less than 200 ppm halogen and/or demineralized water shall be used on SS surfaces. If SS items were shipped or stored so that the surface could have become contaminated with soluble salts (e.g. above deck ship transport, truck shipment on dirt roads close to the ocean or stored close to the ocean) the item shall be high pressure washed with demineralized water to remove as much soluble salt contamination as possible prior to abrasive blast cleaning.

4.3 Surface Preparation

- 4.3.1 Surface preparation for the installation of liquid epoxy material shall be based on the existing surface condition of the pipe or fitting. The following generic situations are addressed:
 - Bare Surfaces
 - Painted Surfaces (FBE, Liquid Epoxy, including Coal Tar Epoxy for CS and SS surfaces)

4.3.2 Bare Surfaces

- 4.3.2.1 For carbon steel surfaces abrasive blasting is the preferred surface preparation method and shall be performed in accordance with SSPC-SP 10 using abrasives as defined in Section 3.1. The surface profile shall be 1.0 3.0 mils for coatings applied to carbon steel surfaces.
- 4.3.2.2 For stainless steel surface abrasive blasting is the preferred surface preparation method and shall be performed in accordance with SSPC-SP 16, using abrasives as defined in Section 3.1. The surface profile shall be a minimum of 0.75 mils for coatings applied to stainless steel surfaces.
- 4.3.2.3 As an alternate, power tool cleaning weld areas and defective areas of the pipe, fittings and valves or exposed surface of installed bolting material in accordance with SSPC-SP 11 is acceptable. The surface profile shall be 1.0 mils minimum.

4.3.3 <u>Painted Surfaces</u>

- 4.3.3.1 Surfaces with existing coating that is to be replaced shall be completely cleaned to a bare surface condition as required in Section 4.3.2 of this specification.
- 4.3.3.2 Where the applied coating is to remain in place (e.g., coating of weld seams where the adjacent FBE coating or liquid epoxy is to remain or for local spot repairs), the coating shall be feathered back a minimum of one inch to one and one half inches (1.0"-1.5"). The feathered area shall be uniformly roughened by brush blast cleaning in accordance with SSPC-SP 7 or hand/power tool cleaning in accordance with SSPC-SP 3 or other approved method.

Note: Remedial work to defects is to be done per Section 4.5 of this specification. A defect that does not go to the substrate does not require feathering.

- 4.3.4 For Stainless Steel surfaces, ferrous metallic (i.e. carbon steel) shot/grit, carbon steel tools, discs, brushes or wheels shall not come into direct contact with stainless steel surfaces. The resulting surface shall be uniformly roughened.
- 4.3.5 After surface preparation is complete and before coating, pressurized air or a vacuum shall be used to remove all dust and abrasive residue. The pressurized air shall be clean and dry as verified in accordance with Section 5.1.3 so as not to contaminate the surface. If field conditions do not allow the use of pressurized air or a vacuum, repair surfaces may be wiped visibly clean of dust, dirt and other surface contaminants using clean lint free rags.
- 4.3.6 Any remaining sharp edges, weld spatter, burrs or porosity exposed after surface preparation and before coating application shall be removed.

4.4 Application of Liquid Epoxy

- 4.4.1 Environmental Conditions
- 4.4.1.1 The application of the liquid epoxy repair materials shall be performed only when the environmental conditions meet the following requirements:
 - The surface of the pipe shall be clean, dry and free of visible moisture.
 - The surface temperature shall be minimum of 5°F above the dew point
 - The relative humidity shall not exceed 90%.
 - The substrate and ambient temperature during coating application and curing shall be a minimum of 50°F and a maximum of 110°F or as defined by the coating manufacturer.
- 4.4.2 Liquid Epoxy Application
- 4.4.2.1 Full application of liquid epoxy coating shall be in accordance with the SELLERS approved procedures and the coating manufacturer's instructions and this specification. If there are any conflicts between the requirements, this specification shall take precedence.
- 4.4.2.2 Application of the coating repair material shall be performed only after acceptance of the surface preparation and uniform roughening of sound coating overlap area and verification of environmental conditions.
- 4.4.2.3 For repair work to provide a neat / uniform repair application mask an area approximately one to one and a half inches (1.0 1.5") inch beyond the repair bare areas (this will be within the feathered area as defined in Section 4.3.3.2). and apply the coating up to the tape. The tape is to be removed while the coating material is still wet and the coating allowed to flow onto the overlapped coating to provide a relatively smooth transition, not a sharp edge.
- 4.4.2.4 The liquid epoxy repair material must be in prepackaged units suitable in size for the repairs being performed or using a multi cartridge gun that automatically dispenses the correct ratio of prepackaged epoxy cure and resin components. Mixing of partial kits of material may be allowed if detailed mixing requirements for specific products are endorsed by the coating manufacturer and are accepted by the BUYER in an approved procedure, as further clarified in Section 6.3.2. Each component shall be premixed prior to being combined for use (cartridge or very viscous/stiff materials do not have to be premixed). The combined epoxy components must be thoroughly mixed, on a separate surface or in a suitable container, to a uniform appearance and consistency prior to application to the substrate.

- 4.4.2.5 The liquid epoxy material may be sprayed, rolled or brush applied in one (1) or more uniform coats to ensure achieving a dry film thickness as listed below. If the multiple coats are applied, the minimum and maximum drying times between coats shall be in accordance with the coating manufacturer's published data and shall be strictly observed.
- 4.4.2.6 FBE coated items and liquid applied epoxy coated items As a minimum, liquid epoxy repair material shall be allowed to cure until fingernail hard prior to performing DFT readings and holiday testing. The dry film thickness of the epoxy repair patch over the prepared bare surface shall be as defined In Table 4, overlap areas will be plus the thickness of the existing coating. All defects must be repaired and retested including holiday testing.

Coating Material	DFT (mils)					
Application to Carbon Steel						
Liquid Epoxy	Primary application	20.0 - 40.0				
Liquid Epoxy Touchup up of FBE and liquid epoxy		20.0 - 30.0				
Scotchkote 323 & Scotchkote 323+	Touchup up of 3M FBE materials	25.0 - 45.0				
Coal Tar Epoxy Touchup Coal Tar Epoxy		16.0 – 24.0				
Application to Stainl	ess Steel					
Liquid Epoxy	Primary application for buried or insulated lines	8.0 – 14.0 (applied in 2 uniform coats at 4-7 per coat)				
Liquid Epoxy Primary application for lines embedded in concrete		4.0-7.0				
Liquid Epoxy	Touchup of Epoxy on SS	8.0 - 14.0				

Table 4 - Coating Material DFT Requirements

4.4.2.7 Coating defects such as voids, drips, overspray, loss of adhesion, bubbling, peeling, fisheyes or inadequate cure are not allowed. Runs and sags greater than 30 mils thickness (45 mils for 3M Scotchkote 323+) (this is above the specified coating thickness so this spot on a SS pipe will be 38-44 mils thick (spec 8-14 + 30 mils)) shall be sanded down followed by applying a thin seal coat of liquid epoxy.

4.5 Remedial Work

4.5.1 Repairs to the epoxy material shall be performed by blasting, grinding or sanding off the defective area and applying a repair patch. Work shall be performed in accordance with Section 4.0.

- 4.5.2 Pin holes or other defects to the substrate shall be opened up to visibly expose the bare metal, preparation of the overlap area to be per Section 4.3.3.2 above with coating application per Section 4.4.
- 4.5.3 Repair work shall be allowed to cure and shall be inspected per the requirements of the original coating application.

5 Tests and Inspections

5.1 General

- 5.1.1 All necessary testing and inspection instruments shall be properly calibrated and maintained. Such equipment shall be available for use by BUYER authorized personnel conducting surveillance of the work.
- 5.1.2 Any defects disclosed during inspection shall be repaired and re-inspected after correction.
- 5.1.3 Prior to using compressed or pressurized air, the quality of the air downstream of the separator shall be tested in accordance with the requirements of ASTM D4285, this test shall be performed at the beginning of each shift and at approximately four (4) hour intervals. All lines shall be tested individually prior to use.
- 5.1.3.1 The test shall also be made after any interruption of the air compressor operation. The air shall be used only if the test indicates no visible contamination, oil, or moisture is present. If contaminants are evident, the equipment deficiencies shall be corrected, and the air stream shall be re-tested. Moisture separators shall be bled continuously.
- 5.1.3.2 Surfaces which are determined to have been blown down or blasted with contaminated air, shall be recleaned or reblasted with clean air and abrasive. Coatings, which have been determined to have been applied using contaminated air, shall be removed and reapplied using clean air.
- 5.1.4 Inspection points shall be established as follows:
 - Prior to start of work
 - Immediately following the surface preparation
 - Immediately prior to material application
 - Following the application of each coat of liquid epoxy coating
 - Following the sufficient curing of the coating perform visual inspections, DFT checks, and holiday testing of the coating
 - Final inspection and sign-off, in accordance with the project requirements

5.1.5 The following lists the required inspection steps and frequencies:

Step #	Coating/Inspection Step	Required Frequency
1	Prior to start of work	100% Visual for damage or defects in the substrate material that may impact coating work.
2	Surface Preparation / Profile	100% Visual on all items and
		Profile check on first item of each method and a minimum of every 20 items thereafter per shift
3	Environmental	At the start of work and every 4 hours thereafter and with sudden changes in weather conditions
4	During Application	100% visual and periodic WFT measurements by applicator
5.1	Following Application: Visual	100% of all items following application and initial cure.
5.2	Thickness	Per Section 5.4.3 for full application and A minimum of one gage reading on each spot repair and one spot reading for each area coated
5.3	Holidays	All coated surfaces 100% inspected.

Table 5 - Coating Inspection Frequency

5.2 Environmental Conditions Inspection

- 5.2.1 Air temperatures (dry and wet bulb) shall be determined using liquid in glass thermometer or an electronic temperature gauge. Relative humidity shall be calculated in accordance with ASTM E337 or read from a psychrometric chart. Dew point temperatures shall be read from a psychrometric chart. Alternatively, electronic instruments dedicated for temperature, dew point and humidity testing may also be used. Surface temperature shall be determined using a bimetallic surface thermometer or electronic surface thermometer/probe. The use of an Electronic Temperature / Humidity / Dew Point device is acceptable if it has current in-date calibration that is traceable to NIST standards.
- 5.2.2 Prior to the application of coating, the environmental conditions shall meet the requirements of Section 4.4.1.
- 5.2.3 All jobsite inspection work will follow the quality program requirements of 24590-WTP-GPP-CON-7101, Construction Quality Control Program.

5.3 Surface Preparation Inspection

5.3.1 Prior to abrasive blasting or mechanical cleaning, surfaces to be coated shall be free of oil, grease, dirt or other contaminants and metal defects detrimental to coating performance. (Refer to Section 4.2)

- 5.3.2 Surface preparation cleanliness shall be in accordance with the written description of the specific SSPC standard. Visual standards SSPC-VIS 1 and/or VIS 3 may be used to supplement the written description; however, the written description supersedes the visual standard in any dispute.
- 5.3.3 The blasted surface profile for carbon steel surfaces shall be from 1.0-3.0 mils; for stainless steel surfaces the profile shall be > 0.75 mils as measured in accordance with ASTM D4417 method A (e.g. visual comparator such as Keane-Tator profile comparator) or method C (e.g. Testex Press-O-Film tape and a spring micrometer). The power tool cleaned surface shall be uniformly roughened. Power tool cleaned surfaces prepared to SSPC-SP 11 shall have a 1.0 mil minimum profile.
- 5.3.4 Grease free chalk shall be used to mark local areas which do not meet the specified requirements (crayons, marking pens and soap stone shall not be used). When marking stainless steel items, the marker shall conform to the requirements of 24590-WTP-3PS-NW00-T0002.

5.4 Liquid Epoxy Inspection

- 5.4.1 Surface cleanliness shall be visually verified immediately prior to application of the liquid epoxy.
- 5.4.2 Applied liquid epoxy material for repair shall be inspected to verify specified overlap defined in Section 4.4.2.3, the minimum dry film thickness defined in Table 4, no visual defects such as runs, sags, embedded particles and that the epoxy material has cured hard.
- 5.4.3 Dry Film Thickness (DFT) Determination
- 5.4.3.1 DFT shall be measured with a DFT gage such as an Electro-Physick "Mikrotest", Positector 2000, Positector 6000 or approved equal. SSPC-PA 2 shall be used as a guide. The gage shall have a range that closely encompasses the thicknesses to be tested. DFT instruments shall be checked for visible damage and for accuracy each shift it is used. These accuracy checks shall be performed using certified coating thickness calibration standards for non-magnetic coating of steel traceable to NIST or approved alternative standards. The standards used shall be in a range that encompasses the actual thicknesses being tested.
- 5.4.3.2 The coating thickness determination procedures are based upon requirements of SSPC-PA 2. The SSPC-PA 2 "Gage Readings" and "Spot Measurements" are applicable. Specifically, a Gage Reading is a single instrument reading and a Spot Measurement is the average of at least three gage readings within a 1.5-inch diameter circle. Acquisition of more than three gage readings within a spot is permitted. Any unusually high or low gage readings that are not repeated consistently are discarded. The average of the acceptable gage readings is the spot measurement.

- 5.4.3.3 Locations of spot measurements, as a minimum, should be taken along accessible sections near the top and bottom axis of pipe that is to be holiday tested. Additional spot measurements in other circumferential locations are designated for larger diameter piping or as specified.
- 5.4.3.4 Coating Thickness Restriction level 2 (as defined in SSPC-PA 2) applies to this scope of work. That is spot average DFT readings cannot be less than specified and can be up to 20% greater than the specified maximum value, provided the overall average is within specification requirements.
- 5.4.3.5 Wet Film Thickness (WFT) readings may be substituted for DFT readings for applications to Stainless Steel and non-ferrous substrates (i.e. Aluminum, Copper, cobalt etc.). WFT shall be calculated based on the specified DFT and the coating material Volume Solids value as printed on the product data sheet, as follows:

WFT =
$$\frac{DFT \ x \ (100\% + \% \ thinner)}{\% \ Solids \ by \ Volume}$$

Pipe Length Diameter		Minimum Number of Spot Measurements	Interval Spacing**			
Carbon Steel Piping						
Short Sections*	Any Diameter	6 randomly spaced (3 near top and 3 near bottom)	Equidistant			
Up to 40 Foot	Up to 40 FootUp to 12 inches8 randomly spaced (4 near top and 4 near bottom)		2 on each 10' section			
Up to 40 Foot	14 to 24 inches	10 randomly spaced (5 near top and 5 near bottom)	2 on each 8' section			
Up to 40 Foot	> 24 inches	12 randomly spaced (3 near top/3 near bottom/3 along each side @ 90° and 180°)	3 on each 10' section			
Longer than 40 Foot	Up to 12 inches	8 randomly spaced (4 near top and 4 near bottom)	2 on each 25% of the length			
Longer than 40 Foot	14 to 24 inches	10 randomly spaced (5 near top and 5 near bottom)	2 on each 20% of the length			
Longer than 40 Foot	> 24 inches	12 randomly spaced (3 near top / 3 near bottom / 3 along each side @ 90° and 180°)	3 on each 25% of the length			
Holiday Repair Any Length	Any Diameter	1 spot reading at holiday (on bare substrate)	N/A			

Table 6 - Number and Locations of DFT Spot Measurements

Pipe Length	Diameter	Minimum Number of Spot Measurements	Interval Spacing**		
Carbon Steel Piping					
Girth Weld Coating Work	Up to 12 inches	Minimum 3 spots equidistant around circumference (on bare substrate) ***	Equidistant		
Girth Weld Coating Work > 12 inches		Minimum 4 spots equidistant around circumference	Equidistant		
Stainless Steel Piping					
Each 15 foot section or portion thereof	Any	4 randomly spaced (2 near top and 2 near bottom)	Equidistant		

* Short Sections = Less than 10-foot length or within 10 foot of valves, fittings, CP Connections or shrink sleeves.

** Equidistant if less than 10 feet apart.

*** Measurements to be taken only where new coating is applied direct to the substrate not where new coating overlaps existing coating.

5.5 Holiday Inspection

- 5.5.1 All applied liquid epoxy shall be 100% holiday tested per the applicable requirements of ASTM D5162, as summarized in Appendix B. No holidays are allowed.
- 5.5.2 All Pinholes or holidays detected shall be repaired in accordance with Section 4.5 of this Specification and retested for both film thickness and holidays.

6 Storage, Handling and Shipping

6.1 General

- 6.1.1 Liquid epoxy repair materials, thinners, cleaning solvents and other flammable materials shall be kept away from combustion sources and shall be stored in metal flammable material storage cabinets. Thinners and solvents shall be transported to the point of use in approved safety containers. Liquid epoxy repair materials may be transported to their point of use in their original prepackaged containers or cartons.
- 6.1.2 Coating materials shall be delivered in manufacturer's original unopened container or packaging. Each container or package shall be clearly identified with the manufacturer's name, product designation, batch or lot number, date of manufacture and shelf life expiration date.
- 6.1.3 Coating material shall be protected from moisture, direct sunlight and temperatures below 40°F or above 100°F unless otherwise allowed by the coating manufacturer's published instructions.

6.2 Shelf Life

- 6.2.1 Coating materials that are older than twenty-four (24) months from the date of manufacture or that exceed the manufacturer's shelf life, if less than twenty-four (24) months, shall not be used and shall be removed from the job site. Where the coating manufacturer's normal shelf life is greater than twenty-four (24) months, and the coating material can be shown to have been stored as specified herein, a one-time extension of three (3) months may be approved provided the criteria below are met. Coating materials that have not been stored or handled in accordance with this section may not have their shelf life extended.
- 6.2.1.1 The shelf life extension shall be based on coating manufacturer laboratory testing of retain samples taken at the time of manufacture or by testing a sample provided from the actual coating material in question. Where testing verifies an outdated coating material still complies with its original design criteria, it is acceptable for shelf life extension.
- 6.2.1.2 New expiration date stickers, provided by the coating manufacturer, shall be affixed to each container prior to release from HOLD. The stickers shall include the product number, batch/lot number, the new expiration date and be suitably marked to indicate that they came from the coating manufacturer.
- 6.2.1.3 A new Product Identity Certification Record (PICR) (see Appendix A) shall be provided by the coating manufacturer that includes the test results and specifically indicates it was provided to document shelf life extension including new expiration date.

6.3 Material Use

- 6.3.1 Coating material containers where the airtight seal has been broken or any of the contents are lost, shall not be used, except as noted below. Epoxy repair materials supplied in cartridge form may have ends capped and may be reused.
- 6.3.2 Coating material containers shall not be opened except for immediate use. If containers are sufficiently large, where minimal use of coating material is expected to result in considerable waste, kits may be sub-divided. Procedures for preparation of smaller kits, for each product, must be approved by Engineering on a case by case basis.
- 6.3.3 Unused material containers shall be returned to storage as soon as possible at the end of each workday. Materials left out for more than ten (10) hours in a storage area without environmental controls (temperature, direct sunlight, etc.) shall not be used and shall be removed from the job site.

7 Quality Assurance

7.1 General

- 7.1.1 The SELLER shall control the quality of items and services to meet the requirements of this Specification, applicable codes and standards, and associated procurement documents. The SELLER shall prepare and maintain documentation to provide evidence of compliance with approved procedures and this Specification. A copy of the documentation shall be included in the shipping documentation.
- 7.1.2 The SELLER, including any lower-tier organizations engaged by him shall be subject to surveillance inspection by the BUYER representative until completion or termination of the procurement. This surveillance inspection does not relieve the SELLER from the responsibility for conformance to the requirements of procurement documents and procedures.
- 7.1.3 The BUYER representative shall be provided with a schedule and shall be notified of all required inspection points prior to the scheduled date for coating activities.
- 7.1.4 All pre-established witness and hold points shall be witnessed by the BUYER unless a written waiver has been issued.
- 7.1.5 The BUYER representative shall be the final authority on the acceptability of surface preparation and material application. Any coating which, in the BUYER representative's judgement, has not been applied in conformance with this Specification, shall be rejected.
- 7.1.6 All coating materials shall be provided with a complete PICR as provided in Appendix A of this specification in accordance with Section 9.2.
- 7.1.7 All personnel shall receive training in the specific project coating requirements and the associated approved work procedures that are relevant to their individual work assignments.
- 7.1.8 The SELLER's coating inspectors shall have previous experience in multi-component epoxy coating. All training shall include the specific project coating requirements, ASTM, SSPC standards and other relevant standards including the approved work procedures.

8 **Configuration Management**

8.1 General

8.1.1 Configuration management is maintained by conformance to standards and procedures. The BUYER shall be promptly notified by SDDR of any potential deviation or conflict with the requirements. Work shall not proceed until a resolution is accepted and proper documentation is provided to the BUYER.

9 **Documentation and Submittals**

9.1 General

- 9.1.1 The SELLER shall provide completed Coating Manufacturer's Identity Certification Record (Appendix A) for every batch of coating material used.
- 9.1.2 The SELLER shall provide a record of all required daily inspections (Example Appendix C) that includes pre-surface preparation, compressed air cleanliness, environmental conditions, surface preparation and roughness, location of field repairs coated, coating application, visual inspection, dry film thickness, holiday testing and all touchup / repair of coating. This record shall include the coating and thinner materials used and the ID of the items coated to provide traceability.
- 9.1.3 All quality documentation shall be available for review by the BUYER representative within 24 hours from the time it is generated.
- 9.1.4 SELLER documentation forms or the way that the actual work will be documented shall be provided by the SELLER as part of the procedure's submittal for review by the BUYER. All original documentation generated during the coating work shall be submitted as defined in the MR and clarified below.
- 9.1.5 Requested deviations from a specification requirement shall be documented on a Supplier Deviation Disposition Request (SDDR) form and submitted for resolution by the BUYER. Unplanned discrepancies are to be documented on a Non Conformance Report (NCR) as applicable, for resolution by the BUYER.
- 9.1.6 A safety program shall be prepared by the SELLER and approved by the BUYER prior to commencing any coating work at the WTP project site.

9.2 Submittals

- 9.2.1 With the quotation:
- 9.2.1.1 The SELLER shall provide an overview of the services to be provided, including degree of surface preparation to be performed, coating material(s) to be applied, including DFT, inspections to be performed, and documentation to be generated. The proposal shall also include any other details to fully describe the scope of work.
- 9.2.1.2 The SELLER shall identify the specific coating products by manufacturer and catalog number/name and shall submit the coating manufacturer's latest published Product Data Sheet.

- 9.2.1.2.1 Conflicts, if any, between the SELLER's normal procedures, the coating manufacturer's recommendations and this Specification shall be brought to the attention of the BUYER for resolution.
- 9.2.1.2.2 The proposed coating shall be one of the approved materials listed in Section 3.2. Alternate materials may be considered if properly submitted for BUYER evaluation and written approval. The burden of proof to justify equality resides with the SELLER and shall include case histories and customer contacts, that include names and phone numbers to verify satisfactory past performance.
- 9.2.1.3 The SELLER shall supply a list of previous project case histories where they applied a two component coating similar to the materials specified herein. This list shall include the type of item coated, the project name, a project client contact name and phone number if possible, for verification.
- 9.2.2 Following Award / As defined in the MR:
- 9.2.2.1 SELLER shall submit a detailed written procedure that addresses material receiving, marking, storage, handling, surface preparation, environmental control, application, curing, inspection, testing, touch-up/repair, and proposed documentation forms. The procedure and documentation forms shall be submitted for the BUYER approval prior to the start of coating work.
- 9.2.2.2 The SELLER shall submit a daily inspection record as part of the work procedures that includes the required elements provided in Appendix C. (Refer to Section 7.1.1)
- 9.2.2.3 The SELLER shall provide records of personnel training and certification. (Refer to Section 1.5).
- 9.2.2.4 The SELLER shall submit a current copy of the Product Data Sheet, application instructions and SDS for each coating product to be applied.
- 9.2.2.5 The SELLER shall submit original coating manufacturer's Product Identity Certification Records (PICR) for every batch of coating material intended for use (Appendix A).
- 9.2.3 Final Documentation
- 9.2.3.1 The final documentation package shall have a cover page that identifies the project, contract number and the report issue date. The report shall contain a table of contents and organized sections that contain the original documents generated during the work.
- 9.2.3.2 There shall be a letter of certification signed by management certifying the work performed complied with all the contract quality requirements

Appendix A

Coating Manufacturer's Product Identity Certification Record

Appendix A Coating Manufacturer's Product Identity Certification Record

Project Name:	Coating Applicator:
Project Number:	Purchase Order No:
Project Location:	Contract Number:
Coating Manufacturer:	Generic Coating Type:
Product Name:	Product Number:

(For multi-component products, provide data for all components on one or more Appendix F forms). (Provide the standard range and actual batch values for each test)

TEST RESULTS		Component A Batch No.		Component B Batch No.	
Test	Method Used	Standard Range	Batch Actual	Standard Range	Batch Actual
Weight per Gallon					
Viscosity					
Flash Point (Typical)					
% Solids by Volume (Typical)					
Cure to recoat time @ 50F, 70F, & 90F (typical)					
Batch Size					
Date of Mfg.					
Shelf Life					
Expiration Date					

COMMENTS:

I hereby certify that the material described above was inspected and/or tested and found to comply with our company Quality Assurance requirements, effective at the time of manufacturing, prior to being released for shipment. These materials will meet or exceed the performance characteristics listed on the applicable

product data documentation in effect at the time of manufacture.

 Signed:
 Date:

 Title:
 Company:

Appendix B

Procedure for Holiday Testing

Appendix B Procedure for Holiday Testing

1. General Requirements

- 1.1. Holiday testing is to be performed based on the thickness of the coating that was applied, i.e. original application, girth weld coating application or holiday repair material. The thickness is to be determined where the coating is applied to the substrate only; every effort should be made to only measure the coating as applied direct to the substrate. If the DFT of the adjacent applied coating is taken into account an excessive voltage may be selected that can damage the newly applied material. In order to minimize the potential for damage holiday test voltage is to be selected based only on the newly applied coating material. Previously applied coatings have been tested by that applicator and are understood to be acceptable.
- 1.2. The high voltage holiday detection equipment shall be a Tinker and Razor Model AP-W holiday detector, or an approved equal with an audible alarm. The liquid epoxy repairs onto stainless steel need only be tested using a Low Voltage Wet Sponge Holiday Detector. Holiday testing instrument calibration and use shall meet the requirements of ASTM D5162.

2. Procedures for Coating Thickness Determination

2.1. Average coating thickness shall be determined as defined in main specification Section 5.4.3. Coating thickness values obtained shall be rounded to the nearest whole number for determining the required holiday test voltage.

3. Holiday Testing of Epoxy Coatings

- 3.1. Using the average DFT value as determined in Section 2.1 above select the holiday test method and voltage as defined below.
- 3.2. Thin film FBE and liquid epoxy coatings, with a DFT less than 20 mils, are to be holiday tested per the requirements of ASTM D5162, Method A (Low Voltage Wet Sponge Testing). A wetting agent such as kodaks Photoflow 200 is to be added to the test water at a rate of ½ oz / gallon.
- 3.3. Thick film coatings, coatings that are greater than 20 mils DFT, are to be tested in accordance with ASTM D5162 Method B (High Voltage Spark Test), Table B-1 below lists test voltages. Values in the table are rounded to the nearest 100V and are consistent with Method B of ASTM D5162. Values were selected based on prior specification requirements, and the lower value from the provided table or the calculation as discussed in ASTM D5162. This provides a voltage that is expected to not damage the applied coating; but is sufficient to bridge the air gap present in the pinhole within the full specified DFT range.

Note: When using certain holiday test equipment, typically older models, it may be necessary to use a slightly higher voltage than listed in Table B-1. Some equipment cannot be adjusted in 100 volt increments therefore; the next higher available setting should be selected.

3.4. An alternate method to calibrate spark test equipment is to adjust the test instrument to the proper voltage for the maximum coating thickness being tested. In selecting the inspection voltage, it is important to provide sufficient voltage to beak the air gap the exists at the holiday. The air gap will vary depending on the total applied film thickness. The voltage required to break a given air gap may also vary due to atmospheric conditions such as relative humidity and air pressure. Ensure that the voltage is high enough to break the air gap equivalent to the highest coating film thickness by separating the exploring electrode from the bare metal substrate using a nonconductive spacer equal to the maximum coating thickness. A sheet of plastic film with a created pinhole may be used for this purpose. The voltage is set high enough to conduct the holiday test only if the spark will jump the gap formed by the spacer. A hole is to be deliberately made in the plastic sheet to simulate a defect in a coating. Excessive voltage may cause a holiday to form in the coating film. This test is intended to isolate and identify defects in the coating and pipe while maintaining the integrity of the coating.

4. Holiday Testing of Previously Buried or Exposed Coatings/Linings

- 4.1. Holiday Testing is intended for use with new coatings/linings applied to metal substrates. Its use on coatings that have been previously buried or exposed to immersion conditions has often resulted in damage to the coating and has produced erroneous detection of discontinuities due to permeation or moisture adsorption of the coating. Deposits may also be present on the surface causing telegraphing (current traveling through a moisture path to a discontinuity, giving an erroneous indication) or current leakage across the surface of the coating due to contamination. The use of a high voltage tester on previously exposed coatings has to be carefully considered because of possible spark through, which will damage an otherwise sound coating. Although a low voltage tester can be used without damaging the coating, it may also produce erroneous results.
- 4.2. In situations where it may be necessary to check for holiday on piping that has been previously buried, clean the surface and allow to dry-out for a minimum of 3 days. It will be necessary to determine the appropriate test voltage for the measured coating thickness being tested and then set the equipment to a test voltage that is 1,000 volts lower, to compensate for the possible presence of retained moisture. This adjustment is based upon test data from the Epoxy Coating Manufacturer's test data (3M Corrosion Protection Product Division CPPD Austin R&D) for dielectric failure voltage and moisture absorption of the coating.
| DFT | Voltage | DFT | Voltage | DFT | Voltage | DFT | Voltage |
|-----|---------|-----|---------|-----|---------|-----|---------|
| 20 | 2,500 | 38 | 3,200 | 56 | 7,500 | 74 | 10,000 |
| 21 | 2,500 | 39 | 3,300 | 57 | 7,500 | 75 | 10,800 |
| 22 | 2,500 | 40 | 5,500 | 58 | 7,500 | 76 | 10,800 |
| 23 | 2,500 | 41 | 5,500 | 59 | 8,000 | 77 | 10,800 |
| 24 | 2,600 | 42 | 5,500 | 60 | 8,000 | 78 | 10,800 |
| 25 | 2,600 | 43 | 5,500 | 61 | 8,000 | 79 | 11,100 |
| 26 | 2,700 | 44 | 6,000 | 62 | 8,000 | 80 | 11,200 |
| 27 | 2,700 | 45 | 6,000 | 63 | 8,500 | 81 | 11,300 |
| 28 | 2,800 | 46 | 6,000 | 64 | 8,500 | 82 | 11,300 |
| 29 | 2,800 | 47 | 6,000 | 65 | 8,500 | 83 | 11,400 |
| 30 | 2,900 | 48 | 6,500 | 66 | 8,500 | 84 | 11,500 |
| 31 | 2,900 | 49 | 6,500 | 67 | 9,000 | 85 | 11,500 |
| 32 | 3,000 | 50 | 6,500 | 68 | 9,000 | 86 | 11,600 |
| 33 | 3,000 | 51 | 6,500 | 69 | 9,000 | 87 | 11,700 |
| 34 | 3,100 | 52 | 7,000 | 70 | 9,000 | 88 | 11,700 |
| 35 | 3,100 | 53 | 7,000 | 71 | 10,000 | 89 | 11,800 |
| 36 | 3,200 | 54 | 7,000 | 72 | 10,000 | 90 | 11,900 |
| 37 | 3,200 | 55 | 7,000 | 73 | 10,000 | | |

 Table B-1

 Holiday Detection Spark Test Voltages for Coatings *

*Based on ASTM D5162-15, "Standard Practice for Discontinuity (Holiday) Testing of Nonconductive Protective Coating on Metallic Substrates"

Table B-1 Notes:

- 1. The surfaces must be clean, dry and fee of oil, dirt and other contaminants.
- 2. Sufficient drying or curing of the coating is to be allowed prior to holiday testing. The length of time required shall be a minimum of 1 days, or as defined by the coating manufacturer. This is to ensure that solvents retained in the coating film do not produce erroneous indicators.
- 3. Coatings that contain electrically conductive fillers or pigments are not suitable for wet sponge or high voltage holiday testing.
- 4. Cathodic Protection Systems should not be in operation when spark testing is performed.

24590-WTP-3PS-PX04-T0002, Rev 5 Application of Liquid Epoxy Coating for Piping Applications

Appendix C

Inspection Form

Daily Work Inspection Form

^PROJECT:	REPORT NO:						^DAT	E:					SHI	FT:		
SPEC NO/REV: ^FEQUIPMENT/AREA: ^SELLER: SUBSTRATE: STEEL/CONCRETE/OTHER: *SELLER: SUBSTRATE: STEEL/CONCRETE/OTHER: *ORK ACTIVITY TIME DRY BULB ^*WET BULB RH % DEW SURFACE BLOTTER *WORK ACTIVITY TIME DRY BULB ^*WET BULB RH % DEW SURFACE BLOTTER WORK ACTIVITY TIME DRY BULB ^*WET BULB RH % DEW SURFACE BLOTTER WORK ACTIVITY TIME DRY BULB ^*WET BULB RH % DEW SURFACE BLOTTER SURFACE PREPARATION: Image: Steelee Ste	^PROJECT:						DAY:	1	M	Т	W	Т	F	S	S	
^SELLER: SUBSTRATE: STEEL / CONCRETE / OTHER: ^ENVIRONMENTAL CONDITIONS: DRY BULB ^*WET BULB RH % DEW SURFACE BLOTTER WORK ACTIVITY TIME DRY BULB ^*WET BULB RH % DEW SURFACE BLOTTER WORK ACTIVITY TIME DRY BULB ^*WET BULB RH % DEW SURFACE BLOTTER PRE-SURFACE Image: Complement of the state of the st	SPEC NO/REV:						^EQU	IPMENT/	ARI	EA:						
Pervironmental conditions: DRY BULB ^*WET BULB RH % DEW POINT °F SURFACE BLOTTER TEMP. °F WORK ACTIVITY TIME DRY BULB ^*WET BULB RH % DEW POINT °F SURFACE BLOTTER TEMP. °F TEMP. °F	^SELLER:						SUBS	TRATE:		ST	EEL	/ CC	NCF	RET	E / OTI	HER:
WORK ACTIVITY TIME DRY BULB TEMP. *F A*WET BULB TEMP. *F RH % DEW POINT *F SURFACE TEMP. *F BLOTTER TEMP. *F Image: Second Se	^ENVIRONMEN	TAL CO	ONDITION	<u>NS</u> :												
PRE-SURFACE PREPARATION: PRE-SURFACE PREPARATION: SURFACE DEFECTS: Yes / No DEFECTS ADDRESSED: Yes / No / N/A ^^SP-1 COMPLETE: MASKING / PROTECTION COMPLETE: SURFACE PREPARATION: CLEANLINESS ACTUAL: CLEANLINESS SPEC: ^CLEANLINESS ACTUAL: PROFILE SPEC: ^PROFILE ACTUAL: METHOD: STORAGE: MATERIALS & MIXING: STORAGE: ^PRODUCT(S): PROPERLY STORED: EXPIRATION DATE: ^^ ^ABATCH NO(S): A: B: C: ^THINNER: BATCH NO.: BATCH MIX TIME(S): MIXED TEMPERATURE: INDUCTION TIME: POT LIFE: APPLICATION START TIME: ^2WFT: TIME TO RECOAT* (Min/Max): * Based on ambient temp during / following application EQUIPMENT:	WORK ACTIV	ITY	TIME	DRY BU TEMP	ULB .ºF	^{^1} WE TEN	T BULB AP. °F	RH %	PC	DEV)IN'	V Г°F	SU TE	RFA MP.	CE °F	BLOT TE	TER ST
PRE-SURFACE PREPARATION: SURFACE DEFECTS: Yes / No DEFECTS ADDRESSED: Yes / No / N/A ^SP-1 COMPLETE: MASKING / PROTECTION COMPLETE: SURFACE PREPARATION: CLEANLINESS SPEC: CLEANLINESS SPEC: ^CLEANLINESS ACTUAL: PROFILE SPEC: ^PROFILE ACTUAL: METHOD: SIZE: MATERIALS & MIXING: STORAGE: ^PRODUCT(S): PROPERLY STORED: EXPIRATION DATE: B: ^ BATCH NO(S): A: BATCH NO(S): A: BATCH NO(S): A: BATCH NO(S): MIXED TEMPERATURE: INDUCTION TIME: POT LIFE: APPLICATION: MIXED SECOND T.U. ^COAT: PRIMER T.U. SECOND T.U. THIRD T.U. ^APPLICATION START TIME: ^APPLICATION START TIME: ^APPLICATION METHOD: EQUIPMENT: * Based on ambient temp during / following application					-					/				-		
PRE-SURFACE PREPARATION: SURFACE DEFECTS: Yes / No DEFECTS ADDRESSED: Yes / No / N/A ^SP-1 COMPLETE: MASKING / PROTECTION COMPLETE: SURFACE PREPARATION: CLEANLINESS ACTUAL: ABRASIVE TYPE: CLEANLINESS SPEC: ^CLEANLINESS ACTUAL: ABRASIVE TYPE: PROFILE SPEC: ^PROFILE ACTUAL: SIZE: METHOD: STORAGE: MATERIALS & MIXING: STORAGE: ^PRODUCT(S): PROPERLY STORED: EXPIRATION DATE: ^ BATCH NO(S): A: B: C: ^THINNER: BATCH NO.: THINNING RATIO: BATCH MIX TIME(S): MIXED TEMPERATURE: INDUCTION TIME: INDUCTION TIME: POT LIFE: POT LIFE: APPLICATION: START TIME: FINISH TIME: ^2WFT: TIME TO RECOAT* (Min/Max): TIME TO FULL CURE*: APPLICATION METHOD: EQUIPMENT: * Based on ambient temp during / following application																
PRE-SURFACE PREPARATION: SURFACE DEFECTS: Yes / No SURFACE DEFECTS: Yes / No DEFECTS ADDRESSED: Yes / No / N/A ^SP-1 COMPLETE: MASKING / PROTECTION COMPLETE: SURFACE PREPARATION: CLEANLINESS SPEC: ^CLEANLINESS SPEC: ^CLEANLINESS ACTUAL: ABRASIVE TYPE: BROFILE ACTUAL: PROFILE SPEC: ^PROFILE ACTUAL: METHOD: STORAGE: MATERIALS & MIXING: STORAGE: MATERIALS & MIXING: STORAGE: ^PRODUCT(S): PROPERLY STORED: EXPIRATION DATE: ^ BATCH NO(S): A: B: C: ^THINNER: BATCH NO.: THINNING RATIO: BATCH MIX TIME(S): MIXED TEMPERATURE: INDUCTION TIME: INDUCTION TIME: POT LIFE: APPLICATION: POT LIFE: ^2OAT: PRIMER T.U. SECOND SECOND T.U. THIRD THIRD T.U. OTHER ^APPLICATION START TIME: FINISH TIME: ^2WFT: TIME TO RECOAT* (Min/Max): TIME TO FULL CURE*: APPLICATION METHOD: EQUIPMENT: *Based on anbient temp during / following application																
SURFACE DEFECTS: Yes / NoDEFECTS ADDRESSED: Yes / No / N/A^SP-1 COMPLETE:MASKING / PROTECTION COMPLETE:SURFACE PREPARATION:CLEANLINESS ACTUAL:CLEANLINESS SPEC:^CLEANLINESS ACTUAL:ABRASIVE TYPE:SIZE:PROFILE SPEC:^PROFILE ACTUAL:METHOD:STORAGE:MATERIALS & MIXING:STORAGE:^PRODUCT(S):PROPERLY STORED:* BATCH NO(S):A:B:C:^THINNER:BATCH NO.:BATCH MIX TIME(S):MIXED TEMPERATURE:INDUCTION TIME:POT LIFE:APPLICATION:START TIME:^COAT:PRIMER T.U.SECOND SECOND T.U.THIRD T.U.^APPLICATIONSTART TIME:^2WFT:TIME TO RECOAT* (Min/Max):APPLICATION METHOD:EQUIPMENT:* Based on ambient temp during / following application* DW MO	PRE-SURFACE P	REPA	RATION:													
^SP-1 COMPLETE: MASKING / PROTECTION COMPLETE: SURFACE PREPARATION: CLEANLINESS ACTUAL: ABRASIVE TYPE: CLEANLINESS SPEC: ^CLEANLINESS ACTUAL: ABRASIVE TYPE: PROFILE SPEC: ^PROFILE ACTUAL: SIZE: METHOD: STORAGE: STORAGE: MATERIALS & MIXING: PROPERLY STORED: EXPIRATION DATE: ^PRODUCT(S): PROPERLY STORED: EXPIRATION DATE: ^ PRODUCT(S): A: B: C: ^THINNER: BATCH NO.: THINNING RATIO: BATCH MIX TIME(S): MIXED TEMPERATURE: INDUCTION TIME: INDUCTION TIME: POT LIFE: POT LIFE: APPLICATION: TIME TO RECOAT* (Min/Max): TIME TO FULL CURE*: ^2WFT: TIME TO RECOAT* (Min/Max): TIME TO FULL CURE*: APPLICATION METHOD: EQUIPMENT: * Based on ambient temp during / following application	SURFACE DEF	ECTS:	Yes / N	0			DEFEC	TS ADDF	RESS	SED	: Ye	es /	No /	N/2	A	
SURFACE PREPARATION: CLEANLINESS SPEC: ^CLEANLINESS ACTUAL: ABRASIVE TYPE: PROFILE SPEC: ^PROFILE ACTUAL: SIZE: METHOD: STORAGE: MATERIALS & MIXING: STORAGE: ^PRODUCT(S): PROPERLY STORED: EXPIRATION DATE: ^ ABATCH NO(S): A: B: C: ^THINNER: BATCH NO.: THINNING RATIO: BATCH MIX TIME(S): MIXED TEMPERATURE: INDUCTION TIME: INDUCTION TIME: POT LIFE: POT LIFE: APPLICATION: START TIME: FINISH TIME: ^2WFT: TIME TO RECOAT* (Min/Max): TIME TO FULL CURE*: APPLICATION METHOD: EQUIPMENT: * Based on ambient temp during / following application	^SP-1 COMPLE	TE:					MASKI	NG / PRC)TE	CTI	ON C	OM	PLET	E:		
CLEANLINESS SPEC: ^CLEANLINESS ACTUAL: ABRASIVE TYPE: PROFILE SPEC: ^PROFILE ACTUAL: SIZE: METHOD: STORAGE: MATERIALS & MIXING: STORAGE: ^PRODUCT(S): PROPERLY STORED: EXPIRATION DATE: ^ PRODUCT(S): A: B: C: ^ THINNER: BATCH NO.SI A: C: ^THINNER: BATCH NO.SI THINNING RATIO: BATCH MIX TIME(S): MIXED TEMPERATURE: INDUCTION TIME: NDUCTION TIME: POT LIFE: POT LIFE: APPLICATION: START TIME: FINISH TIME: ^2WFT: TIME TO RECOAT* (Min/Max): TIME TO FULL CURE*: APPLICATION METHOD: EQUIPMENT: * Based on ambient temp during / following application	SURFACE PREP	ARATI	ON·													
PROFILE SPEC: ^PROFILE ACTUAL: SIZE: METHOD: STORAGE: MATERIALS & MIXING: STORAGE: ^PRODUCT(S): PROPERLY STORED: EXPIRATION DATE: ^ BATCH NO(S): A: B: C: ^THINNER: BATCH NO.: THINNING RATIO: BATCH MIX TIME(S): MIXED TEMPERATURE: NDUCTION TIME: INDUCTION TIME: POT LIFE: APPLICATION: * FINISH TIME: ^COAT: PRIMER T.U. SECOND T.U. THIRD T.U. ^APPLICATION START TIME: FINISH TIME: ^2WFT: TIME TO RECOAT* (Min/Max): TIME TO FULL CURE*: APPLICATION METHOD: EQUIPMENT: *	CLEANLINESS	SPEC	:	^CLEA	ANLI	NESS A	ACTUAL	•	A	BR	ASIV	ΈT	YPE:			
METHOD: STORAGE: MATERIALS & MIXING: * ^PRODUCT(S): PROPERLY STORED: EXPIRATION DATE: ^PRODUCT(S): A: B: C: ^ABATCH NO(S): A: B: C: ^THINNER: BATCH NO.: THINNING RATIO: BATCH MIX TIME(S): MIXED TEMPERATURE: NUXED TEMPERATURE: INDUCTION TIME: POT LIFE: APPLICATION: * POT LIFE: ^COAT: PRIMER T.U. SECOND SECOND T.U. THIRD T.U. OTHER ^APPLICATION START TIME: FINISH TIME: * * ^2WFT: TIME TO RECOAT* (Min/Max): TIME TO FULL CURE*: APPLICATION METHOD: EQUIPMENT: * Based on ambient temp during / following application + + + + * Based on ambient temp during / following application + + + + + * Based on ambient temp during / following application + + + + +	PROFILE SPEC	:		^PROI	FILE	ACTU	AL:					SIZ	ZE:			
MATERIALS & MIXING: ^PRODUCT(S): PROPERLY STORED: EXPIRATION DATE: ^ BATCH NO(S): A: B: C: ^ THINNER: BATCH NO.: THINNING RATIO: BATCH MIX TIME(S): MIXED TEMPERATURE: INDUCTION TIME: INDUCTION TIME: POT LIFE: APPLICATION: POT LIFE: ^COAT: PRIMER T.U. SECOND SECOND T.U. ^ APPLICATION START TIME: FINISH TIME: ^2WFT: TIME TO RECOAT* (Min/Max): TIME TO FULL CURE*: APPLICATION METHOD: EQUIPMENT: * Based on ambient temp during application	METHOD:										STO	ORA	GE:			
^PRODUCT(S): PROPERLY STORED: EXPIRATION DATE: ^ BATCH NO(S): A: B: C: ^THINNER: BATCH NO.: THINNING RATIO: BATCH MIX TIME(S): MIXED TEMPERATURE: MIXED TEMPERATURE: INDUCTION TIME: POT LIFE: APPLICATION: POT LIFE: ^COAT: PRIMER T.U. SECOND SECOND T.U. ^ APPLICATION START TIME: FINISH TIME: ^2WFT: TIME TO RECOAT* (Min/Max): TIME TO FULL CURE*: APPLICATION METHOD: EQUIPMENT: * Based on ambient temp during / following application	MATERIALS & N	AIXIN	G:													
^ BATCH NO(S): A: B: C: ^THINNER: BATCH NO.: THINNING RATIO: BATCH MIX TIME(S): MIXED TEMPERATURE: MIXED TEMPERATURE: INDUCTION TIME: POT LIFE: POT LIFE: APPLICATION: POT LIFE: POT LIFE: ^COAT: PRIMER T.U. SECOND T.U. THIRD T.U. OTHER ^2WFT: TIME TO RECOAT* (Min/Max): TIME TO FULL CURE*: APPLICATION METHOD: EQUIPMENT: * *	^PRODUCT(S):				PRO	PERLY	STOREI	D:	E	XPI	RAT	ION	DAT	ΓE:		
^THINNER: BATCH NO.: THINNING RATIO: BATCH MIX TIME(S): MIXED TEMPERATURE: MIXED TEMPERATURE: INDUCTION TIME: POT LIFE: POT LIFE: APPLICATION: SECOND T.U. THIRD T.U. OTHER ^COAT: PRIMER PRIMER T.U. SECOND T.U. THIRD T.U. OTHER ^APPLICATION START TIME: FINISH TIME: ITIME TO FULL CURE*: ^2WFT: TIME TO RECOAT* (Min/Max): TIME TO FULL CURE*: APPLICATION METHOD: EQUIPMENT: TIME TO FULL CURE*:	^ BATCH NO(S):	1	4:			B:					C:				
BATCH MIX TIME(S): MIXED TEMPERATURE: INDUCTION TIME: POT LIFE: APPLICATION: POT LIFE: ^COAT: PRIMER PRIMER T.U. SECOND T.U. THIRD T.U. OTHER ^COAT: PRIMER PRIMER T.U. SECOND T.U. THIRD T.U. OTHER ^2WFT: TIME TO RECOAT* (Min/Max): TIME TO FULL CURE*: APPLICATION METHOD: EQUIPMENT: * Based on ambient temp during / following application	^THINNER:				BAT	CH NO	.:		Т	HIN	ININ	GR	ATIC):		
INDUCTION TIME: POT LIFE: APPLICATION: FINER PRIMER T.U. SECOND SECOND T.U. THIRD THIRD T.U. OTHER ^COAT: PRIMER PRIMER T.U. SECOND SECOND T.U. THIRD THIRD T.U. OTHER ^APPLICATION START TIME: FINISH TIME: FINISH TIME: ^2WFT: TIME TO RECOAT* (Min/Max): TIME TO FULL CURE*: APPLICATION METHOD: EQUIPMENT: * Based on ambient temp during / following application	BATCH MIX T	IME(S)):				MIXED	TEMPE	RAT	'URI	E:					
APPLICATION: ^COAT: PRIMER PRIMER T.U. SECOND T.U. THIRD T.U. OTHER ^APPLICATION START TIME: FINISH TIME: ^2WFT: TIME TO RECOAT* (Min/Max): TIME TO FULL CURE*: APPLICATION METHOD: EQUIPMENT: * Based on ambient temp during / following application	INDUCTION T	ME:					POT LI	FE:								
^COAT: PRIMER PRIMER T.U. SECOND THIRD THIRD T.U. OTHER ^ APPLICATION START TIME: FINISH TIME: <td>APPLICATION:</td> <td></td>	APPLICATION:															
^ APPLICATION START TIME: FINISH TIME: ^2WFT: TIME TO RECOAT* (Min/Max): TIME TO FULL CURE*: APPLICATION METHOD: EQUIPMENT: * Based on ambient temp during / following application EQUIPMENT:	^COAT: PRIM	1ER	PRIMER	RT.U.	SEC	OND	SECON	D T.U. '	THI	RD	TH	HIRD) T.U	. (OTHEF	2
^2WFT: TIME TO RECOAT* (Min/Max): TIME TO FULL CURE*: APPLICATION METHOD: EQUIPMENT: * Based on ambient temp during / following application Following application	^ APPLICATIO	N			STA	RT TIM	/IE:		F	FINI	SH T	IME	:			
APPLICATION METHOD: EQUIPMENT: * Based on ambient temp during / following application	^2WFT:		TIME TO	O RECOA	AT* (Min/M	ax):		Г	TIM	Е ТО	FUI	LC	URE	·*:	
* Based on ambient temp during / following application	APPLICATION	METH	IOD:				EQUIPM	IENT:	•							
	* Based on ambient te	emp duri	ng / followin	g applicatio	on											
APPLIED COATING:	APPLIED COATI	<u>NG</u> :														
^VISUAL INSPECTION (IMPERFECTIONS FOUND): Y / N TYPE OF IMPERFECTION:	^VISUAL INSPEC	TION (IMPERFE(CTIONS F	OUN	D): Y / I	N TYF	PE OF IMP	ERF	ECT	<u>'ION:</u>	<u> </u>				
DRY FILM THICKNESS: SPEC: ^ACTUAL: EQUIP. USED:	DRY FILM THICK	NESS:	SPEC:		^	ACTUA	AL:		EQ	UIP.	USE	D:				
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	TOUCH UP AND	REDAT	R COMPLI	TE· V	/ N /	# FUUP N/A			EQ CP	UIP.	USE FD	ט: יע				
					. 11 /	11/11					<i>D</i> .					

(Use reverse side or attach extra pages as needed)

INSPECTOR'S PRINTED NAME / SIGNATURE / DATE

Note: while documentation of a data point may not be required for submittal in accordance with this form the supplier is expected to be able to prove that all requirements of this specification have been met.

^ Mandatory data entry. Other entries should be completed where information is available.

^{^1} Wet bulb not required when accepted equipment automatically calculates dew point and humidity.

^{^2} Required only for application to concrete substrates, may be used as a substitute for DFT on non-ferrous substrates.

^{^3} Required only when applying coatings for immersion service and underground applications



	_		
		R11874321	
N.S.	OTY	BECHTEL STK CODE	
3 3	75' 0' 4' 9 '	PPPCØBØGØXØ2 PPPCØBØGØXØ2	
3	1	PFQNV6ØG16ØG	
N. S.		BECHTEL	
(INCHES)	<u>OTY</u>	STK CODE	
	1 1 1 1 1		
	N. S. INCHES)	N. S. <u>INCHES</u>) <u>OTY</u> 3 75' 0' 3 1 N. S. <u>INCHES</u>) <u>OTY</u> 1 1 1 1 1 1 1	N. S. OTY BECHTEL INCHES) OTY STK CODE 3 75' 0' PPPCØBØGØXØ2 3 1 PFONV6ØG1 6ØG 3 1 PFONV6ØG1 6ØG N. S. OTY BECHTEL INCHES) OTY BECHTEL 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1

DOUBLE CONTAINED PIPE INNER CORE SEE SHEET 051 FOR OUTER JACKET

ISSUED BY RPP-WTP PDC RLD-ZS66991001.DGN

-P.4-

3947\24590-BOF

							12
1							ms(
18	REV SUPT NOTE, CLINE	SH	dss	CF	DWK	H.	P <
ŝ	ISSUED FOR CONSTRUCTION	JS	кк	CC	SH	JMC	ldo
	DESCRIPTION	DRN	CHKD	STRESS	RVWD	APVD	nbo
	REVISION HISTORY						1×
	RIVER PR WASTE TR 2435 STE RICHLAND,	OTECT REATME VENS , WA	ION PR NT PL CENTE 99354	ROJECT ANT R PLACE			c:\wor
	CONTRACT No: DE-AC27-	01RV	14136				NAL
-P1	F-M6-PWD-00058002 CONST. W	ORK A	REA: 2	25G			хFI
	BALANCE OF FACI ISOMETRIC	LIT	IES				PE: BOF_
LD	-ZS-66991-W31A-3	P	IPE SF	EC: W3	81 A		Υ
)-B	OF-P3-RLD-ZS66991001				REV	1	IS(
Y:lr	pholdow PLOT DATE: 1/22/2018	10	1:37:12	ΔM	SHT	1 OF	1



E BØGØXØ BØGØXØ L DE
IL IDE BØGØXØ BØGØXØ L DE
BØCØXØ BØGØXØ L DE
BØGØXØ: BØGØXØ: L DE
DE
L DE
З
10
Yk
J L APVD
JK. APVD
yk APVD
JK APVD
J/L APVD
JK APVD
APVD
APVD



WITCH N. P.				
COMPONENT DESCRIPTION	N.S.	DTY	BECHTE	L
	Tananear		0111 201	
2TP316/316L, SMLS, SCH40S, BBE	3	117'Ø	PPPCØ	BØGØXØ2
MATERIALS				
COMPONENT DESCRIPTION	N.S.	OTY	BECHTEL STK COL	DE
S BOE-BLD-H00195 WEIDED PIPE ATTCHMT				
BOF-RLD-H00196, WELDED PIPE ATTCHMT BOF-RLD-H00197, WELDED PIPE ATTCHMT		i		
BOF-RLD-H00199, WELDED PIPE ATTCHMT BOF-RLD-H00200, WELDED PIPE ATTCHMT		11		
BUF-RLU-HUUZUI, WELDED PIPE ATTCHMT		1		
NTENTS OF THIS DOC	UME	NT A	RE	
EROUS WASTE PERM	AIT AF	FEC	TINC	3
in the second				
LIDIE CONTAIL	IFD	DI	DE	
UBLE CONTAIL	NED	PI	PE	
UBLE CONTAII	NED E SF	PI	PE	
UBLE CONTAII NER CORE SEI 3 FOR OUTER	NED E SH	PI	PE	
UBLE CONTAIL NER CORE SEI 3 FOR OUTER	NED E SH JAC	PI	PET	
UBLE CONTAII NER CORE SEI 3 FOR OUTER	NED E SH JAC	PI HEE KE	PET	
UBLE CONTAIL NER CORE SEI 3 FOR OUTER	NED E SH JAC	PI	PET	
UBLE CONTAIL NER CORE SEI 3 FOR OUTER Reviewed No Constructions Org Name: Pipe Stress & Support- Paced: May 13, 2016, 1:27 pm		PI HEE KE	PE T T	
UBLE CONTAIL NER CORE SEI 3 FOR OUTER		PI HEE KE	PE T T	
UBLE CONTAIL NER CORE SEI 3 FOR OUTER		PI HEE KE	PET	
UBLE CONTAIL NER CORE SEI 3 FOR OUTER Reviewed No Comments Br Chritie Charles Dr Charles D			PET	
UBLE CONTAIL NER CORE SEI 3 FOR OUTER Reviewed No Comments Dr Christie Charles Hange Dr Christie Charles Hange Handel Hange Top Stress & Support- Paced: May 13, 2016, 1:27 pm			PETT	
UBLE CONTAIL NER CORE SEI 3 FOR OUTER Reviewed No Comments By: Cirstine Charge - Vestings Dry Comments By: Cirstine Charge - Vestings Dry Comments By: Cirstine Charge - Vestings Dry Comments Dry Comm			PE T T	JK
UBLE CONTAIL NER CORE SEI 3 FOR OUTER Reviewed No Comments Dig Name: Pipe Stress & Support Read: May 13, 2016, 127 pm		PI	PE T T SH, s RVWD	APVD
UBLE CONTAIL NER CORE SEI 3 FOR OUTER Reviewed No comments By Christic Orage Stress & Support- Paced: May 13, 2016, 127 pm		PI	PE T T S S RVWD	SK
UBLE CONTAIL NER CORE SEI 3 FOR OUTER Reviewed No Comments By: Christine Change - Istings Dry Tomme: Provide & Support- Paced: May 13, 2016; 1:27 pm UMUNIN 7057 ISSUED FOR CONSTRUCTION DESCRIPTION REVISION HISTORY REVISION HISTORY	NED Shares	PI	PE T T	SK
UBLE CONTAIL NER CORE SEI 3 FOR OUTER Reviewed No Comments By Christie	NED Shares	PI HEE KE	PE T T S RVWD	JÆ
UBLE CONTAIL NER CORE SEI 3 FOR OUTER Providence June Description REVISION HISTORY REVISION HISTORY REVISION HISTORY REVISION HISTORY RIVER PRO WASTE TR 2435 STEV RICHLAND, CONTRACT NO: DE-AC27-C	NED Shares	PI HEE KE SKE	PE T T S RVWD	APVD
UBLE CONTAIL NER CORE SEA 3 FOR OUTER Reviewed No Comments By: Christing Charge Vesses By: Christian Charge Vesse	NED SAC	PI HEE KE	PE T T	JK
UBLE CONTAIL NER CORE SEA 3 FOR OUTER Reviewed No Comments By: Grainer Charge Grant Construction ISSUED FOR CONSTRUCTION DESCRIPTION REVISION HISTORY REVISION HISTORY RIVER PRO WASTE TRI 2435 STE RICHLAND, CONTRACT No: DE-AC27-C TF-M6-PWD-00058002 BALANCE OF FACIL ISOMFTRIC	NED Shares	PI HEE KE SKE	PE T T	JK
UBLE CONTAIL NER CORE SEA 3 FOR OUTER Reviewed No Comments By Chame Provide & Suppor- Paced: May 13, 2016, 1:27 pm	Reston Prection P VERSIGN DRN CHKI DRN CHKI	PI HEE KE SKE		Sk
UBLE CONTAIL NER CORE SEA 3 FOR OUTER Reviewed No Comments US Office of the Decomments	NED JAC	PI HEE KE SKE STRES STRE	PE T T SRVWD	APVD
UBLE CONTAIL NER CORE SEI 3 FOR OUTER Providence Theory Dig Name Prog. Stress & Support- Paced: May 13, 2016, 1:27 pm	NED JAC	PI HEE KE SKE STRES STRE	PE T T S RVWD	APVD



DP MATERIALS				
COMPONENT DESCRIPTION	N. S. (I NCHES)	ΩΤΥ	BECH STK (TEL
31272316/3161 SM S SCHARE DDC				
312TP316/316L, SMLS, SCH40S, BBE	3	60' 2' 9' 5'	PPP(PPP(OBØGØXØ Obøgøxø
LD MATERIALS				
COMPONENT DESCRIPTION	N. S. (INCHES)	OTY	BECHT STK C	EL ODE
T, BOF-RLD-H00202, WELDED PIPE ATTCHMT				
PT, BOF-RLD-H00203, WELDED PIPE ATTCHMT PT, BOF-RLD-H00204, WELDED PIPE ATTCHMT PT, BOF-RLD-H00205, WELDED PIPE ATTCHMT		1		
T. BOF-RLD-H00206, WELDED PIPE ATTCHMT T. BOF-RLD-H00207, WELDED PIPE ATTCHMT T. BOF-RLD-H00207, WELDED PIPE ATTCHMT		1		
		1		
LIBLE CONTAIN		ווס		
UBLE CONTAIN NER CORE SEE 4 FOR OUTER		PII EE KE	PE T T	
UBLE CONTAIN NER CORE SEE 4 FOR OUTER	NED E SH JAC	PII EE KE	PE T T	
UBLE CONTAIN NER CORE SEE 4 FOR OUTER	PIED S SH JAC	PII EE KE	PE T T	
UBLE CONTAIN NER CORE SEE 4 FOR OUTER	PIED SH JAC	PII EE KE	T	
UBLE CONTAIN NER CORE SEE 4 FOR OUTER	PIED SH JAC	PII EE KE	PE T T	VII
UBLE CONTAIN NER CORE SEE 4 FOR OUTER Browner WWWWWWWWWWWWWWWWWWWWWWWWWWWWWWWWWWWW	PIED SH JAC	PII EE KE	SH RVWD	APVD
UBLE CONTAINAL NER CORE SED 4 FOR OUTER Browner Browner WINNER 77058 ISSUED FOR CONSTRUCTION DESCRIPTION REVISION HISTORY	PIED SAL JAC	PII EE KE	SH RVWD	<u>V</u> L APVD
UBLE CONTAIN NER CORE SEE 4 FOR OUTER WINDER WINDER Y7058 ISSUED FOR CONSTRUCTION DESCRIPTION REVISION HISTORY WASTE TREZ Z435 STEVE RIVER PROT	PECTION PRO	PII EE KE	SH RVWD	APVD
UBLE CONTAINAL NER CORE SEE A FOR OUTER Reviewed Mode Commercial Sector Provide May 13, 20 Provide May 13, 20 ISSUED FOR CONSTRUCTION DESCRIPTION REVISION HISTORY RIVER PROT WASTE TREA 24,35 STEVE RICHLAND, W CONTRACT No: DE-AC27-01	PECTION PRO	PII EE KE	SH RVWD	APVD
UBLE CONTAIN NER CORE SED 4 FOR OUTER Reviewed No Comme Product May 13, 20 Product May 13, 20 Product May 13, 20 T7058 ISSUED FOR CONSTRUCTION DESCRIPTION REVISION HISTORY RIVER PROD WASTE TREA 2435 STEVE RICHLAND, W CONTRACT No: DE-AC27-01 F-M6-PWD-00058002 CONST. WOR	PIED SHAC SHAC SHAC SHAC SHAC SHAC SHAC SHAC	PII EE KE	SH RVWD	APVD
UBLE CONTAIN NER CORE SEE 4 FOR OUTER Reviewed No comment Providence of the sec No comment Providenc	PIED SALAC	PII EE KE	SH RVWD	APVD
UBLE CONTAIN NER CORE SES A FOR OUTER Reviewed Market Ses Y7058 ISSUED FOR CONSTRUCTION DESCRIPTION REVISION HISTORY CONTRACT No: DE-AC27-01 F-M6-PWD-00058002 CONST. WOR BALANCE OF FACILL ISOMETRIC ZS-66991-W31A-3	PIPE SPEC	PII EE KE		APVD
UBLE CONTAIN NERCORE SEE Notes and for the second	PIPE SPEC	PII EE KE	SH RVWD	APVD 0



MATERIALS	N. S.	BECHTEL	1
COMPONENT DESCRIPTION	(INCHES) DTY	STK CODE	
2TP316/316L, SMLS, SCH40S, BBE	3 117'0	PPPC0B0G0X0	2
MATERIALS			
COMPONENT DESCRIPTION	N. S. (INCHES) OTY	BECHTEL STK CODE	
<u>s</u>			
BOF-RLD-H00209, WELDED PIPE ATTCHMT BOF-RLD-H00210, WELDED PIPE ATTCHMT BOF-RLD-H00211, WELDED PIPE ATTCHMT BOF-RLD-H00212, WELDED PIPE ATTCHMT BOF-RLD-H00213, WELDED PIPE ATTCHMT BOF-RLD-H00214, WELDED PIPE ATTCHMT	1 1 1 1 1 1 1		
NTENTS OF THIS DOO BEROUS WASTE PERM	CUMENT	ARE CTING	
NER CORE SE			N D-75660010
NER CORE SE 5 FOR OUTER	NED P E SHE JACK	IPE ET ET	temp/24590-B0F-P3-RI D-75669910
NER CORE SE 5 FOR OUTER	NED P E SHE JACK	IPE ET ET	ton temp > 24590-B0F-P3-RLD-75669910
NER CORE SE 5 FOR OUTER	NED P E SHE JACK		Deskton/temo/24590-B0F-P3-RLD-75669910
NER CORE SE 5 FOR OUTER	NED P E SHE JACK	IPE ET ET	nael/Desktan/tema/24590-80F-P3-RI D-75669910
NER CORE SE 5 FOR OUTER Reviewed No comm By Constance Place University of the second Description	NED P E SHE JACK		wiennel/Deskton/temn/24590-B0F-P3-RI D-75669910
NER CORE SE SFOR OUTER	NED P E SHE JACK		sers/wiengel/Deskton/temp/24590-B0F-P3-RI 0-75669910
ISSUED FOR CONSTRUCTION REVISION HISTORY REVISION HISTORY REVISION HISTORY RIVER PR WASTE TH 2435 STE RICHLAND	NED P E SHE JACK		C:\\lsers\wiende\\Deskton\temp\24590-R0F-P3-RID-75669910
ISSUED FOR CONSTRUCTION REVISION HISTORY REVISION HISTORY	NED P ESHE JACK	SH_W RESS RVWD APV	INAL C:\\]sers\wiende\\Desktan\temp\24590-B0F-P3-RID-75669910
ISSUED FOR CONSTRUCTION DESCRIPTION REVISION HISTORY REVISION HISTORY CONTRACT No: DE-AC27- PTF-M6-PWD-00058002	PERSECTION PROJECTION		xFINAL C:NIsers\winnael\Deskton\temo\24590-R0F-P3-RID-7S669910
ISSUED FOR CONSTRUCTION REVISION HISTORY REVISION HISTORY REVISION HISTORY CONTRACT No: DE-AC27- PTF-M6-PWD-00058002 CONST.W BALANCE OF FACI ISOMETRIC	NED P SHE JACK	SH_W RESS RVWD APV	PF: ROF xFINAL C:VIIsers/wiende/Neskton/temo/24590-ROF-P3-RI D-75669910
ISSUED FOR CONSTRUCTION REVISION HISTORY REVISION HISTORY REVISION HISTORY REVISION HISTORY REVISION HISTORY REVISION HISTORY RIVER PR WASTE TH 2435 STE RICHLAND CONTRACT No: DE-AC27- PTF-M6-PWD-00058002 CONST. W BALANCE OF FACI ISOMETRIC	PIPE SPEC:	VV31A	TYPE: ROF &FINAL C:\lsers\wiende\Deskton\temp\24590-ROF-P3-RLD-7S669910
ISSUED FOR CONSTRUCTION DESCRIPTION REVISION HISTORY CONTRACT No: DE-AC27- PTF-M6-PWD-00058002 CONST. W BALANCE OF FACI ISOMETRIC D-ZS-66991-W31A-3 BOF-P3-RLD-ZS66991005	PIPE SPEC:	VV31A	INDER AFINAL C:NIsers/wiende/Deskton/temo/24590-ROF-P3-RLD-7566a010



MATERIALS	NC			DECUTEI	
COMPONENT DESCRIPTION	(INCHE	5) 01	Y	STK COD	E
2TP316/316L, SMLS, SCH40S, BBE 2TP316/316L, SMLS, SCH40S, BBE	3 3	52' 9'	Ø" 5"	PPPCØB PPPCØB	060x02 060x02
MATERIALS					
COMPONENT DESCRIPTION	N, S.	S) QT	Y	BECHTEL STK COD	E
S , 80F-RLD-H00215, WELDED PIPE ATTCHMT , B0F-RLD-H00216, WELDED PIPE ATTCHMT , B0F-RLD-H00217, WELDED PIPE ATTCHMT , B0F-RLD-H00218, WELDED PIPE ATTCHMT , B0F-RLD-H00220, WELDED PIPE ATTCHMT , B0F-RLD-H00220, WELDED PIPE ATTCHMT					
NTENTS OF THIS DOO BEROUS WASTE PERM			A	RE	G
UBLE CONTAI	NE			PE	
UBLE CONTAI NER CORE SE 6 FOR OUTER	ed brokents	D F HE CK		PE T T	
OUBLE CONTAIL NER CORE SE 6 FOR OUTER	ed mentange Spess & Suppor			PE T T	
VUBLE CONTAIL NER CORE SE 6 FOR OUTER	ed mm-thangs 3, 2016, 128 pm			PET	
VUBLE CONTAIL NER CORE SE 6 FOR OUTER	ed ments a, 2016, 1:28 pm			PE	
UBLE CONTAI NER CORE SE 6 FOR OUTER	ed iments a, 2016, 1:28 pm		RES	PE T T	APVD
UBLE CONTAI NER CORE SE 6 FOR OUTER	ed maintange spess & Support		RES	PE T T	APVD
VUBLE CONTAIN NER CORE SE 6 FOR OUTER	ed mailer bases a Support a 2016, 128 pm	PLANT N PROJ NTER F	RES	PE T T	M
UBLE CONTAIN NER CORE SE 6 FOR OUTER	ed markings spess 4 Support 3, 2016, 1-28 pm	HKD ST	RES	PE T T	APVD
UBLE CONTAINAL NER CORE SE 6 FOR OUTER Review With Condition of the second secon	ed mechange space a support spress & sup	HKD ST N PROJ PLANT N PROJ N PLANT	RES	PE T T	APVD
UBLE CONTAIN NER CORE SE 6 FOR OUTER	ed me-thange space a support a 2016, 128 pm	PLANT N PROJ N PROJ NTER F 354	RES	PE T T	APVD
UBLE CONTAIN NER CORE SE 6 FOR OUTER	A 2016, 128 pm OTECTION VENS CE WA 99 OTRV14 ORK ARE PIP	PROJ HKD ST N PROJ PLANT NTER F 354 136 A: 03A S E SPEC		PE T T	M APVD
VUBLE CONTAIN NER CORE SE 6 FOR OUTER	A SUBSEA SUPPORT	PROJUNCT	RES	PE T T S RVWD	APVD



MATERIALS	AL C		ACCUTC	
COMPONENT DESCRIPTION	(INCHES)	OTY	STK COD	E
2TP316/316L, SMLS, SCH40S, BBE	3	117' 0'	PPPCØB	ØGØXØ2
MATERIALS				
COMPONENT DESCRIPTION	N. S. (INCHES)	<u>סדץ</u>	BECHTEL STK COD	Ē
BDF-RLD-H00221, WELDED PIPE ATTCHMT BDF-RLD-H00222, WELDED PIPE ATTCHMT BDF-RLD-H00223, WELDED PIPE ATTCHMT BOF-RLD-H00224, WELDED PIPE ATTCHMT BOF-RLD-H00225, WELDED PIPE ATTCHMT BOF-RLD-H00226, WELDED PIPE ATTCHMT		1 1 1 1 1		
NTENTS OF THIS DO BEROUS WASTE PER	CUME MIT AF	NT A	RE	3
UBLE CONTAI	NED	PI	PE	
UBLE CONTAI NER CORE SE 7 FOR OUTER	NED E SH) PI HEI CKE	PE ET T	
UBLE CONTAI NER CORE SE 7 FOR OUTER	MED ESP JAC		PE ET T	
UBLE CONTAI NER CORE SE 7 FOR OUTER	NEC E SH JAC		PETT	
UBLE CONTAI NER CORE SE 7 FOR OUTER	MED Solution Ments 2016. 128 pm		PE ET T	Yk
UBLE CONTAI NER CORE SE 7 FOR OUTER	MED JAC			V L APVD
UBLE CONTAINER NER CORE SE 7 FOR OUTER Provide	NEC SAC	PROJECC PROJECC		VK APVD
UBLE CONTAINED TO BALE CONTAINED	NEC Sease JAC	PROJECC DI STRET		V K APVD
UBLE CONTAI NER CORE SE 7 FOR OUTER Image: State of the second se	NEC Sease JAC	PROJECC PROJECC PLANT TER PLA 66 03A		APVD
UBLE CONTAINER DUBLE CONTAINER DESCRIPTION DESCRIPTION REVISION HISTORY CONTRACT No: DE-AC27- TF-M6-PWD-00058002 CONST. V BALANCE OF FAC ISOMETRIC	NEC Sease JAC	PROJECC PROJECC D STREE PROJECC PLANT FER PLA 66 03A		APVD
UBLE CONTAINED TO BALANCE OF FAC USANT DE -AC27- TF-M6-PWD-00058002 CONTRACT No: DE -AC27- TF-M6-PWD-00058002 CONST. V	NEC Sease JAC	PROJECC D STREE PROJECC PLANT FER PLA 66 03A S SPEC: N		APVD
UBLE CONTAINER Soft FOR CORE SES TOGS TOGS TOGS TOGS TOGS TOGS TOGS TEVIEWER TEVIEWER TEVIEWER TEVIEWER TEVER TEVIEWER TEVEN TEVIEWER RIVER PROVIDENTION DESCRIPTION RIVER PROVIDENTION RIVER PROVIDENT RIVER PROVID	NEC Sease JAC	PROJEC D STREE PROJEC PLANT FER PLA 4 66 03A S SPEC: V	PE T T SS RVWD	APVD



N. S. (1 NCHES)	DIV	-	
N. S. (INCHES)	DTV	DED. DEL	
	U 11	STK COU	DE
33	90′ 9° 9′ 5°	PPPCØE PPPCØE	80G0X0 80G0X0
N. S. (INCHES)	OTY	BECHTEL STK COD	E
	1 1 1 1 1 1		
	IT A FEC	RE	3
ewed omments to Chang - Ichang? I: Pipe Strees & Suppor	Reston		
Aay 13, 2016, 1:29 pm			
Nay 13, 2016, 1:29 pm			
Gay 13, 2016, 1229 pm			
17 VL		SH	JA
TE DRN CHKD	STRES	S RVWD	SA
TECTION P	STRES	SH. SRVWD	SM APVD
DRN CHKD DECTION PL EATMENT PL VENS CENTE WA 99354	STRES ROJECT ANT R PLAC	SRVWD	APVD
DRN CHKD VIECTION PL CHKD VENS CENTE WA 99354 DIR V 14 136	STRES ROJECT ANT R PLAC	SRVWD	APVD
DRN CHKD DRN CHKD DEATMENT PL VENS CENTE WA 99354 DIR V14136 DRK AREA:	STRES ROJECT ANT R PLAC	SRVWD	APVD
DRN CHKD DRN CHKD DRN CHKD DRN CHKD DEATMENT PL VENS CENTE WA 99354 D1RV14136 DRK AREA: LITIES	STRES ROJECT ANT R PLAC	SH SRVWD	APVD
DRN CHKD DRN CHKD DEATMENT PL VENS CENTE WA 99354 DIRV14136 DRK AREA: LITIES	STRES ROJECT ANT R PLAC 03A	SRVWD SRVWD	APVD
	NED JAC	NED PI E SHEE JACKE	NED PIPE SHEET JACKET



		~~~~~		-
SHOP MATERIALS				
COMPONENT DESCRIPTION	N. S. (INCHES)	OTY	BECHTEL STK CODE	
PIPE		-		
PE, A312TP316/316L, SMLS, SCH40S, BBE	3	89' 8*	PPPCØ80G0X02	
FIELD MATERIALS				
COMPONENT DESCRIPTION	N. S. (INCHES)	OTY	BECHTEL STK CODE	
SUPPORTS				
PE SUPT, BOF-RLD-H00235, WELDED PIPE ATTCHMT PE SUPT, BOF-RLD-H00236, WELDED PIPE ATTCHMT PE SUPT, BOF-RLD-H00237, WELDED PIPE ATTCHMT PE SUPT, BOF-RLD-H00238, WELDED PIPE ATTCHMT PE SUPT, BOF-RLD-H00239, WELDED PIPE ATTCHMT		1 1 1 1		
CONTENTS OF THIS DOO ANGEROUS WASTE PERM	CUMEN VIT AF	NT A	RE TING	
				1009.DGN
DOUBLE CONTAIL INNER CORE SE 059 FOR OUTER	NED E SH JAC	PI HEE KE	PE ET T	-3-RLD-ZS6699
			- 6	-

		No Com By: Christine Chano Org Name: Pipe S Placed: May 13,	nents g-kduung2 Stress & Supp 2016, 1:29 pr	port-Reston	IJ		
7063							
, ISSUED FOR CO	NSTRUCTIO	N	TE	KK		SH	W
D	DESCRIPTION		DRN	CHKD	STRESS	RVWD	APVD
REVIS	ION HIS	TORY					
CONTRACT	No: [	RIVER PR WASTE TE 2435 STE RICHLAND	OTECT REATME VENS WA	ION PF NT PL CENTE 99354	ROJECT ANT R PLACE	5	
TF-M6-PWD-0005	8002	CONST. W	ORK A	REA: (	).3A	-	-
BALAN	BALANCE OF FAC		LIT	IES			
D-ZS-66991-W3	31A-3		P	IPE SF	EC: W3	1A	
BOF-P3-RLD-Z	S6699100	9			-	REV	0
wiengel	PLOT DATE:	5/12/2016	1	0:58:0	1 AM	SHT	1 OF



	III	R118	354	<b>       </b> 14	_
TERIALS					
OMPONENT DESCRIPTION	N. S.	<u>(S)</u>	Y	BECHTEL STK CODE	E
3, SMLS, STD WT, BBE 3, SMLS, STD WT, BBE	6 6	74	2'	PPPCH20 PPPCH20	0501 04 0501 04
105, CL3000 DW 89,72 DEG., EL - 90, SR, BW, STOWT 03₩P316•, SCH40S	6X1 6	1		PFLSGZ PFQNG50 PFCCV60	XØ1 PNN Ø53CØK ØG1 DØK
ATERIALS					
DMPONENT DESCRIPTION	N. S. (INCHE	<u>(S)</u>	Y	BECHTEL STK CODE	E
BOF-RLD-HØØ1 27		,			
Reviewed No Comments By: Christine Change	Ŋ				
Reviewed No Comments By: Cristine Charge? Org Name: Pipe Stress & Support - Reston Paced: May 02, 2017 REV 1 NOTE: REV DIMS, PAINT CODE, M INSUL BREAK LOCATION	NOTES, ISS		) E	ΥY	
Reviewed No Comments By: Cristing Charge - Reston Praced: May 02, 2017 REV 1 NOTE: REV 1 NOTE: REV DIMS, PAINT CODE, N INSUL BREAK LOCATION	NOTES, ISS RPP	SUEI -WT	) E P I	SY PDC	
Reviewed by Constitute Charge Org Name: Pipe Stress & Support - Restor Paced: May 02, 2017 REV 1 NOTE: REV DIMS, PAINT CODE, M INSUL BREAK LOCATION	NOTES, ISS RPP.	UEI WT	) E P I	PDC	AFE
Reviewed By Creatine Charge Kharge Org Name: Pipe Stress & Support - Reaton Placed: May 02, 2017 REV 1 NOTE: REV DIMS, PAINT CODE, M INSUL BREAK LOCATION SEE REV 1 NOTE SSUED FOR CONSTRUCTION	NOTES, ISS RPP JFT	UEI -WT	P I IC	PDC	JAC
Reviewed by Constitute Charge - Kharge Org Name: Pipe Stress & Support - Renton Placed: May 02, 2017 REV 1 NOTE: REV DIMS, PAINT CODE, 1 INSUL BREAK LOCATION INSUL BREAK LOCATION SEE REV 1 NOTE ISSUED FOR CONSTRUCTION DESCRIPTION			D E P I IC TRES	PDC SH, SH SRVWD	JMC APVD
Reviewed W: Creating Charge : Kharge Org Name: Pipe Stress & Support - Restor Placed: May 02, 2017 REV 1 NOTE: REV DIMS, PAINT CODE, M INSUL BREAK LOCATION SEE REV 1 NOTE ISSUED FOR CONSTRUCTION DESCRIPTION REVISION HISTORY REVISION HISTORY RIVER WASTE 2435 : RICHLA	NOTES, ISS RPP, JFT JFT DRN PROTECTIO STEVENS C AND, WA 99	SUEI -WT ZB CHKD S ON PRO ENTER 9354	D E P I IC IRES	SY DC SH SH SRVWD	JMC APVD
Reviewed By Creating Charge & Construction Rev 1 NOTE: REV 1 NOTE ISSUED FOR CONSTRUCTION DESCRIPTION REVISION HISTORY REVISION HISTORY REVISION HISTORY RIVER WASTE 2435 - RICHLA	NOTES, ISS RPP, JFT JFT DRN PROTECTIC TREATMENS STEVENS C AND, WA 99	SUEI -WT ZB CHKD S ON PRO IT PLAN ENTER 9354	D E P I IC IRES	ST SH SRVWD	JMC APVD
Reviewed by Creating Charge Stress & Support - Reator Placed May 02 2017 REV 1 NOTE: REV 1 N	NOTES, ISS RPP. JFT JFT DRN V7-01RV14 T. WORK AR CILITII	CHKD S CHKD S CHCH S CH	D B P I IC IRES JECT IT PLAC	S PDC SH S RVWD	JMC
Reviewed by Creating charge Creating Charge shares Org Name: Pipe Stress & Support - Renton Tacced: May 02, 2017 REV 1 NOTE: REV 1 NOTE: REV 1 NOTE: REV 1 NOTE: REV 1 NOTE: REV 1 NOTE: REV 1 NOTE ISSUED FOR CONSTRUCTION DESCRIPTION REVISION HISTORY REVISION HISTORY RIVER WASTE 24.35 RICHLA CONTRACT No: DE-AC2 F-M6-PWD-00058002 CONST BALANCE OF FAM ISOMETR	NOTES, ISS RPP. JFT JFT DRN V7-01RV14 T. WORK AR CILITII	UEI -WT ZB CHKD S DN PRO IT PLAN ENTER 9354 4136 EA: 25 ES	P I IC IRES JECT T PLAC	BY DC SH SRVWD	JMC APVD
Reviewed by Creating Charges & Support - Reator Parced: May D2 2017 REV 1 NOTE: REV 1 NOTE: REV DIMS, PAINT CODE, M INSUL BREAK LOCATION SEE REV 1 NOTE ISSUED FOR CONSTRUCTION DESCRIPTION REVISION HISTORY REVISION HISTORY RIVER WASTE 2435 : RICHLA CONTRACT No: DE - AC2 F-M6-PWD-00058002 CONST BALANCE OF FAM ISOMETR ZS-66991-W31A-6	NOTES, ISS RPP. JFT JFT JFT DRN PROTECTIC TREATMEN STEVENS C STEVENS C STEVENS C IND, WA 99 7-01RV14 T. WORK AR CILITII IC	SUEI -WT ZB CHKD S ON PRO ENTER 9354 4136 EA: 25 ES ES	D E P I IC IRES JECT PLAC	SY DC SH SH SRVWD	JMC APVD



MATERIALS			
	N. S.	OTY	BECHTEL
CONTONENT DESCRIPTION	(TNCHES)	011	SIK CODE
6-B, SMLS, STO WT, BBE 6-B, SMLS, STO WT, BBE	6 6	92' 7' 9' 5'	PPPCH2050104 PPPCH2050104
ITENTS OF THIS EROUS WASTE P	DOCUMEI ERMIT AF	NT A	RE TING
UBLE CONT		PI	PE FT
UBLE CONT FER JACKET	AINED SEE S ER CO	PI SHE RE	PE EET
UBLE CONT FER JACKET 002 FOR INN	AINED SEE S ER CO	PI SHE RE	PE
UBLE CONT FER JACKET 002 FOR INN Protein Un Manie Pars Protein William 7064	AINED SEE S ER CO	PI	PE
UBLE CONT FER JACKET 002 FOR INN Reviewe No Comm Dro Manne: Proce Dro Manne: Proce Dro Manne: Proce Dro Manne: Proce Dro May 17, 2004	AINED SEES ER CO	PI	PE
UBLE CONT FER JACKET 002 FOR INN Reviewe No Comm Dig Manner Pipe S Plood May 17.3	AINED SEE S ER CO	PI	
UBLE CONT ER JACKET 002 FOR INN Reviewe Mo Comm Dro Manne Pipe S Plood May 17.3 15SUED FOR CONSTRUCTION DESCRIPTION	AINED SEES ER CO		
UBLE CONT FER JACKET DO2 FOR INN Reviewe Browning Comp Dig Name: Pipe S Plood: May 17, 2 UIII 15SUED FOR CONSTRUCTION DESCRIPTION REVISION HISTOR	AINED SEE S ER CO		PE EET SH. ML
UBLE CONT ER JACKET DO2 FOR INN Reviewe Mo Comme Dry Marrie Pipe S Pload May 17. 15SUED FOR CONSTRUCTION DESCRIPTION REVISION HISTOR' WAY 24.3 RIC	AINED SEE S ER CO C C C C C C C C C C C C C C C C C C	PROJECT PROJECT LANT ER PLA	PE EET Soft Mile SS RVWD APVD
UBLE CONT ER JACKET DO2 FOR INN Reviewe Mo Comme Provintine Creat Dry Marine Prop S Pload May 17. 15SUED FOR CONSTRUCTION DESCRIPTION REVISION HISTOR REVISION HISTOR REVISION HISTOR	AINED SEE S ER CO C C C C C C C C C C C C C C C C C C	PROJECT D STRES	STAL MARCON
UBLE CONT ER JACKET DO2 FOR INN Reviewe Vorume Page 3 Pacet May 17.3 UIIIIIIII 7064 ISSUED FOR CONSTRUCTION DESCRIPTION REVISION HISTOR 24.3 RIC CONTRACT No: DE-A	AINED SEER CO ER CO C C C C C C C C C C C C C C C C C C	PROJECT D STRES PROJECT LANT ER PLA 4 6 034	SRVWD APVD
UBLE CONT ER JACKET DO2 FOR INN Reviewe Mo Comme Dy Comme Dy Comme Dy Comme Dy Comme Dy Comme Dy Comme Dy Comme Comme Dy Comme Dy Comme Dy Comme Dy Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme Comme C	AINED SEE S ER CO ER CO ER CO ER CO ENTRE S ENTRE & ENT ENTRE & STEVENS CHILAND, WA 9935 CC27-01RV1413 ENTST. WORK AREA: ACILITIES RIC	PROJECT D STRES PROJECT PLANT ER PLA 4 6 03A	SH. M.
UBLECONT CONTRACT No: DE-A PTF-M6-PWD-00058002 CO BALANCE OF F ISOMET	AINED SEE S ER CO C C C C C C C C C C C C C C C C C C	PROJECC D STRES D STRES D STRES CLANT ER PLA 4 6 03A 5 SPEC: V	PE EET S S RVWD APVD
UBLE CONT ER JACKET DO2 FOR INN Reviewe Mo Comme Dig Marnie Pipe S Pload May 17.3 TO64 ISSUED FOR CONSTRUCTION DESCRIPTION REVISION HISTOR REVISION HISTOR REVISION HISTOR RIV WA 24.3 RIC CONTRACT No: DE-A TFF-M6-PWD-00058002 CO BALANCE OF F ISOMET D-ZS-66991-W31A-6 BOF-P3-RLD-ZS66991052	AINED SEECO SERCO CO Control Control State 121 pm Control State 121 pm Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control Control	PROJECT D STRES PROJECT D STRE	PE EET



ATERIALS				
		N. S.		BECHTEL
JUMPONENT DESCRIPTION		(INCHES)	QTY	STK CODE
B, SMLS, STO WT, BBE	1.11	6	117' 0'	PPPCH20501 04
TENTS OF THI	IS DOCI	IME		RE
ROUS WASTE	PERMI	TAF	FEC	TING
	and the second			
IBLE CON	TAIN	ED	DI	DE
JBLE CON	TAIN	ED	PI	PE
JBLE CON ER JACKE	ITAIN	ED	PI	PE
JBLE CON ER JACKE 03 FOR IN	ITAIN T SE NER	ED E S CO	PI	PE ET
JBLE CON ER JACKE 03 FOR IN	ITAIN T SE NER	ED ES CO	PI SHE RE	PE
JBLE CON ER JACKE 03 FOR IN	ITAIN T SE NER	ED E S CO	PI SHE RE	PE
JBLE CON ER JACKE 03 FOR IN	ITAIN T SE NER	ED ES CO	PI SHE RE	PE
JBLE CON ER JACKE 03 FOR IN	Reviewed NER		PI SHE RE	PE
JBLE CON ER JACKE 03 FOR IN	Reviewed No Commen By Confidence Change Sitema Dry Name: Change Sitema Pictoritime Change Sitema Pictoritime Change Sitema Change Sitem	ED ES CO	PI SHE RE	PE
JBLE CON ER JACKE 03 FOR IN	Reviewed NER Providence Charger - Islan Org Name: Pipe Stress & Pleased: May 13, 2016, 12	ED ES CO	PI SHE RE	PE
JBLE CON ER JACKE 03 FOR IN	Reviewed NER With Charles - Lither Org Name: Pipe Stress & Pilacet: May 13, 2016, 1;	ED ES CO		PE
JBLE CON ER JACKE 03 FOR IN	Reviewed NER By Charlene Charles By Charlene Charles Packade: May 13, 2016, 1:	ED ES CO		PE
JBLE CON ER JACKE 3 FOR IN 3 FOR IN	Reviewed NER By Contange - Islan Org Name: Pipe Stress & Pilaced: May 13, 2016. 1:	ED ES CO		SH JA
JBLE CON ER JACKE 03 FOR IN	Reviewed No Commen Di Christie Charge Ster Di Christie Charge Ster Di Di Di Christie Charge Di Christie Charge Ster Di Char			SH M SRVWD APVD
JBLE CON ER JACKE 33 FOR IN 7065	REVIEWED NER Providence Character By Christiane Character By Character By Christiane Character By Charact	ED ES CO ts Support Restin 31 pm		SH M
JBLE CON ER JACKE 3 FOR IN 7065	Reviewed NER NER With the stars by contained and	ED ES CO Support Restor at pri	PI SHE RE	PE ET SH ML SRVWD APVD
JBLE CON ER JACKE 3 FOR IN 2065	Reviewed No Comment Di Christer Charge Di Christer Charge Di Christer Charge Di Christer Di Christer Charge Di Christer Di Chr	ED ES COO ts Support Restor 31 pm	PI SHE RE SHE SHE SHE SHE SHE STRESS	PE ET SH M SRVWD APVD
JBLE CON ER JACKE 3 FOR IN 3 FOR IN 3 FOR IN 3 FOR IN 7065	Reviewed NER By Content Charles By Content Charles	Support Restor Support Restor Support Restor RN CHKC CTION P MENT PI S CENTE 99354 V14136	PI SHE RE SHE SHE SHE SHE SHE SHE SHE SHE SHE SH	PE ET
JBLE CON ER JACKE 3 FOR IN 2065 SSUED FOR CONSTRUCTION DESCRIPTION REVISION HISTO CONTRACT NO: DE M6-PWD-00058002	Reviewed NER NER Dro Commen Dry: Ontel The Stress & Placed: May 13, 2016, 12 Placed: May 14, 201	ED ES CO Support Resident Stipm Support Resident Stipm CTION P MENT PU SCENTE 99354 V14136E AREA:	PI SHE ROJECT ANT ROJECT ANT R PLACE	SH M
JBLE CON ER JACKE 3 FOR IN 3 FOR IN 55UED FOR CONSTRUCTION DESCRIPTION REVISION HISTO WE-PWD-00058002 BALANCE OF	Reviewed No Comment Dig Name: Pipe Stress & Piteorither Charge Dig N	ED ES Support-Restor 31 pm Support-Restor 31 pm EN CHKE CTION P MENT PI S CENTE 99354 V14136 AREA: TIES	PI SHE RE SHE SHE SHE SHE SHE SHE SHE SHE SHE SH	SH M SRVWD APVD
JBLE CON ER JACKE 33 FOR IN 35 FOR IN 35 JEST CONSTRUCTION DESCRIPTION REVISION HISTO CONTRACT NO: DE M6-PWD-00058002 BALANCE OF ISOME	Reviewed Mocommen By Cristine Charge - Islan Org Name: Pipe Stress & Praced: May 13, 2016, 12 Praced: May 13, 2016, 12 Pr	Support Restor Support Restor TIES	PI SHE RE SHE SHE SHE SHE SHE SHE SHE SHE SHE SH	SH M
JBLE CON ER JACKE 33 FOR IN 35 FOR IN 35 JOS I	Reviewed NER By Contendence By Conte	Support Resident Support Resident Suppor	PI SHE ROJECT ANT ROJECT ANT ROJECT ANT ROJECT ANT ROJECT ANT ROJECT ANT ROJECT ANT ROJECT ANT ROJECT ANT ROJECT ANT ROJECT ANT ROJECT ANT ROJECT ANT ROJECT ANT ROJECT ANT ROJECT ANT ROJECT ANT ROJECT ANT ROJECT ANT ROJECT ANT ROJECT ANT ROJECT ANT ROJECT ANT ROJECT ANT ROJECT ANT ROJECT ANT ROJECT ANT ROJECT ANT ROJECT ANT ROJECT ANT ROJECT ANT ROJECT ANT ROJECT ANT ROJECT ANT ROJECT ANT ROJECT ANT ROJECT ANT ROJECT ROJECT ANT ROJECT ANT ROJECT ANT ROJECT ANT ROJECT ANT ROJECT ANT ROJECT ANT ROJECT ANT ROJECT ANT ROJECT ANT ROJECT ANT ROJECT ANT ROJECT ANT ROJECT ANT ROJECT ANT ROJECT ROJECT ANT ROJECT ANT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJECT ROJ	SH M SH M SRVWD APVD
JBLE CON ER JACKE 3 FOR IN 2065 SUED FOR CONSTRUCTION DESCRIPTION REVISION HISTO CONTRACT NO: DE M6-PWD-00058002 BALANCE OF ISOME S-66991-W31A-6 F-P3-RLD-ZS66991053	Reviewed No Commen Dry Name: Pipe Stress & Picoritine Charge Dry Name: Pipe Stress & Piacet: May 13, 2016, 1: Piacet: May 14, 2016, 1: Piacet: May	EDES COO Support Resolution Step Support Resolution Step RN CHIKE CTION P MENT PI 993554 V141366 AREA: TIES PIPE SI	PI SHE SHE SHE SHE STRESS ROJECT ANT R PLACE	BIA REV 0





COMPONENT DESCRIPTION       Inches)       OTY       BECI         -B, SMLS, STD WT, BBE       6       117'0'       PPI         -B, SMLS, STD WT, BBE       6       117'0'       PPI         JTENTS OF THIS DOCUMENT ARE EROUS WASTE PERMIT AFFECTII	HTEL CODE PCH20501 04
B. SMLS. STD WT. BBE 6 117'0' PPI	PCH2050104
TENTS OF THIS DOCUMENT ARE ROUS WASTE PERMIT AFFECTII	
	NG
JBLE CONTAINED PIPE ER JACKET SEE SHEE 5 FOR INNER CORE	Ē
;7	
	H JK
SSUED FOR CONSTRUCTION # CX S, DESCRIPTION DRN CHKD STRESS RVV	
SSUED FOR CONSTRUCTION     Image: Construction     Image: Construction     Image: Construction     Stress     Stress     Stress     RVV       DESCRIPTION     DRN     CHKD     STRESS     RVV       REVISION     HISTORY       Image: Construction     River     PROTECTION     PROJECT       WASTE     TREATMENT     PLANT     2435     STEVENS     CENTER     PLACE       RICHLAND, WA     99354	
SUED FOR CONSTRUCTION     Image: Constru	
SSUED FOR CONSTRUCTION       Image: Constructi	S BOE VETMAL
SSUED FOR CONSTRUCTION       Image: Constructi	TYPE: ROF STMAL

![](_page_125_Figure_0.jpeg)

TERIALS			DEQUITE
OMPONENT DESCRIPTION	I INCHES	OTY	STK CODE
B. SMLS. STD WT. BBE B. SMLS. STD WT. BBE	6 6	52' Ø' 9' 5'	PPPCH20501 04 PPPCH20501 04
ITENTS OF THIS I	DOCUMEI	NT A	RE
ROUS WASTE PI	ERMIT AF	FEC	TING
12 1 2 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	No. 2 Comment	1.	
JBLE CONT	AINED	PI	PE
JBLE CONT	AINED SEE S	PI	PE
JBLE CONT ER JACKET 06 FOR INN	AINED SEE S ER CO	PI	PE EET
JBLE CONT ER JACKET 06 FOR INN	AINED SEE S ER CO	PI	PE
JBLE CONT ER JACKET 06 FOR INN	AINED SEE S ER CO	PI SHE RE	PE
JBLE CONT ER JACKET 06 FOR INN	AINED SEE S ER CO	PI BHE RE	PE
JBLE CONT ER JACKET 06 FOR INN	AINED SEE S ER CO error construction of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the	PI SHE RE	PE
JBLE CONT ER JACKET 06 FOR INN	AINED SEES ER CO	PI BHE RE	PE
JBLE CONT ER JACKET 06 FOR INN	AINED SEES ER CO		PE
JBLE CONT ER JACKET 06 FOR INN	AINED SEES ER CO		PE EET SH IN
JBLE CONT ER JACKET 06 FOR INN	AINED SEES ER CO		PE EET SH IN SS RVWD APVD
JBLE CONT ER JACKET 06 FOR INN Reviewed Base ISSUED FOR CONSTRUCTION DESCRIPTION REVISION HISTORY RIVE	AINED SEES SER CO Prosteres & Support Restor by 17, 2018, 1:21 pm	PI SHE DRE	PE EET SH IN SS RVWD APVD
JBLE CONT ER JACKET 06 FOR INN Reviewed Solution SSUED FOR CONSTRUCTION DESCRIPTION REVISION HISTORY WAS 243 RICH	AINED SEES SEES Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charles Charle	PROJEC LANT ER PLA	PE EET SH M SS RVWD APVD
USUED FOR CONSTRUCTION REVISION HISTORY REVISION HISTORY	AINED SEES SEES Comments Char Lease Char Streams Char Str	D STRE	PE ET SH. M. SS RVWD APVD
JBLE CONTACT No: DE-AC	AINED SEE S SEE S SER CO Puese & Support Restor by 17. 2016, 121 pm	PROJEC PROJEC PROJEC PLANT 6 03A	PE EET SH. ME SS RVWD APVD
JBLE CONTACT No: DE-AG CONTRACT No: DE-AG CONTRACT No: DE-AG BALANCE OF F. ISOMET	AINED SEES SERCO emed omments competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents competents com	PI BHE D STRE	PE ET SH. M. SS RVWD APVD
JBLE CONTACE ER JACKET 06 FOR INN Review 15SUED FOR CONSTRUCTION DESCRIPTION REVISION HISTORY REVISION HISTORY REVISION HISTORY REVISION HISTORY REVISION HISTORY CONTRACT No: DE-AC CONTRACT No: DE-AC CONTRACT No: DE-AC CONTRACT No: DE-AC CONTRACT No: DE-AC CONTRACT No: DE-AC	AINED SEESS ERCO emed omments Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Competents Com	PI BHE D STRE D STRE PROJEC LANT ER PLA 6 03A 5 SPEC: 1	PE EET S SH. M. SS RVWD APVD T CE
JBLE CONTACE OF FOR INNI SUED FOR CONSTRUCTION DESCRIPTION REVISION HISTORY REVISION HISTORY CONTRACT No: DE-AC CONTRACT NO: DE-AC	AINED SEESS ERCO wed omments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments Programments P	PROJEC D STRE PROJEC LANT ER PLA 6 03A SPEC: 1	PE EET S S RVWD APVD T CE

![](_page_126_Figure_0.jpeg)

ITENTS OF THIS DOCUMENT ARE		
	ATERIALS	
	COMPONENT DESCRIPTION	N. S. BECHTEL (INCHES) OTY STK CODE
ITENTS OF THIS DOCUMENT ARE	-B, SMLS, STO WT. BBE	6 117' 0* PPPCH20501 04
LINUUS VASTE FERIVIT AFFECTING T	NTENTS OF THIS EROUS WASTE P	DOCUMENT ARE
	UBLE CONT	AINED PIPE
JBLE CONTAINED PIPE	EK JACKEI	SEE SHEEL
JBLE CONTAINED PIPE ER JACKET SEE SHEET	OT FOD ININI	
JBLE CONTAINED PIPE ER JACKET SEE SHEET 07 FOR INNER CORE	07 FOR INN	ER CORE
JBLE CONTAINED PIPE ER JACKET SEE SHEET 07 FOR INNER CORE	07 FOR INN	ER CORE
JBLE CONTAINED PIPE ER JACKET SEE SHEET 07 FOR INNER CORE	07 FOR INN	
JBLE CONTAINED PIPE ER JACKET SEE SHEET 07 FOR INNER CORE	07 FOR INN	ER CORE
JBLE CONTAINED PIPE ER JACKET SEE SHEET 07 FOR INNER CORE	07 FOR INN	ER CORE
JBLE CONTAINED PIPE ER JACKET SEE SHEET 07 FOR INNER CORE	07 FOR INN	ER CORE
JBLE CONTAINED PIPE ER JACKET SEE SHEET 07 FOR INNER CORE		ER CORE
JBLE CONTAINED PIPE         ER JACKET SEE SHEET         O7 FOR INNER CORE         Market         Market <td>15SUED FOR CONSTRUCTION DESCRIPTION</td> <td>ER CORE</td>	15SUED FOR CONSTRUCTION DESCRIPTION	ER CORE
JBLE CONTAINED PIPE ER JACKET SEE SHEET 07 FOR INNER CORE         OT FOR INNER CORE         OT FOR INNER CORE         INNER CORE         OT FOR CONSTRUCTION         ISSUED FOR CONSTRUCTION         ISSUED FOR CONSTRUCTION         DESCRIPTION         DRN CHKD STRESS RVWD APVD         DESCRIPTION         DRN CHKD STRESS RVWD APVD	07 FOR INN	ER CORE
USUED FOR CONSTRUCTION       Image: Brand Stress Rest         ESUED FOR CONSTRUCTION       Image: Brand Stress Rest         DESCRIPTION       Image: Brand Stress Rest         DESCRIPTION       Image: Brand Stress Rest         Image: Brand Stress Rest       Image: Brand Stress Rest         Image: Brand Stress       Image: Brand Stress Rest	07 FOR INN	ER CORE
JBLE CONTAINED PIPE ER JACKET SEE SHEET 07 FOR INNER CORE         Image: Contract See Sheet S	O7 FOR INN	ER CORE
JBLE CONTAINED PIPE ER JACKET SEES SHEET OT FOR INNER CORE         Image: I	07 FOR INN CONTRACT No: DE-A F-M6-PWD-00058002 CO	ER CORE
JBLE CONTAINED PIPE ER JACKET SEE SHEET 07 FOR INNER CORE         Image: See Sheet 07 For Inner Prostree State 07 For Construction         Image: Support Reader Description         Image: See See See See See See See See See S	O7 FOR INN O7 FOR INN CONTRACT No: DE-A F-M6-PWD-00058002 CO BALANCE OF F ISOMET	ER CORE
UBLE CONTAINED PIPE ER JACKET SEE SHEET OT FOR INNER CORE         INNER CORE CONSTRUCTION PROJECT         <td colspan="</td> <td>O7 FOR INN O7 FOR INN CONTRACT NO: DE-A F-M6-PWD-00058002 CO BALANCE OF F ISOMET ZS-66991-W31A-6</td> <td>ER CORE Were Protection Project Market New 13, 2016, 1:32 pm Market New 14, 2016, 1:32 pm Ma</td>	O7 FOR INN O7 FOR INN CONTRACT NO: DE-A F-M6-PWD-00058002 CO BALANCE OF F ISOMET ZS-66991-W31A-6	ER CORE Were Protection Project Market New 13, 2016, 1:32 pm Market New 14, 2016, 1:32 pm Ma
JBLE CONTAINED PIPE ER JACKET SEE SHEET O7 FOR INNER CORE         Image: See Sheet OF Contract         Image: See See See See See See See See See S	07 FOR INN 069 ISSUED FOR CONSTRUCTION DESCRIPTION REVISION HISTOR' REVISION HISTOR' REVISION HISTOR' RIV WA 243 RIC CONTRACT NO: DE-A F-M6-PWD-00058002 CO BALANCE OF F ISOMET ZS-66991-W31A-6 OF-P3-RLD-ZS66991057	ER CORE

![](_page_127_Figure_0.jpeg)

MATERIALS			-	
COMPONENT DESCRIPTION	INCHES	) OTY	STK CODE	
-B, SMLS, STD WT, BBE -B, SMLS, STD WT, BBE	66	90' 9' 9' 5'	PPPCH205 PPPCH205	501 04 501 04
	001111			
EROUS WASTE PE	RMIT A	FFEC	TING	
			DE	_
UBLE CONTA		) PI	PE	
UBLE CONTA	SEE	O PI SHE	PE	
UBLE CONTA ER JACKET : 08 FOR INNE	NEE SEE R CO	) PI SHE DRE	PE	
UBLE CONTA ER JACKET : 08 FOR INNE	AINED SEE R CC	D PI SHE DRE	PE	
UBLE CONTA ER JACKET S 08 FOR INNE	ewed commentarys Pros Stress & Support R		PE	
UBLE CONTA ER JACKET S 08 FOR INNE	ewed comments to Charge Holmong the Stress & Support Pa day 17, 2016, 121 pm		PE	
UBLE CONTA ER JACKET : 08 FOR INNE	ewed comments to Page 15 ang tay 17, 2016, 121 pm		PE	
UBLE CONTA ER JACKET : 08 FOR INNE	ewed comments are Class - biological control - biol		PE	
JBLE CONTA ER JACKET : 08 FOR INNE	ewed comments aby 17, 2016, 121 pm		PE ET	Ke
JBLE CONTA ER JACKET S 08 FOR INNE	ewed comments are Class - biological comments represent a Support R day 17, 2016, 121 pm		PE EET SIH SS RVWD A	PVD
UBLE CONTA ER JACKET S 08 FOR INNE 08 FOR INNE Proof 0 ISSUED FOR CONSTRUCTION DESCRIPTION REVISION HISTORY	ewed comments aby 17, 2016, 121 pm		PE EET SIH SS RVWD A	PVD
UBLE CONTA ER JACKET 08 FOR INNE	ewed comments recharge learning representations about 17, 2016, 121 pm		PE ET SIH SS RVWD A	PVD
UBLE CONTAC ER JACKET S 08 FOR INNE	ewed comments in Charg-testing2 is Pipe Stroke & Support R day 17, 2016, 121 pm	PROJECT PLANT VITER PLAN 36	PE ET SIH SS RVWD A	PVD
UBLE CONTACT No: DE-AC2 TF-M6-PWD-00058002 CONST	ewed comments rectarged biological experiences reprosed support day 17, 2016, 121 pm	PROJECT PROJECT PLANT NTER PLA 36 A: 03A	PE ET SIH SS RVWD A	PVD
UBLE CONTACT No: DE-AC2 TF-M6-PWD-00058002 CONST USOURT DESCRIPTION	ewed commentations and the support and the sup	PROJECT PLANT NTER PLAN 36 A: 03A	PE ET SS RVWD A	W.L.
UBLE CONTACT No: DE-AC2 TF-M6-PWD-00058002 CONST BALANCE OF FA ISOMETR	ewed comments terpine State and Tree DRN CH PROTECTION Tree Tree And Area Stevens Cen And Area Cillitie Cillic	PROJECC PROJECC PROJECC PLANT VTER PLA 36 A: 03A S SPEC: V	PE ET SS RVWD A	PVD
UBLE CONTACT No: DE-AC2 TF-M6-PWD-00058002 CONS BALANCE OF FA ISOMETR D-ZS-66991-W31A-6 BOE-P3-RI D-ZS66991058	ewed comments in Charge Holdings in Pripe Stress & Support PA day 17, 2016, 121 pm PROTECTION TREATMENT STEVENS CEP AND, WA 993 27-01R V141 T, WORK AREA CILITIE IC	PROJEC PROJEC PLANT VIER PLAN 36 A: 03A S SPEC: V	PE ET SS RVWD A	PVD

![](_page_128_Figure_0.jpeg)

MATERIALS	J			67		0.000.000	
COMPONENT	DESCRIPTION	2	(1N	S. CHES)	OTY	BECHTEL STK COL	DE
-B, SMLS,	STD WT. BBE		6		89' 8'	PPPCH2	2050104
TEN	TS OF	THIS D		AEN AE		RE	2
RUI	JS VVA	SIEPE		Аг	FEC	TING	2
2. See 2.							
JBL	ECO	DNT	AINE	D	PI	PE	
JBL	E CO	ONT/		D	PI	PE	-
	E CO		AINE SEE	Des	PI	PE	Ē
UBL ER 09 I	E CO JACI FOR	ONT/ KET	AINE SEE ER C	D	PI HE RE	PE	Ē
UBL ER 09 I	E CO JACI FOR	ONT/ KET INNE	AINE SEE ER C	Deso	PI BHE RE	PE	Ē
UBL ER 09 I	E CO JACI FOR	ONT/ KET INNE	AINE SEE ER C		PI SHE RE	PE EE1	C
UBL ER 09 I	E CO JACI FOR	Reviewed No Comme Grane Page Str	AINE SEE ER C		PI SHE RE	PE	Ē
UBL ER 09 I	E CO JACI FOR	Reviewed No Comme Province Change Org Name: Pape Star Place: May 17, 20	AINE SEE ER C		PI HE RE	PE	C
	E CO JACI FOR	Reviewed No Comm Br Channe Chang St Phase: May 17, 20	AINE SEE ER C		PI HE RE	PE	
UBL ER 09 I	E CO JACI	Reviewed No Comm Br Charles Charles Org Name: Charles Phaced: May 17, 20	AINE SEE ER C			PE	
	E CO JAC FOR	CTION	AINE SEE ER C			PE ET	JML
JBL ER 09 F	OR CONSTRUE	CTION	AINE SEE ER C			PE EET SH SS RVWD	JML
	OR CONSTRUCT	CTION HISTORY	AINE SEE ER C		PI BHE RE	PE ET	M
	OR CONSTRUCT DESCRIP	CTION RIVE CTION PTION RIVE RIVE WAST 2435 RICH	AINE SEE ER C			PE EET SH SS RVWD	ML
UBL ER 09 I 09 I ISSUED F	OR CONSTRUE DESCRIP EVISION H	CTION HISTORY REVEWED No Comme By Conserved May 17, 20 CTION HISTORY RIVE WAST 2435 RICH	AINE SEE ERC	D S O S O S O S O S O S O S O S O S O S	PI SHE RE SHE RE SHE RE SHE SHE SHE SHE SHE SHE SHE SHE SHE SH	PE EET SH SS RVWD	JML APVD
UBL ER 09 I UBL 071 ISSUED F RE CONT F-M6-PWD	OR CONSTRUCT DESCRIP VISION H	CTION Placed May 17, 20 CTION PISTORY RIVE BY CHARGE Org Name: Pape Star Placed: May 17, 20 CTION Placed: May 17, 20 CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION CTION	AINE SEE ER C	D S S C HKI ION F C HKI C HKI ION F R I A 13 ( REA:	PI SHE RE STRES	PE ET	M
UBL ER 09 I 071 ISSUED F RE CONT F-M6-PWD BA	E CO JAC JAC JAC OR CONSTRUE DESCRIP EVISION F EVISION F OUD58002 LANCE ISC	CTION PTION HISTORY REVIEWED AND COMP BY CHANGE CON CON CON CON CON CON CON CON	AINE SEE ER C	DSCO CHKI ION PP CENTI 99354 IA130 REA: IES	PI SHE RE STRES OSA	PE EI SH SS RVWD	APVD
UBL ER 09 I 071 ISSUED F RE CONT F-M6-PWD BA	ACT No: -00058002 LANCE ISC -01-W31A-6	CTION HISTORY RIVE CON CON CON CON CON CON CON CON CON CON	AINE SEE ER C	DSCO	PI SHE RE SHE SHE SHE SHE SHE SHE SHE SHE SHE SH		APVD
UBL ER 09 I UBL 071 ISSUED F RE CONT F-M6-PWD BA	ACT No: -00058002 LANCE ISC 01-W31A-6	CTION HISTORY RICH DE-AC COR FA	AINE SEE ER C	Don PP CENTIP CHKI IA130 REA: IES	PI SHE RE SHE RE SHE SHE SHE SHE SHE SHE SHE SHE SHE SH	PE EET S S RVWD	JML APVD

![](_page_129_Figure_0.jpeg)

![](_page_129_Picture_1.jpeg)

OP MATERIALS			
COMPONENT DESCRIPTION	N.S. (INCHES)	<u>OTY</u>	BECHTEL STK CODE
A312TP316/316L, SMLS, SCH40S, BBE A312TP316/316L, SMLS, SCH40S, BBE	2 2	1 2′ 4" 4′ 9"	PPPCØBØGØXØØ PPPCØBØGØXØØ
ELD MATERIALS			
COMPONENT DESCRIPTION	N.S. (INCHES)	OTY	BECHTEL STK CODE
DRTS			
JPT, HLW-HDH-H10107, WELDED PIPE ATTCHMT		1	

#### DOUBLE CONTAINED PIPE INNER CORE SEE SHEET 051 FOR OUTER JACKET

## PLACE SPOOL A IN DRM

NOT POR/ ETE	E: ATED 24590-WTP-EIE-M REV., CHANGES NOT ID	IS-14-0003 REV DENTIFIED	.0				
1	Checked By: Scott Sheldon - dssheldo Org Name: HLW Plant Design Placeet: Apr 26, 2021	Reviewed No Comments By: Chi-Kei Fan - cfan Org Name: Plant Design Placed: May 07, 2021	<b>V</b>		Reviewed lo Comm : Les Leyland - LA g Name: Plant D aced: Apr 26, 20	<b>Dents</b> LEYLAN esign 121	<b>V</b>
21	SEE REV.1NOTE						JMC
/10	ISSUED FOR CONSTRUC	TION	RP	ZB	KM	LHS	РM
Е	DESCRIPT	ION	DRN	CHKD	STRESS	RVWD	APVD
	REVISION H	ISTORY					
		RIVER PRO WASTE TR 2435 STE RICHLAND,	DTECTI EATME VENS WA S	ON PF NT PL CENTE 99354	ROJECT ANT R PLACE		
	CONTRACT No:	DE-AC27-(	D1RV1	4136			
0-HL	W-M6-HDH-00002002	CONST. W	ORK A	REA: (	)3		
	HLW VITRIFIC	CATION E	BUII	_DII	NG		
HDF	I-PA-00033-W31A-2		P	IPE SF	PEC: W3	31A	
0-⊢	ILW-P3-HDH-PA0003	3001				REV	1
3Y:t	amcgee PLOT DA	TE: 4/23/2021	12	2:19:52	РМ	SHT	.1 OF

![](_page_130_Figure_0.jpeg)

PERMITTED

NOT

CHANGES

- MANUAL

GENERATED

COMPUTER

Issued by
<b>RPP-WTP PDC</b>

DP MATERIALS			
COMPONENT DESCRIPTION	N.S. (INCHES)	<u>OTY</u>	BECHTEL STK CODE
312TP316/316L, SMLS, SCH40S, BBE	4	12′11"	PPPCØBØGØXØ3
NGS			
), LR, BW, A403WP316/316L-S, SCH40S	4	6	PFQNV6ØG16ØH
LD MATERIALS			
COMPONENT DESCRIPTION	N.S. (INCHES)	QTY	BECHTEL STK CODE
DRTS			
JPT, HLW-HDH-H10108		1	

## DOUBLE CONTAINED PIPE OUTER JACKET SEE SHEET 001 FOR INNER CORE

#### PLACE SPOOL B IN DRM SHOP TO PROVIDE SPOOL D ONLY

W-P3-HDH-PA00033051.DGN

NOT POR/ ETE	E: ATED 24590-WTP-EIE-MS-14-0003 REV REV., CHANGES NOT IDENTIFIED	<i>'</i> .0				
1111	Checked By: Scott Sheldon - dssheldo Org Name: HLW Plant Design Placed: Apr 27, 2021	V		Reviewed No Comn r: Les Leyland - LA rg Name: Plant D aced: Apr 27, 20	d nents ALEYLAN Design 021	<b>V</b>
						71 10
-21	SEE REV 1NOTE				(	IMC
/10	ISSUED FOR CONSTRUCTION	RP	ZB	KM	LHS	PM
E	DESCRIPTION	DRN	CHKD	STRESS	RVWD	APVD
	REVISION HISTORY					
	RIVER PRI WASTE TR 2435 STE RICHLAND,	OTECT REATME VENS , WA	ION PF INT PL CENTE 99354	ROJECT ANT R PLACE	-	
	CONTRACT No: DE-AC27-	01RV	14136			
0-HL	_W-M6-HDH-00002002 CONST.W	'ORK A	REA: (	)3		
	HLW VITRIFICATION I ISOMETRIC	BUI	LDII	NG		
HDF	IDH-PA-00033-W31A-4			PEC: W3	31A	
90-⊢	ILW-P3-HDH-PA00033051				REV	1
BY:t	amcgee PLOT DATE: 4/27/2021	-	2:08:19	PM	SHT	.1 OF

![](_page_131_Figure_0.jpeg)

![](_page_131_Picture_1.jpeg)

OP MATERIALS			
COMPONENT DESCRIPTION	N.S. (INCHES)	QTY	BECHTEL STK CODE
A312TP316/316L, SMLS, SCH40S, BBE A312TP316/316L, SMLS, SCH40S, BBE	2 2	1 3′ 4" 4′ 9"	PPPCØBØGØXØØ PPPCØBØGØXØØ
ELD MATERIALS			
COMPONENT DESCRIPTION	N.S. (INCHES)	<u>OTY</u>	BECHTEL STK CODE
DRTS			
JPT, HLW-HDH-H10105, WELDED PIPE ATTCHMT		1	

## DOUBLE CONTAINED PIPE INNER CORE SEE SHEET 051 FOR OUTER JACKET

# PLACE SPOOL A IN DRM

REV 1 NOTE: INCORPORATED 24590-WTP-EIE-MS-14-0003 REV.0 COMPLETE REV., CHANGES NOT IDENTIFIED

1	Checked By: Scott Sheldon - dssheldo Org Name: HLW Plant Design Placed: Apr 26, 2021	Reviewed No Comments By: Chi-Kei Fan - Cán Org Name: Plant Design Placed: May 07, 2021			<b>ents</b> EYLAN sign 1	<b>V</b>	
22	SEE REV 1NOTE						7Mc
'10	ISSUED FOR CONSTRUC	TION	RP	ZB	КM	LHS	PM
E	DESCRIPT	TION	DRN	CHKD	STRESS	RVWD	APVD
	REVISION H	ISTORY					
RIVER PROTECTION PROJECT WASTE TREATMENT PLANT 2435 STEVENS CENTER PLACE RICHLAND, WA 99354							
)-HL	W-M6-HDH-00002002	CONST. W	ORK A	REA: (	)3		
ł	HLW VITRIFI ISC	CATION E	BUII	DII	NG		
١D⊢	I-PA-00041-W31A-2		P	IPE SF	PEC: W3	81A	
0-H	LW-P3-HDH-PA0004	1001				REV	1
3Y:t	amogee PLOT D	ATE: 4/23/2021	12	2:19:56	РМ	SHT	.1 OF

![](_page_132_Figure_0.jpeg)

COMPUTER GENERATED - MANUAL CHANGES NOT PERMITTED

Issued by				
<b>RPP-WTP PDC</b>				

DP MATERIALS			
COMPONENT DESCRIPTION	N.S. (INCHES)	Y	BECHTEL STK CODE
3312TP316/316L, SMLS, SCH40S, BBE	4	13′11"	PPPCØBØGØXØ3
NGS			
0, LR, BW, A403WP316/316L-S, SCH40S	4	6	PFQNV6ØG16ØH
ELD MATERIALS			
COMPONENT DESCRIPTION	N.S. (INCHES)	OTY	BECHTEL STK CODE
DRTS			
JPT, GU-G-4		1	

## DOUBLE CONTAINED PIPE OUTER JACKET SEE SHEET 001 FOR INNER CORE

PLACE SPOOL B IN DRM SHOP TO FABRICATE SPOOL D ONLY

	E: ATED 24590-WTP-EIE-MS-14-0003 REV REV., CHANGES NOT IDENTIFIED Checked	.0		eviewed o Comm	ents	
4	By: Scott Sheldon - dssheldo View By: Chi-Kei Fan - cfan Org Name: HLW Plant Design Placed: Apr 26, 2021 Org Name: Plant Design Placed: May 07, 2021	By: Les Leyland - LALEYLA Org Name: Plant Design Placed: Apr 27, 2021			EYLAN sign 1	
21	SEE REV 1NOTE					7Mc
10	ISSUED FOR CONSTRUCTION	RP	ZB	КM	LHS	PM
E	DESCRIPTION	DRN	CHKD	STRESS	RVWD	APVD
	REVISION HISTORY					
RIVER PROTECTION PROJECT WASTE TREATMENT PLANT 2435 STEVENS CENTER PLACE RICHLAND, WA 99354						

	ISSUED FOR CONSTRUCTION	RP	ZB	КM	LHS	РM	
	DESCRIPTION	DRN	СНКД	STRESS	RVWD	APVD	000
	REVISION HISTORY						+ / 1
RIVER PROTECTION PROJECT WASTE TREATMENT PLANT 2435 STEVENS CENTER PLACE RICHLAND, WA 99354							
CONTRACT No: DE-AC27-01RV14136							I V I V
LW-M6-HDH-00002002 CONST. WORK AREA: 03							
HLW VITRIFICATION BUILDING ISOMETRIC							
	I-PA-00041-W31A-4	P	IPE SF	PEC: W3	31A		>   -
ILW-P3-HDH-PA00041051					REV	1	1 O I

10:47:29 AM

PLOT DATE: 4/26/2021

![](_page_133_Figure_0.jpeg)

#### **GENERAL NOTES**

- GENERIC SYMBOLS AS SHOWN ON THIS DRAWING ARE APPLICABLE FOR ALL BUILDINGS.
- BASED ON RPP-WTP SITE GENERAL ARRANGEMENT PLAN 2. (24590-BOF-C2-C12T-00002), 1ST FLOOR ELEVATIONS(FT.): PTF = 678'-0" HLW = 682'-0" LAW = 681'-0"/7 LAB = 677'-0"
- 3. { DELETED ア
- FOR PIPING SPECS. S11E, S11B, W11B, AND S32B THE 4. SUBSTITUTION OF STUB-INS FOR LATERALS ARE NOT PERMITTED FOR ANY LABORATORY PIPING SYSTEM UNLESS SPECIFIED ON THE ISOMETRIC.
- 5. (DELĚTĚĎ
- SEE 24590-WTP-PW-P30T-00001 FOR FIELD WELD 6. FABRICATION REQUIREMENTS FOR BUTT WELDS UNLESS NOTED OTHERWISE ON THE ISOMETRIC. THIS DRAWING DOES NOT APPLY WHEN THE BUTT WELD IS AT A FITTING, eg. ELL, REDUCER, TEE, IN LINE VALVE, ETC.
- TO ACCOMMODATE FOR FIELD FIT-UP REQUIREMENTS 7. THE SHOP SPOOL FABRICATOR IS TO ADD 6" TO THE DIMENSION SHOWN FOR ONLY ONE OF THE TWO SPOOLS THAT MAKE UP THE JOINT. THE SPOOL WITH THE ADDITIONAL 6" SHALL HAVE A PLAIN END FOR THE FIELD TO PREP AS REQUIRED, DESIGNATED AS FFW-XX WHERE ŜPOŎĽŚ ĂŘĚ ŠHIPPĚĎ ŴIŤHĚ BĚVĚĽĚĎ ĚŇĎŠ, FIELD MAY UTILIZE OR REMOVE WHEN NECESSARY
- MULTI SHEET ISOS ARE USED IN CASES WHERE A SINGLE 8. SPOOL CAN NOT BE SHOWN ON ONE SHEET. THESE HAVE A SHEET NUMBER SUCH AS 101 OR 201 INSTEAD OF THE STANDARD 001 OR 002. THESE ISOMETRICS WILL HAVE A **REFERENCE ISO INDEX AND THE STATEMENT** "MULTI SHEET ISO" SHOWN ABOVE THE TITLE BLOCK ON EACH SHEET OF THE MULTI-SHEET ISOMETRIC SERIES. THERE IS NO WELD JOINT AT THE DRAWING CONTINUATION REFERENCE POINT(S) OF A PARTIAL SPOOL THAT IS SHOWN ON A MULTI-SHEET ISOMETRIC. EXAMPLE: NO WELD BETWEEN SHEET 101 AND 201 A SPOOL PIECE MARK WILL BE SHOWN ONLY ON THE FIRST SHEET OF THE MULTI-SHEET ISOMETRICS. THIS NOTE IS NOT RETROACTIVE. ISSUED MULTI SHEET ISOS WILL NOT BE REVISED DUE TO REVISIONS OF THIS NOTE.
- 9. THE DESIGNATION SHOWN ON PIPING ISOMETRICS REQUIRING SLOPE AND INDICATING A MINIMUM (i.e. MIN.) REQUIREMENT, IS INTENDED FOR THE FIELD INSTALLATION AND NOT FOR SHOP FABRICATION OF SPOOLS. THE PIPE SHOP FABRICATOR SHALL BUILD TO THE SLOPE INDICATED ON THE ISOMETRIC
- 10. PIPE SPOOL FABRICATORS ARE ALLOWED TO MITER PIPE/AND OR FITTING ENDS ANGULARITY BY A MAXIMUM OF 2 DEGREES TO ACHIEVE SLOPE REQUIREMENTS FOR WELDED CONNECTIONS.
- 11. ISOMETRIC DRAWINGS REPRESENT THE FINAL INSTALLED CONFIGURATION. WHERE SPOOLS CONTAINING VALVES ARE REQUIRED TO BE FABRICATED AND SHIPPED TO SUPPORT SCHEDULE, CONSTRUCTION MAY CHOOSE TO INSTALL THE VALVES IN THE FIELD. IN THESE CASES, VALVES SHOWN ON THE ISOMETRICS WILL NOT BE INSTALLED BY THE SUPPLIER AND THE PIPE WILL BE SPOOLED THROUGH IN PLACE OF THE VALVE. THE WELDS SHOWN AS SHOP WELDS WILL BE FIELD WELDS IN THESE CASES AND THE VALVES WILL BE FIELD MATERIAL INSTEAD OF SHOP MATERIAL.

![](_page_134_Figure_14.jpeg)

FORM B_SDN2.DGN 01/2003